

THE ROLE OF BIG DATA FOR THE DEVELOPMENT OF OPPORTUNITIES FOR SMMES

by

LUYANDA LINCOLN SWAARTBOOI

Thesis submitted in fulfilment of the requirements for the degree of

Master of Technology: Business Information Systems

in the Faculty of Business and Management Sciences

At the Cape Peninsula University of Technology

Supervisor: Dr AC de la Harpe

District Six Campus, Cape Town November 2022

CPUT copyright information

The thesis may not be published either in part (in scholarly, scientific or technical journals), or as a whole (as a monograph), unless permission has been obtained from the University

DECLARATION

I, LUYANDA LINCOLN SWAARTBOOI, declare that the contents of this dissertation represent my own unaided work, and that the thesis has not previously been submitted for academic examination towards any qualification. Furthermore, it represents my own opinions and not necessarily those of the Cape Peninsula University of Technology.

Signed

Date

ABSTRACT

Digitalisation, the internet and mobile data are offering large amounts of unstructured data that can be analysed and utilised by Financial Service Providers (FSPs) to achieve better effectiveness and competitive advantages. Big data analytics (BDA) is a tool for supporting the competitive advantages of FSPs by increasing data driven strategies.

With the new stream of technologies, FSPs are facing challenges on how to utilise their existing data in creating new innovative products or enhancing the existing services so that their customers can benefit. Despite the high adoption rate of digital strategies by South African FSPs, little is known on how these FSPs can use Big Data (BD) to differentiate their customer segments and to create a competitive advantage. The following research questions are asked: (i) What are the challenges FSPs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage and (ii) How can FSPs utilise BD analytics in order to create a competitive advantage?

The aim of this study is to explore how FSPs can use BD to differentiate their customer segments and to create a competitive advantage. The objective of the study is to (i) determine if FSPs sees the value of BD for their businesses, (ii) identify any challenges faced by FSPs to include BD strategy to their businesses, (iii) identify FSPs strategy to meet their customer's needs and (iv) examine the way FSPs are measuring their success.

An inductive research approach are followed on this study. Qualitative survey (experts) using interviews by means of semi-structured questionnaires were used during data collection. Fifteen FSPs were used to obtain experts (15) within the SMMEs. The sampling was done by non-randomly, purposively and convenient selected techniques. Data was collected by recording the interviews. Recordings were transcribed and the analysed. A thematic analysis was done on the data. The study considered several ethical issues and the researcher made sure that all research ethical standards were adhere to.

Keywords: Big Data; FSPs; BD analytics; customer segments; competitive advantage; and qualitative survey interview

iii

ACKNOWLEDGEMENTS

I would like to sincerely thank:

- My supervisor, Dr Andre De la Harpe without your guidance, support, wisdom, empathy and understanding, I would not have arrived at this point. I want to thank you for your patience and time, I will forever be indebted to you. May the Lord above abundantly and grant you more wisdom. Thank you for not allowing me to give up half way this thesis.
- My parent, Ncediwe Pamella Swaartbooi, you are always ready to support my aspirations and believe in me even more than I had believed in myself. I truly appreciate you!
- My wife, Zovuyo Grace Xuma, your support counts so much.
- My siblings, Vatiswa Swaartbooi and Asanda Swaartbooi, thank you for your support. I truly appreciate you guys.
- My sisters from other mothers, Yollanda Majokweni and Ntakadzeni Tshidada, thank you for your support and encouragement you provided to me during the study.
- My friends, associates, colleagues more special Simlindile Abongile Bantom, thank you for your encouragement and support you provided me during this study.

DEDICATION

This dissertation is dedicated to my late father, Loyiso Lovender Swaartbooi, although you are no longer present in our lives, I still feel your impact every day!

TABLE OF CONTENTS

DECLA	RATION	ii
ABSTR	АСТ	iii
ACKNO	WLEDGEMENTS	iv
DEDICA	ATION	v
TABLE	OF CONTENTS	vi
LIST O	F FIGURES	xi
LIST OI	F TABLES	xii
ABBRE	VIATIONS	xiii
DEFINI	TIONS AND KEY CONCEPTS	xiii
СНАРТ	ER ONE: INTRODUCTION	1
1.1	Introduction	1
1.2	Background to problem statement	3
1.3	Statement of the research problem	4
1.4	Research questions	5
1.5	Research aim	5
1.6	Objectives	6
1.7	Research methodology	6
1.7.1	Research philosophy	6
1.7.1.1	Ontology	6
1.7.1.2	Epistemology	6
1.8	Research approach	7
1.9	Research strategy	7
1.9.1	Case study	7
1.9.2	Unit of analysis	7
1.9.3	Unit of observation	7
1.9.4	Sampling	7
1.10	Data collection	8
1.11	Data analysis	8
1.12	Ethics	8
1.13	Delineation of the research	9
1.14	Headline findings	9
1.15	Conclusion	10
1.16	Contribution of the research	10
1.17	Summary	10
1.18	Outline of thesis structure	11

СНАРТИ	ER 2: LITERATURE REVIEW	.12
2.1	Introduction	.12
2.2	Background	.13
2.3	The definitions of small, medium and macro enterprises	.14
2.4	Small, medium and macro enterprises' contribution to the South African economy	16
2.5	The status of small, medium and macro Enterprises in South Africa	.17
2.6	The challenges facing SMMEs	. 18
2.7	Big data definitions	.19
2.8	Characteristics of BD solution for SMMEs	.24
2.8.1	Flexibility and choice	.24
2.8.2	Simplicity	.25
2.8.3	Cost	.25
2.9	SWOT analysis for BD	.25
2.9.1	Strength	.25
2.9.2	Weaknesses	.26
2.9.3	Opportunities	.26
2.9.4	Threats	.26
2.10	Innovation strategy to guide successful implementation of BD by SMMEs	.27
2.11	Big data strategic capabilities	.28
2.12	The role of BD in decision-making	.29
2.12.1	Structured decision making	.30
2.12.2	Unstructured decision making	.31
2.13	The role of BD in customer segmentation	. 31
2.14	The role of BD in innovation	. 32
2.15	Big data and cloud computing for SMMEs	. 33
2.16	Big data analytics	.36
2.17	The competitive advantage of using BD analytics	. 37
2.18	Big data management	. 38
2.19	Big data management tools	. 39
2.20	Big data and SMMEs	.40
2.21	Adoption of BD by SMMEs	.41
2.21.1	Introduction	.41
2.21.2	Technology Organisation Environment (TOE) Framework	.41
2.21.3	Technology Acceptance Model (TAM)	.42
2.21.4	Diffusion of Innovation Theory (DOI)	.42
2.22	Summary	.42
СНАРТИ	ER THREE: RESEARCH METHODOLOGY	.44
3.1	Introduction	.44

3.2	Research philosophy	45
3.2.1	Introduction	45
3.2.2	Ontology	46
3.2.2.1	Introduction	46
3.2.2.2	Objectivism	46
3.2.2.3	Subjectivism	46
3.2.3	Epistemology	47
3.2.3.1	Introduction	47
3.2.3.2	Interpretivism	47
3.2.3.3	Positivism	48
3.2.3.4	Critical realism	49
3.3	Research approach	50
3.3.1	Introduction	50
3.3.2	Deductive approach	51
3.3.3	Abductive approach	51
3.3.4	Inductive approach	51
3.4	Research strategy	52
3.4.1	Case study	52
3.4.2	Sampling	52
3.4.3	Unit of analysis with the aim to explore	53
3.4.4	Unit of observation	53
3.5	Data collection	53
3.5.1	Introduction	53
3.5.2	Permission	54
3.5.3	Interview process	54
3.6	Data analysis	54
3.7	Ethical considerations	54
3.7	Chapter summary	55
CHAPT	ER FOUR: ANALYSIS AND FINDINGS	57
4.1	Introduction	57
4.2	Problem statement	58
4.3	Research questions	58
4.4	Aim of the study	58
4.5	Research objectives	58
4.6	Cases of the study	58
4.7	The participants	60
4.8	Analysis	61
4.9	Findings	62

4.9.1	Research question 1	63
4.9.1.1	RSQ 1.1: What are the perceptions of FSPs towards the value of BD for their	
	businesses?	63
4.9.1.2	RSQ 1.2: What are the factors affecting FSPs when adopting BD analytics for	
	creating a competitive advantage?	72
4.9.2	Research Question 2	75
4.9.2.1	RSQ 2.1: How do FSPs determine the needs of their customers?	75
4.9.2.2	RSQ 2.2: How can FSPs incorporate BD in their business strategies?	76
4.10	Summary of findings	78
4.10.1	Findings RQ1:RSQ 1.1	78
4.10.2	Findings RQ1: RSQ 1.2	79
4.10.3	Findings RQ2: RSQ 2.1	79
4.10.4	Findings RQ 2: RSQ 2.2	80
4.11	Headline findings	80
4.12	Categories and themes development	80
4.13	Themes	83
4.14	Chapter summary	83
CHAPT	ER FIVE: DISCUSSION	85
5.1	Introduction	85
5.2	Theme 1: Management	85
5.3	Theme 2: Training	88
5.4	Theme 3: Data Management	91
5.5	Theme 4: Finance	94
5.6	Theme 5: Marketing	95
5.6.1	Volume	97
5.6.2	Velocity	97
5.6.3	Variety	97
5.7	Chapter Summary	98
CHAPT	ER SIX: CONCLUSIONS AND RECOMMENDATIONS	99
6.1	Introduction	99
6.2	Conclusion	99
6.3	Research questions answered	. 100
6.3.1	RQ1: What are the challenges FSPs face when using BD analytics in building an	nd
	differentiation customer segmentation to create a competitive advantage?	.100
6.3.2	RQ2: How can FSPs utilise BD analytics in order to create a competitive	
	advantage?	.100
6.4	Aim of the study	. 101
6.5	Recommendations	.101

6.6	Reflection	101
6.7	Future research	102
REFE	RENCES	103
APPE	ENDIX A: INTERVIEW GUIDE QUESTIONS	126
APPE	NDIX B: CONSENT LETTER	128
Appe	ndix D1-D14: INTERVIEW TRANSCRIPTION P1-15	132
APPE	NDIX D1: INTERVIEW TRANSCRIPTION P1	132
APPE	NDIX D2: INTERVIEW TRANSCRIPTION P2	138
APPE	NDIX D3: INTERVIEW TRANSCRIPTION P3	142
APPE	NDIX D4: INTERVIEW TRANSCRIPTION P4	146
APPE	NDIX D5: INTERVIEW TRANSCRIPTION P5	149
APPE	NDIX D6 AND D7: INTERVIEW TRANSCRIPTION P6 AND P7	153
APPE	NDIX D8: INTERVIEW TRANSCRIPTION P8	158
APPE	NDIX D9: INTERVIEW TRANSCRIPTION P9	163
APPE	NDIX D10: INTERVIEW TRANSCRIPTION P10	167
APPE	NDIX D11: INTERVIEW TRANSCRIPTION P11	171
APPE	NDIX D12: INTERVIEW TRANSCRIPTION P12	173
APPE	NDIX D13: INTERVIEW TRANSCRIPTION P13	179
APPE	NDIX D14: INTERVIEW TRANSCRIPTION P14	
APPE	NDIX D15: INTERVIEW TRANSCRIPTION P15	

LIST OF FIGURES

Figure 1.1: Layout of Chapter One	1
Figure 2.1: Layout of Chapter Two	12
Figure 2.2: Number of documents reported on BD	13
Figure 2.3: Summary of the "5Vs" of BD	21
Figure 2.4: BD Value Chain	23
Figure 2.5: Factors that influence SMMEs to adopt cloud computing	35
Figure 3.1: Layout of Chapter 3	44
Figure 3.2: Four paradigms of analysis of social theory	46
Figure 4.1: Layout of Chapter 4	57
Figure 4.2: Creating a summary from the responses	62
Figure 5.1: Layout of Chapter Five	85
Figure 5.2: Sources of BD	92
Figure 5.3: BD technology architecture	93
Figure 6.1: Layout of Chapter Six	99

LIST OF TABLES

Table 2.1: Average breakdown for SMMEs according to size or class	16
Table 2.2: Some opportunities and challenges faced by SMMEs when implementing	
innovation strategies for BD	27
Table 2.3: Key success factors	33
Table 2.4: The main benefits of cloud computing for SMMEs	35
Table 3.1: The principles of interpretive research in the study	48
Table 3.2: Illustrates positivist approach versus interpretive approach	49
Table 3.3: Deduction, induction and abduction: From reason to research	51
Table 4.1: Description of the FSPs who participated in the study	59
Table 4.2: Participant's role, number of years' experience and highest level of education	60
Table 4.3: An example of a transcribed interview	61
Table 4.4: Findings for RSQ 1.1	78
Table 4.5: Findings for RSQ 1.2	79
Table 4.6: Findings for RSQ 2.1	79
Table 4.7: The findings for RSQ 2.2	79
Table 4.8: The findings for RSQ 2.2	80
Table 4.9: Summary of RSQs, 28 findings (28), categories (11) and themes (5)	80
Table 4.10: Represents the link between research questions and the themes developed.	83

ABBREVIATIONS

Abbreviation	Full word / term
BD	Big data
FSPs	Financial service providers
BDA	Big data analytics
SMMEs	Small, medium and macro enterprises
NSBASA	The National Small Business Act of South Africa
NSB Act: 14	The National Small Business Amendment Act of 2003 and 2004
ICT	Information and communications technology
RQ	Research question
RSQ	Research sub-question
CEO	Chief Executive Officer
CFO	Chief Financial Officer
SWOT	Strengths, weaknesses, opportunities, threats
IEEE	Institute of Electrical and Electronics Engineers
CPUT	Cape Peninsula University of Technology
ERP	Enterprise resource planning
CRM	Customer relationship management
SCM	Supply chain management
Three Vs	Velocity, variety and volume
GDP	Gross Domestic Product
SA	South Africa
5Vs	Volume, velocity, value, variety and veracity
IT	Information technology
CDR	Call detail records
KM	Knowledge management
BDM	Big data management
TOE	Technology organisation Environment Framework
TAM	Technology Acceptance Model
DOI	Diffusion of Innovation
NIST	The National Institute of Standards and Technology

DEFINITIONS AND KEY CONCEPTS

Definitions and key concepts	References
Accounting is defined as "a system to document all the information about the business transactions and events".	Chen et al. (2020:1197)
Big Data is defined as a "large and complex data assets that require cost effective management and analysis for extraction of insights from them".	Tabesh et al. (2019:348)
Big Data is defined as "data that exceeds the processing capacity of conventional database systems"	Santos and Costa (2020:13)
Big Data is also define as a "large data sets that are often challenging to examine and investigate due to their complexity and variability".	Gupta et al. (2020:949)
Big Data management is defined as "a vital requirement to secure data while also allowing users to query and retrieve the data.	Alhajj et al. (2020:05)
Big Data management can be defined as "the organisation, administration, storage and management of large volumes of structured and unstructured data sets".	Ghavami (2020:04)
Cloud Computing is defined as a "paradigm that enables utility computing, for example the leasing of computing resources in real time, with minimal interaction with the provider".	Aceto et al. (2020:03)

Cloud computing is a "model in which computer processing, storage software, software and other services are supplied as a set of virtualized resources over the internet". The National Institute of Standards and Technology (NIST) has defined cloud computing as a "model for enabling ubiquitous, convenient, on demand network access to a shared pool of configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction" Customer segmentation is defined as a "process of dividing a Munusamy and Murugesan
storage software, software and other services are supplied as a set of virtualized resources over the internet". The National Institute of Standards and Technology (NIST) has defined cloud computing as a "model for enabling ubiquitous, convenient, on demand network access to a shared pool of configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction" Customer segmentation is defined as a "process of dividing a Munusamy and Murugesan
set of virtualized resources over the internet".The National Institute of Standards and Technology (NIST) has defined cloud computing as a "model for enabling ubiquitous, convenient, on demand network access to a shared pool of configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction"Yamin and Makrami (2015:854)Customer segmentation is defined as a "process of dividing a Munusamy and MurugesanMunusamy and Murugesan
The National Institute of Standards and Technology (NIST) has defined cloud computing as a "model for enabling ubiquitous, convenient, on demand network access to a shared pool of configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction" Customer segmentation is defined as a "process of dividing a Munusamy and Murugesan
defined cloud computing as a "model for enabling ubiquitous, convenient, on demand network access to a shared pool of configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction"
convenient, on demand network access to a shared pool of configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction"
configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction"
storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction"
provisioned and released with minimal management effort or service provider interaction"
service provider interaction" Customer segmentation is defined as a "process of dividing a Munusamy and Murugesan
Customer segmentation is defined as a "process of dividing a Munusamy and Murugesan
edeterner beginerkaller ib dernied de a probebe er arraing a manadamy and maragedan
broad customer base into manageable sub-groups of customers" (2020:1922)
Data Management is defined as "a vital requirement to secure Alhajj et al. (2020:05)
data while also allowing users to query and retrieve the data".
Data Management is also a "reality that represents a set of Nda and Tasmin (2019:5009)
challenges involving Big Data modelling, storage, retrieval
analysis and visualisation for several areas in organisations".
Finance is defined as "ways of breaking down physical basis of Purcell et al. (2020:444)
earnings that become central to securitised flows of funds
generated from different asset classes".
Information Technology refers to "the idea that IT assets might Pare et al. (2020:01)
not be viewed as strategic by top executive's goals and
implementing its strategies".
Segmentation is defined as the "process of dividing the total Thoeni et al. (2016:2195)
market into a number of smaller, more homogeneous submarkets
and termed market segments"
Small, Makro, and Medium Enterprises are defined according Joubert et al. (1999:24)
The National Small Business Act of South Africa (NSBASA) as
"distinct business entities which are not part of a group of
companies
Small, Makro, and Medium Enterprises are defined as "separate Section 1 of the National Small
and distinct business entity, including co-operative enterprises Business Act of 1996 as amended
and nongovernmental organisations, managed by one owner or by the National Small Business
more which including its branches or subsidiaries, if any is Amendment Acts of 2003 and 2004
predominancy carried on in any sector of sub sector of the (NSD Act. 14)
Small Makro and Modium Enterprises are defined as a "actution Moleke (2012:12)
to attain economic development from establishing traditional
family husinesses employing over a hundred people with the aim
of reducing poverty in the developing economies"

CHAPTER ONE: INTRODUCTION





1.1 Introduction

Big Data (BD) increases challenges and opportunities for Small, Macro and Medium size enterprises (SMMEs), information technology (IT), and research communities (Wang & Wang, 2020). By leveraging BD, SMMEs can grow its organisational agility, adaptability and resilience to overcome the competitive environment by becoming a valuable and innovative organisation (Han & Trimi, 2022). In a complex business environment, agility is a crucial success factor, since it provides innovative and a competitive advantage (Vahid et al., 2022). According to Vahid et al. (2022) agility is important for firms operating in different contexts with

multiple characteristics. BD is forcing organisations to change their business models, strategies and management decision practices (Fernandez-Vidal et al., 2022).

According to Santos and Costa (2020), there are new technical model changes on how organisations are collecting, storing, processing and analysing data. SMMEs need to be aware of these new technological models that can improve their business value. BD is key to any business to make decisions and offers both opportunities and challenges (Ghavami, 2020). According to Ghavami (2020) BD is well known by its three key elements as the three Vs which are: (i) volume, (ii) velocity and (iii) variety. Volume is the data size or quantity of data, velocity is the speed and frequency of processing data and variety is the type of data that is being analysed (Ghasemaghaei & Calic, 2020). BD can change the FSPs innovation's ability by effectively and efficiently increasing the relevance between customer preferences and product offering (Ghasemaghaei & Calic, 2020).

BD is arguable difficult to define because of determining the exact point where data is viewed as big. In many cases BD is often defined by its core elements (Santos & Costa, 2020). Tabesh et al. (2019:348) define BD as a "large and complex data assets that require cost effective management and analysis for extraction of insights from them". Santos and Costa (2020:13) defined BD based on the technological restrictions as "data that exceeds the processing capacity of conventional database systems". BD is also define by Gupta et al. (2020:949) as a "large data sets that are often challenging to examine and investigate due to their complexity and variability". For the purpose of this research, BD is defined as data sets (volume, variety & velocity) that are so large or complex than normal analytical or processing capability aimed to assist SMMEs to gain a competitive advantage and in decision making cannot be done by standard analytical tools.

SMMEs are defined according The National Small Business Act of South Africa (NSBASA) as "distinct business entities which are not part of a group of companies" (Joubert et al., 1999:24). Like any other country the definition of SMMEs in South Africa is of a major concern with various researchers giving it different definitions (Kamble et al., 2020). According to Kamble et al. (2020) SMMEs are defined in different ways through either their number of employees, their turnover bands or through the combination of both as stated in the National Small Business Act 1996. In the South African perspective, the SMME sectors is made up of small businesses that are set up from macro sized enterprises such as family traditional businesses employing 50 people or less with a turnover of less than 10 million rand (Cant, 2020).

According to Khoase et al. (2020) South African macro enterprises employ between 11 and 50 employees per enterprise. South African medium enterprises employ between 51 and 250 employees. Micro enterprises in South Africa employ between 1 and 10 employees. In South Africa, SMMEs are defined in Section 1 of the National Small Business Act of 1996 as amended by the National Small Business Amendment Acts of 2003 and 2004 (NSB Act:14) as a "separate and distinct business entity, including co-operative enterprises and

nongovernmental organisations, managed by one owner or more which including its branches or subsidiaries, if any is predominantly carried on in any sector or sub sector of the economy". For the purpose of this research, SMMEs are defined through its combination of both the number of employees (50 people or less) and the turnover band (less than 10 million rand). Some organisations are benefiting from their stored data, having successfully implemented competencies to utilise and benefit from these large structured and unstructured data sets. These organisations have launched initiatives to complement their analytical proficiencies (Ogbuokiri et al., 2015; Akhtar et al., 2019). To date, most attention regarding BD focusses on large enterprises. The diffusion of low-cost data producing devices (e.g. Cell phones and internet-enabled cameras), easy access to very large data systems, and the spread of data analytics expertise mean that BD analytics is a reality, fundamental and available to business for decision making (Carbonell, 2016; Ghavami, 2020).

With the progressive globalisation of the economy, many organisations are investing in BD to discover innovative techniques to differentiate themselves from their competitors (Ghasemaghaei & Calic, 2020). According to Rialti and Marzi (2020) an organisation is innovative when it has an ability to exploit the current market opportunities and at the same time investigating new opportunities. When determining the best strategy for a company it is essential that the data the company collects is properly analysed. Due to the arising potential of BD, SMMEs need to start investing on systems to transform large data sets into information that can improved performance and decision making (Rialti & Marzi, 2020).

In this research focus is placed on the use of BD within SMMEs with special reference to Financial Service Providers (FSPs). FSPs are used as an example of SMMEs (Chapter One section 1.2; Chapter Three, section 3.4.2). FSPs were chosen as case studies since the Financial Advisory and Intermediary Services Act 37 (2002) is forcing FSPs to manage all written voice and other communication by means of Information and Communication Technology (ICT). This Act are creating huge volumes of data for the FSPs that need to be securely stored, available on demand and analysed.

1.2 Background to problem statement

People, processes, data, and objects have been grouped together to make networked connections more relevant and valuable by transforming information which creates new experiences, wide opportunities, growth, innovation and customer relationship for SMMEs (Abazi, 2016; Santos & Costa, 2020). In South Africa there is a need for SMMEs to be innovative, create new business opportunities and business models, to be competitive since most corporates have already adopted digitisation strategies in their quest to create a competitive advantage in the industry they are functioning in. According to Abu-Salih et al. (2021) BD has the potential to support SMMEs in making intelligent decisions based on evidence rather than intuition for their businesses.

ICT plays a vital role in the growth and development of SMMEs towards participating in knowledge management by facilitating connectivity that assists them to create and deliver products on a global scale (Rehman et al., 2016; Marnewick & Bvuma, 2020). According to Kossaï and Piget (2014), and Walaza et al. (2020) ICT is identified as a major contributor towards the economy in developed countries, and can be utilised to increase the economic growth and decrease poverty. However, there is limited knowledge on BD and SMMEs in developing countries. BD is creating opportunities for current and future research topics for example on SMMEs to gain the strategic competitive advantage since large companies have already invested in leveraging it (Matt et al., 2020).

Globalisation and driven interest of SMMEs into BD orchestrate SMMEs to develop and utilise BDA to bring technological innovation to them (Saleem et al., 2020). Even though SMMEs may not have BD, they can still insource BD from other organisations. Once insourced they can use the BD sets to seek for innovative opportunities, new business opportunities, and intelligent business decision making. BD is well known, but the majority of people are thinking it only exist and benefit large organisations. However, there are some sets of BD that exists for SMMEs and they can gain useful advantages which can improve SMME's performance (Alshahri, 2016; Saleem et al., 2020).

1.3 Statement of the research problem

SMMEs do not utilise innovation to improve performance, contribute to their competitiveness and to grant them access to international markets (Matekenya & Moyo, 2022). BizConnect (2014:02) indicates that on average "50 percent of all start-up businesses fail within 24 months". The lack of innovative skills, experience, finance, new market development, planning as well as the inability to implement innovative ideas are key contributors to start-up business failures.

With the new stream of technologies, FSPs are facing challenges on how to utilise their existing data in creating new innovative products or enhancing the existing services. According to Saleem et al. (2020), there is a need for SMMEs to re-evaluate BD strategies by reviewing their competitive landscape of products. If SMMEs and more specifically FSPs do not perform this exercise they can lose their competitive advantage, sales momentum and market share. SMMEs can utilise BDA to create a clear picture of their customers and their demands so that they can make well informed decisions when creating marketing strategies (Nasrollahi et al., 2021).

According to Matt et al., (2020), the global economy is a strong competitor of SMMEs. SMMEs are required to establish new types of innovative and digital strategies in order for them to maintain their competitive advantage. BD can assist SMMEs to increase their agility, adaptability and competitive advantage (Han & Trimi, 2022). BD not just bring opportunities for FSPs, but also some challenges that need to be addressed. SMMEs are facing challenges

in acquiring BD knowledge, in creating and implementing BD strategies, in analysing their data and cannot afford to purchase sophisticated computer systems to perform analysis (O'Connor & Kelly, 2017; Liu et al., 2020). BD is made up of enormous datasets that exceeds the collection, management, use and processing capabilities of human beings in an appropriate time (Chuah & Thurusamry, 2021). FSPs do not use BD to differentiate their customer segments and are losing their competitive advantage resulting in a negative impact on market share and potential profit gains.

1.4 Research questions

The research questions and research sub-questions are presented in Table 1.1.

Table 1.1: Research problem, research questions (RQs), research sub-questions (RSQs), methodology	and
objectives	

Research problem	FSPs do not use BD to differentiate their customer segments and are losing their competitive advantage resulting in market share as well as potential profit loses.		
RQ1	What are the challenges FSPs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?		
Research sub-questions	Research method(s)	Objectives	
RSQ 1.1: What are the perceptions of FSPs towards the value of BD for their businesses?	Interviews by means of semi- structured questionnaires.	Determine if FSPs sees the value of BD for their businesses.	
RSQ 1.2: What are the factors affecting FSPs when adopting BD analytics for creating a competitive advantage?	Interviews by means of semi- structured questionnaires.	Examine the way FSPs are measuring their success, is it: Products/services. Sentiment (Net promoter score [NPS]) Revenue	
RQ2	How can FSPs utilise BD analytics in order to create a competitive advantage?		
Research sub-questions	Research method(s)	Objectives	
RSQ 2.1: How do FSPs determine the needs of their customers?	Interviews by means of semi- structured questionnaires	Determine FSPs strategies to meet their customer's needs.	
RSQ 2.2: How can FSPs incorporate BD in their business strategies?	Interviews by means of semi- structured questionnaires.	 Identify any challenges faced by FSPs to include BD strategy to their businesses, is it because of: Lack of innovative ideas Lack of technology expertise ROI too low / costs too high Lack of expertise with a digital business model 	

*RSQ = Research sub-question

1.5 Research aim

The aim of this study is to explore how FSPs can use BD to differentiate their customer segments and create a competitive advantage.

1.6 Objectives

- i) Determine if FSPs see the value of BD for their businesses.
- ii) Identify any challenges faced by FSPs to include BD strategy to their businesses.
- iii) Identify FSPs strategy to meet their customer's needs.
- iv) Examine the way FSPs are measuring their success.

1.7 Research methodology

1.7.1 Research philosophy

As stated by Holden and Lynch (2019:401), research philosophy consists of two main approaches namely: i) ontology and ii) epistemology. Research philosophy has been defined by Saunders et al. (2009:127) as "a technique a researcher look on the world by developing knowledge and determining the nature of that knowledge". Research philosophy empowers the researcher to determine an approach, to comprehend the reason behind the choice, and to choose the research strategy and methods of that strategy (Tashakkori & Creswell, 2007). Methodology is used for solving problems systematically, and provides a scientific investigation on how research should be conducted which also include all procedures used by the researcher to perform specific tasks and processes for the research (Myers, 2013). In the next sub-section ontology and epistemology is discussed and the philosophical stance of the research declared.

1.7.1.1 Ontology

Ontology has been defined by Zhu et al. (2015:27) as an "idea invented from the philosophy which mainly describes the existing knowledge". Ontology is anxious on how we view the reality and the assumptions behind that reality. The ontological assumption is what establish the reality, which means researchers must take a stand regarding how things are in reality and on how things really work (Scotland, 2012). Ontology can be used by researchers to solve problems related to data analysis, data search and data integration (Yilmaz et al., 2018). Ontology is mainly divided into objectivism and subjectivism (Chapter 3, section 3.2.3). The ontology of this research was based on a subjectivism stance.

1.7.1.2 Epistemology

Epistemology is the nature and the method of knowledge creation. According Scotland (2012:09), the epistemology assumptions are "concerned on how to create the knowledge, gather the knowledge and also communicate the gathered knowledge". Epistemology's concern is on how researchers go about in attaining knowledge around the world (Mati, 2016). There are mainly three epistemological types used to conduct a research that have been mentioned by Rahman (2022:40) namely: i) Positivism, ii) Interpretivist and iii) Critical Realism. (Chapter 3, section 3.2.2).

This research was based on an interpretivist paradigm because the research was conducted among people rather than the objects and the researcher acknowledged the different views from interviewees.

1.8 Research approach

The research approach to theory development can be deductive, abductive or inductive (Saunders et al., 2019). Aljaroodi et al. (2020:3) state that "the deductive approach is a top down approach which means it's from theory to data, and the inductive approach is a bottom up method which means theory is created by finding relationships between the themes that develop from participant's opinions". Currently little is known on how SMMEs are utilising BD, what challenges they are facing and which factors should be consider when considering BD strategies and analytics (Chong et al., 2015; Verma & Bhattacharyya, 2017). As a result and in an attempt to contribute towards the body of knowledge an inductive approach was followed (Chapter 3, section 3.3).

1.9 Research strategy

1.9.1 Case study

A case study involves the collection of data by means of interviews, questionnaires, observation and archives, generating qualitative and quantitative data (Yin, 2003). A multiple case study strategy (Saunders et al., 2019) was chosen as the research strategy as the research aims to explore how FSPs can use BD to differentiate their customer segments and create a competitive advantage. The main criteria for the selection of a multiple case study was FSPs with a formal structure, and with less than 50 employees.

1.9.2 Unit of analysis

The unit of analysis was the 15 FSPs (Chapter 3, section 3.4.3, Chapter 4, section 4.3) in the Western Cape and Gauteng provinces of South Africa. FSPs from other provinces were not included due to time and financial constraints.

1.9.3 Unit of observation

The unit of observation was the FSP's decision makers' i.e. two (2) Chief Executive Officer (CEO), one (1) Head of Risk Compliance, one (1) Chief Financial Officer (CFO), two (2) Managing Directors, one (1) Fund Manager, one (1) Financial Advisor Category 2, two (2) Chief Operating Officers, one (1) Head of business / Chief Operating Officer, two (2) Owners and key Individuals, one (1) IT Manager and one (1) Principal Owner of the business.

1.9.4 Sampling

Ruxwana et al. (2010:17) describe a sample as "a process whereby certain part of the population is selected by the researcher using a specific technique to produce knowledge about the population and also to uncover multiple realities about such a population". When

targeting a diverse population using a correct sampling strategy is very necessary. South African FSPs with a formal structure with a potential of using BDA as a strategic tool in making decisions were purposively, non-random and convenient sampled. Snowballing was also used to increase the number of FSPs for the study.

1.10 Data collection

Data collection was done using interviews with an interview guide (Appendix A) with semistructured questions. Interviews is a qualitative research method that requires conducting interview with a small number of participants (Jamshed, 2017). This method normally provides context to other data offering a complete view of what happened and why. The interviews were recorded with the permission of the participant.

1.11 Data analysis

There are two data analytical techniques, namely: i) qualitative and ii) quantitative techniques (Chong et al., 2015 Verma & Bhattacharyya, 2017). This research made use of qualitative data analytical techniques. Content and thematic analysis are the two frequently used methods when performing qualitative data analysis (Vaismoradi et al., 2013). The data was analysed by means of a thematic analysis (White et al., 2017). The recorded interviews were transcribed and mailed back to the participant in order to validate the transcriptions. Once the data had been validated, it was coded, recoded and then summarised, categorised and thematic developed.

1.12 Ethics

The current study adhered with the CPUT research ethic policy. The compliance certificate was provided by the University Ethics Committee, before the researcher was given a go ahead to start the research. All the participants agree to participate in the research study by signing the consent letter, and each participant was guaranteed the confidentiality and the right to withdraw from the research at any given time if needed to do so. Data and all the information provided by the participants were kept confidential and have been protected.

Cooper and Schindler (2003:56) define ethics as "rules or behaviour standards which distinguish and guide moral principles that govern a person's behaviour and the relationships with other people". According to Myers (2013:34) principles of ethics consist of the following elements: i) honesty, ii) plagiarism, iii) informed consent and iv) permission to publish. Resnik (2015:2) further defines the principles of ethics as:

Honestly: All the findings must be reported by the researcher honestly without any made-up data

Integrity: All promises and agreement made to the interviewees or participant must be followed *Openness:* A researchers must be open to the criticism or any ideas raised by the participants

Confidentiality: The data must be protected and treated confidentially at all times by the researcher

The study also considered several ethical issues and the researcher made sure that all questions asked don't offend the participant in any way. All appointments for the qualitative interviews were made in advance and were used to get informed consent. As stated by Welman et al., (2005) the researcher needs to pay attention to the following ethical consideration:

Permissions: acquiring permission from the participant to carry out the research was an essential requirement. The researcher used an email to send out invitations to all the FSPs (Appendix B) and in most of the times physical visits to the FSPs was done. Electronic follow-ups via the use of email were done before interviews took place.

The researcher assured all the participants about the rights of their privacy. During the interview process all the participants were asked to freely accept and signed the consent letter to participate of the research. All the participants signed a letter of consent to participate in the research.

Assurance, unethical tactics and techniques were not used by the researcher during interviewing. In this study, a semi-structured questionnaire was used for data collection, all interviews were guided by an interview guide (Appendix A). Some participants requested interview questions prior to interview in order to prepare for the interview. All interviews were recorded by the researcher, and participants were asked upfront before the interview starts to grant permission to record the interview. The records taken from the interviews were transcribed and texted analysed for additional analysis using inductive approach.

1.13 Delineation of the research

The following were excluded from the study:

- All FSPs without formal structure where there had been a potential of using BDA as a strategic tool in making decisions.
- ii) All FSPs operating from other provinces beside Western Cape Province or Gauteng Province.
- iii) FSPs that are not categorise as SMMEs for example large corporates.
- iv) FSP that are not registered under the Financial Institution.

1.14 Headline findings

Twenty-eight (28) findings were derived from the interviews and seven (7) headline findings were identified as listed below.

Headline finding 1: Most of the FSPs are aware of BD.

Headline finding 2: FSPs do not have the knowledge and experience to utilise BD for their organisations.

Headline finding 3:	FSPs lack time and reliable resources, cost, skills, and collecting			
	reliable data as a major constraints to implement BD.			
Headline finding 4:	FSPs have challenges in adopting BD.			
Headline finding 5:	FSPs are seeing benefits in BD to create competitive advantages for			
	their organisations.			
Headline finding 6:	FSPs mentioned that BDA can be used to identify gaps, assist in terms			
	of identifying customers' needs.			
Headline finding 7:	FSPs do not have a strategy for BD.			

1.15 Conclusion

BD can be utilised by FSPs to create various values, such as:

- i) Creating transparency by simple making BD more easily accessible to all the stakeholders can create tremendous value. FSPs can utilise BD to improve quality.
- ii) Enabling experimentation to discover needs, expose variability and improve performance – While creating and storing more transactional data, FSPs can collect more accurate and detail performance data from their product inventories. Use the data to analyse their performance and also manage their current performance to higher levels.
- iii) Innovating new business models, products and services BD will enables FSPs to create new products and services. FSPs can either enhance the existing products and services or invent a new business models.
- iv) Segmenting their customers BD will allow FSPs to create highly specialised segmentations and tailor their products and service to meet those customer's needs.
- v) FSPs should gain fundamental advantage from BD to make better decisions and explore new opportunities.

1.16 Contribution of the research

The research contributed to the general body of knowledge by providing guidelines for South African's FSPs on the effective usage of BD to differentiate their customer segments and also create competitive advantage.

1.17 Summary

BD has drawn the attention of the ICT world as many organisations collect and store more data than ever before as their businesses depend on it. FSPs can embrace BD and FSPs need to start looking at using BDA so that they can be able to analyse the needs of their customers. The aim of this study is to explore how FSPs can use BD to differentiate their customer segments and to create a competitive advantage.

The study purposefully and conveniently selected South African FSPs (15) in the Gauteng Western Cape and Gauteng Provinces. These FSPs have formal structure, and the potential of using BDA as a strategic tool in making decisions. One person per FSP was purposefully and conveniently selected as unit of observation.

Data was collected by means of interviews using semi-structured questionnaires. All the data were summarised and categorised and there after a thematic analysis was conducted. Ethics were observed in accordance with the Cape Peninsula University of Technology ethics governance, policies and procedures.

1.18 Outline of thesis structure

Chapter One addresses the introduction to the research as well as the research problem, background to problem statement, statement of the research problem, aim of study, research objectives, research methodology, research strategy, contribution of the research and chapter summary.

Chapter Two addresses the literature review which covers SMMEs definitions, SMME's contribution to the South African economy, the status of SMMEs in South Africa, the challenges facing SMMEs, BD definitions, Dimensions of BD, characteristics of BD solution for SMMEs, SWOT analysis for BD, innovation strategy to guide successful implementation of BD by SMMEs, strategic capabilities, the role of BD in decision-making, the role of BD in customer segmentation, the role of BD in innovation, BD and Cloud Computing for SMMEs, BD analytics, the competitive advantage of using BD analytics, BD management, BD and SMMEs, adoption of BD by SMMEs using theories of a new technology framework and chapter summary.

Chapter Three elaborates on the research philosophy, approach, strategy, aim, data collection, data analysis, ethical consideration and chapter summary.

Chapter Four reports on the data analysis and summarised findings following the research questions.

Chapter Five addresses the themes of the study with regard the research questions.

Chapter Six provides the conclusions, recommendations and future research.

The next Chapter (2) presents the literature review.

CHAPTER 2: LITERATURE REVIEW



Figure 0.1: Layout of Chapter Two

2.1 Introduction

The literature review is a substantial part of any academic studies because it assists in providing arguments and justification about the phenomena being studied (Ho, 2016). Thus, this review used the key concept derived from the problem statement, research questions and aim of this study, which was to explore how FSPs can use BD to differentiate their customer segments and create a competitive advantage.

This chapter presents an review of work which has been published related to this study. The key areas of the study within which the review was carried out include BD, SMMEs, BD

analytics; characteristics of BD, solution for SMMEs, and the use of BD in customer segmentation. Peer reviewed articles, reports, conference papers and articles were sourced from CPUT's subscribed databases libraries. Data bases such as ScienceDirect, Social Science journals, EBSCOhost, the Institute of Electrical and Electronics Engineers (IEEE) and Google Scholar were utilised to conduct the comprehensive literature study.

This chapter provides a summary on how BD can be useful for SMMEs. The Chapter is constructed as follows: i) definitions of SMMEs, ii) SMME's contributions to the South African economy, iii) BD definitions, iv) dimensions of BD, v) characteristics of BD solution for SMMEs, vi) SWOT Analysis of BD, vii) innovation strategy to guide successful implementation of BD by SMMEs, viii) strategic capabilities of BD, ix) the role of BD in decision-making (structured decision making and Unstructured decision making), x) the role of BD for customer segmentation, xi) BD in innovation, xii) BD and cloud computing for SMMEs, xiii) Big Data analytics, xiv) the competitive advantage of using BD analytics, xv) BD management, vxi) BD and SMMEs, xviii) adoption of BD by SMMEs and xix) summary.

2.2 Background

The term BD was created in 1997 by Cox and Ellsworth. Although research was slow on BD it exploded since 2011 (Figure 2.2) and has been one of the most trending topics in technological environments. BD has been considered as one of the breakthrough technological development of our times (Blasiak, 2014; Emrouznejad & Charles, 2019).



Figure 0.2: Number of documents reported on BD (Scopus, 2 July, 2022)

BD has great potential to revolutionise any area of knowledge and has an impact on any aspect of our lives (Emrouznejad & Charles, 2019). Large amounts of data have been generated continuously from social network platforms, blogs, sensor-enabled ecosystems such as Internet of Things, digital devices including smartphones, large scale enterprise systems such as Enterprise Resource Planning (ERP), Customer Relationship Management (CRM) and Supply Chain Management (SCM) (Bhadani & Jothimani, 2016; Baesens et al., 2018; Emrouznejad & Charles, 2019).

According to Dubey et al. (2019) BD is mentioned as a competitive powerful tool for any organisation to make better effective decisions and to improve operational performance. There

has been a focussed increase on BD from researchers because BD has an ability to influence any organisation from different sectors to become more competitive when utilised to its potential abilities. It is a big game charger in most contemporary industries (Ogbuokiri et al., 2015; Kumar et al., 2019). Some of these potential abilities of BD are to bring new business opportunities for an organisation, assist any organisation to satisfy their customer's needs and also help these organisations in creating innovative products to gain more revenue.

Volume, Variety, Variability, Value and Veracity have been mentioned by Schosser (2020:131) as the most commonly characteristics of BD. However, the three Vs: Velocity, Variety and Volume have been used in defining the BD concept (Ndambo, 2016; Schosser, 2020). Velocity being the speed used to produce and process data or rate of data generation, Volume the quantities of large data stored or the amount of data coming from traditional and non-traditional sources and Variety being a graphical image view of the available different data sets or the amount of data currently available (Ndambo, 2016; Schosser, 2020). Two basic features are used by Fuster and Scherrer (2016:10) to describe BD namely: "the ability to access, utilisation of huge amount of data so as to meet the conditions of 3V's and the technique to process and use data for unidentified patterns".

SMMEs can enhance their existing operating processes by starting to transmit all the data coming from different data sources into a single storage device, analyse it for making better decision in order to gain more profitable, satisfying and long lasting relationship with their customer (Silva et al., 2016; Zhang et al., 2020). Emrouznejad and Charles (2019) as well as as well as Sriram (2022) state that BD can be the most important powerful tool for an organisation in creating more business opportunities, making better and faster decisions and help them develop new products and service. SMMEs can use BD in making more informed decisions, provide better services to improve communities and also creates new opportunities for business avenues (Abdulla, 2015; Emrouznejad & Charles, 2019). SMMEs can lower their cost of collecting data if they buy the existing data from other organisations and use it to achieve their business objective or if they can obtain their available data sets.

2.3 The definitions of small, medium and macro enterprises

There is no consensus on defining an SMME because researchers acknowledge that SMMEs are easier to describe than to define. SMMEs play an important role in both economic growth and employment opportunities (Mamman et al., 2015; Falahat et al., 2020). In order to describe the situation, it is important to explain what is meant by SMMEs. Numerous definitions for SMMEs are to be found in the literature. SMMEs are differently defined across countries and regions. SMMEs can be defined according to their size or structure of the economy. However, the main common parameters used to define SMMEs are staff headcounts, turnover and invested capital (Mamman et al., 2015; Thrassou et al., 2020). SMMEs are defined by Maloka (2013:12) as a "solution to attain economic development from

establishing traditional family businesses employing over a hundred people with the aim of reducing poverty in the developing economies". The National Act of South Africa (NSBASA) defines SMME as "a different trade entity which doesn't form part of the group of companies, but if there are any subsidiaries or branches for that particular SMME they must also be included when assessing the size" (Joubert et al., 1999). In South Africa SMMEs are defined as businesses that includes formally registered, informal and non-VAT registered organisations in which owners are fully involved in management and employ less than 200 employees (Strydom, 2017; Cant, 2020).

As in many other countries the definition of SMMEs in South Africa is a major concern with various researchers giving different myriad of SMMEs definitions (Kamble et al., 2020; Lesejane, 2021). According to Kamble et al.(2020), SMMEs are defined in different ways through either their number of employees, either their turnover bands or through the combination of both as stated in the National Small Business Act 1996. The definition of SMME through its size is very necessary but not sufficient when the researcher wants to understand a sector where the realities are not only common but also dynamic (Kamble et al., 2020). In South Africa, SMMEs are defined in Section 1 of the National Small Business Act of 1996 as amended by the National Small Business Amendment Acts of 2003 and 2004 (NSB Act:14) as a "separate and distinct business entity, including co-operative enterprises and nongovernmental organisations, managed by one owner or more which including its branches or subsidiaries, if any is predominantly carried on in any sector or sub sector of the economy". According to the National Small Business Act 102 of 1996 and its amendment of 2003, SMMEs are classified as either small, micro or medium enterprise when using a condition such as number of employees, annual turnover or total capital assets. For the purpose of this research, SMMEs are defined through its combination of both the number of employees (50) and the turnover band (less than R50m). SMMEs have been described on the White Paper on National Strategy for the Development and Promotion of Small Business in South Africa into four categories (Ladzani & Netswera, 2016; Moyo & Gumbo, 2021). In Table 2.1 the proposed categories and subsectors namely, agriculture, mining, manufacturing construction, retailing and transport are presented (Ladzani & Netswera, 2016; Moyo & Gumbo, 2021). Table 2.1 summarises the four categories of SMMEs.

Table 0.1: Average breakdown for SMMEs according to size or class (Ladzani & Netswera, 2016:227; Lesejane, 2021:16)

Size or Class	Description	Total Full-time equivalent of paid employees	Total Annual Turnover	Total Gross Asset Value
Medium	"Still mainly owner-managed, but decentralised management structure with the division of labour and operates from fixed premises with all formal requirements".	<200	<25.00 million	<8.00 million
Small	"More established than very small enterprise, formal and registered, fixed business premises and owner-managed, but more complex management structure".	<50	<10.00 million	<3.00 million
Very Small	"Part of the formal economy, use technology, less than ten paid employees and include self- employed artisans and professionals".	<10	<2.50 million	<0.70 million
Micro	"Usually the owner and family, formal, no license, formal business premises, labour legislation turnover below the VAT registration level of R300 000 per year, basic business skills and training, and potential to make the transition to a viable formal small business".	<5	<0.50 million	<0.10 million

In many developing countries such as South Africa, firms with more than 200 employees are considered large, whereas in Europe there is an upper limit of 99-499 for medium enterprises (Mamman et al., 2015; Moyo & Gumbo, 2021). Other countries in Africa, for example, Ghana consider firms with 6 to 9 employees as small enterprises, whereas Vietnam defines 50-100 employees as small enterprises (Mamman et al., 2015; Lesejane, 2021). In the next section the contribution of SMMEs to the South African economy is discussed.

2.4 Small, medium and macro enterprises' contribution to the South African economy

Small businesses worldwide have been understood as a main economy contributors for employment, creation of country's wealth and are considered as valuable assets for any economy (Strydom, 2017; Cant, 2020). Evidence from a study by Mbinda and Spencer (2016) and Cant (2020) on the risks connected to the work force of SMMEs indicate that SMMEs have an ability to address all the challenges of skills strategies to improve the quality of life and job creation which could assist in Gross Domestic Product (GDP) of the South African economy. According to Mantje and Rambe (2021), SMMEs are considered as the driving force in the South African economy because they add value to achieve socio-economic objectives such as poverty eradication and contributes to eradication of joblessness keeping national

unemployment rates low. In 2015, Statistics South Africa (Stats SA, 2015) has indicated that SMMEs are contributing roughly 42 percent to the South Africa's gross domestic product with 60 percent contribution to the employment rate. In 2018, the South African unemployment official rate stood at 28 percent (Rungani & Potgieter, 2018; Cant, 2020). According to Oji et al. (2017) and Lesejane (2021) SMMEs are fundamental to the economic growth because they have an unique abilities to create job opportunities and are notable as pillars for sustaining the economies of most countries.

Access to finance by SMMEs has been indicated by Ndayizigamiye and Khoase (2021) as major constrain. Some of these SMMEs are getting funding from families and friends because the majority of banks and financial institution are heavily charging them high interest rates with indemnity requirements. South African SMMEs are also being assisted by public and private supporting institutions to secure funding (Khoase et al., 2018; Maduku & Kaseeram, 2021). These institutions also provide strategic and operational support to the SMMEs during start-up and growth phases (Ndayizigamiye & Khoase, 2021).

In order for SMMEs to be able to achieve their primary business objective in sustaining economic growth to reduce unemployment, inequality and poverty, they need to be able to manage basic management activities namely: analysing, planning, implementing, and coordinating (Strydom, 2017; Maduku & Kaseeram, 2021). In the same manner, marketing activities are also required for SMMEs to be able to develop, build and maintain their products and services for customers (Nair & Ganesh, 2016; Steinhoff et al., 2019).

2.5 The status of small, medium and macro enterprises in South Africa

In order to emphasise the important role BD can play towards SMMEs in South Africa (SA), the current status of SMMEs in SA are presented. SMME's have been declared by the SA's finance minister as critical role players in both economic growth and employment opportunities (Falahat et al., 2020). According to Thrassou et al. (2020) the SA economy is losing jobs. SA has been reported with the world highest failure of SMMEs growth with a rate of 75 percent. The following has been identified by Troisi et al. (2018) as a major impediments for SA's SMMEs success:

- i) Financial (Internal factor)
- ii) Markets (External factor)
- iii) Management (Internal factor)
- iv) Economic (External factor)
- v) Infrastructure (External factor)

SMMEs are contributing 50 percent of the gross domestic product and are meant to address the 34.5 percent of unemployment disaster in SA (Falkena et al., 2021). Even though SMMEs have been regarded as a potential generators for job creation, the SA government means to provision the sector has been minimal for success (Taylor et al., 2018; Thrassou et al., 2020).

2.6 The challenges facing SMMEs

According to Falkena et al. (2021) SMMEs are declining in supplying services and goods in rural areas and townships. As a result large corporations started meeting the demands of customers call for high quality of products with value for money on pricing, greater convenience and providing a wide range of choices. This now resulted in gaining a competitive advantage for large organisations over SMMEs. SMMEs are struggling to play at larger scale and scope and they don't want to spend more since they consider spending as a waste (Steinhoff et al., 2019).

The following has been identified by Steinhoff et al. (2019:60) as the causes that lead to the failure of SA's SMMEs:

- i) Lack of basic skills and establishing an SMME for wrong reasons
- ii) Lack of adequate market
- iii) Lack of inadequate business plan
- iv) Lack of financial literacy and poor financial management
- v) Inability to secure funding

However, SMMEs do require much support from large organisations, government agencies and financial institutions in order for them to become more competitive (Khoase et al., 2018; Falkena et al., 2021).

According to Alsheikh et al. (2016:26) and Zhang et al. (2019:86) research conducted by Forrester conclude that the majority of business entities are only effectively utilising less than five percent of their existing data because of the following reasons: i) dealing with BD is costly, ii) to get more useful it must be collected from all different sources, and also iii) BD doesn't just contain only the old existing data but also involve any data currently growing at a fast rate. organisations can utilise their existing data to create valuable insights into business processes and some can implement technologies which can help them to resolve BDA to enhance organisation productivity. But, SMMEs are still in the process of getting solutions to compete better in their markets (Baskaran, 2019; Saleem et al., 2020). According to Baskaran (2019), organisational expectations of BD from their managers are to either implement or in improving their BDA techniques.

Most organisations are facing many challenges with the implementation of BD analytics of which some of these challenges are linked with the acceptance of BD by users as an important indicator for organisations to improve their productivity (Saleem et al., 2020). Other challenges are associated with cost in collecting data, storing the collected data and systems to decodify these extremely large datasets (Kache & Seuring, 2017; Rialti & Marzi, 2020).

Since the world is transforming into a growing and changing digital ecosystem, data has been indicated as new a commodity for any organisation to gain opportunities for new business innovation and create a clear picture of customer's demands to make well informed decisions in designing strategies (Dautov & Distefano, 2017; Nasrollahi et al., 2021). Technological

advancements are forcing SMMEs to look for better ways of doing business (Ikram et al., 2017; Nasrollahi et al., 2021). However, SMMEs are faced with challenges to invest in data management systems, some don't even have a qualified personnel to handle large quantities of data in order to attain valuable information and it is imperative that SMMEs receive all the necessary support for these challenges (Mattera, 2018; Marnewick & Bvuma, 2020). It is with no doubt SMMEs are operating in era of dynamic data coming from different resources such as social networks, online transactions, sensors, smart devices web pages, images, videos and personal browsing history and which is currently presented in different formats likely structured, semi structured and unstructured data format (Li et al., 2017; Gentsch, 2019). Technology and innovation has been pointed by Sen et al. (2016) and Marnewick and Bvuma (2020) as a most important strategy for SMME's growth and with BD recommended as a key drivers for it. SMMEs can use BD to increase productivity, responsiveness, anticipation and meeting customer's needs from making better decisions to gain competitive edge (Sen et al., 2016; Nasrollahi et al., 2021).

2.7 Big data definitions

Researchers across different industries see BD as core topic more specifically in computer science and information technology because it changes the way industries operate by providing opportunities for the organisation to collect, generate, analyse and process a huge amount of data (AI-Emadi et al., 2015; Nasrollahi et al., 2021). The phenomenon of large amounts of data is constantly increasing from dozens of terabytes of data (TB) to petabytes (PB) of data in single data set, hence BD analytical techniques need to be applied on BD sets (Elgendy & Elragal, 2016; Gentsch, 2019). The BD model is applied to all data which doesn't adapt to the normal structures of the existing database (AI-Emadi, Zakir & Seymour, 2015; Kalra & Ranjan, 2021). The following technologies for BD have been mentioned by Santos and Costa (2020:222) namely: i) Hadoop, ii) HDFS, iii) NoSQL, iv) MapReduce, v) MongoDB, vi) Cassandra, vii) PIG, viii) HIVE, and ix) HBase. The above-mentioned BD technologies work together in achieving the same end goal of getting the value from the existing unused data (AI-Emadi et al. 2015; Santos & Costa, 2020).

Tabesh et al.(2019:348) define BD as "large and complex data assets that require cost effective management and analysis for extraction of insights from them". Santos and Costa (2020:13) defined BD based on the technological restrictions as "data that exceeds the processing capacity of conventional database systems". BD is also define by Gupta et al. (2020:949) as "large data sets that are often challenging to examine and investigate due to their complexity and variability". BD has also been defined by Zhao-hong et al. (2018:202) as a "mass, high speed, data rich information asset that re quires efficient, new forms of information processing to enhance insight, make good decision and optimise processes".

For the purpose of this study, BD is defined as data sets that are so large or complex than normal analytical or processing capability aimed to help SMMEs to gain a competitive advantage and in decision making.

Scholars are classifying BD as the next big thing for innovation, the fourth model of science, the next cutting edge for innovation, competition and productivity, the next management uprising and are also identifying BD as the next revolution in the science and technology (Sen et al., 2016; Wu & Guan, 2018). SMMEs are required to influence their technical and management performance capabilities. This is done by constantly increasing their amount of data of which size and complexity is beyond their capability to manage, store and analyse it to retrieve value for the organisation (Shah et al., 2017; Santos & Costa, 2020).

Literature has many definitions of BD. Some researchers are using a V-model with five V's to describe BD others usually define BD using its three characteristics called 3Vs (Baaziz & Quoniam, 2013; Vajjhala & Ramollari, 2016; Schosser, 2020). According to Wang and Wang (2020) BD is distinguished by volume, velocity, variety and veracity. BD is an abstract theory because there is no single definition for it but different explanations about it are constantly stated by different research scholars, scientific and technological enterprises, data analysts and technical practitioners (Shah et al., 2017; Wu & Guan, 2018).

According to Saleem et al. (2020) SMMEs can utilise BD to transform their traditional business into a modern innovative business that can lead to adopting new methods and strategies to improve SMME's performance. Mining useful information from BD and then processing it can helps SMMEs to improve customer services from customer's purchase analysis, enhance marketing strategies and sustain long term risk analyses (Wang & Wang, 2020). South African SMMEs can utilise BD to achieve their objectives of gaining competitive advantage, satisfy their customer's needs or for better decision making organisation and have an ability to respond to customers quickly. This is possible if SMMEs adopt the V-model (Blasiak, 2014; Rahman & Aldhaban, 2015; Matt et al., 2020) as illustrated in Figure 2.2 below:



Figure 0.3: Summary of the "5Vs" of BD (Li, Chen & Shang, 2022:02)

The concepts in Figure 2.2 has been used by Chehbi-Gamoura et al. (2020) and Li et al. (2022) to summarise BD dimensions. The five characteristics of BD are defined as follows:

- i) Volume: refers to the quantity of data which is being collected or generated every second and it contains data from traditional and non-traditional sources (Charles & Gherman, 2013; Bhadani & Jothimani, 2016; Schosser, 2020). The generation of data has increased at a fast rate from being a terabytes to petabytes hence volume has form part of the five characteristics of BD (Bhadani & Jothimani, 2016; Alwan & Ku-Mahamud, 2020). According Giacalone and Scippacercola (2016) the estimated data that will be generated by 2020 will amount to 35 thousand billion gigabytes. As stated by Li et al. (2015) and Schosser (2020) volume refers to the large volumes of data being generated. Volume has also been defined by Nda and Tasmin (2019) as terrific amount of data which is regularly challenging to be stored, processed, analysed and presented.
- ii) Velocity: refers to the speed of purchase information generated and the flow of data into the FSPs sales journeys. SMMEs can analysis the growth of data coming to their organisation. Driven data marketing can be utilised by SMMEs to collect, analyse and process insights from BD and also drive customer engagement to improve marketing (Schosser, 2020). According Nda and Tasmin (2019) velocity can be referred as a growing rate at which data streams within an organisation.
- iii) Variety: refers to the amount of data currently available for SMMEs to utilise it for marketing. Marketing strategies can be implemented for both end user and data sources. SMMEs must combine both their transactional data and unstructured data coming from different sources so that marketing can use them effectively. For better customer engagement SMMEs need to connect both their analytics analysis and marketing communication features in return they will gain more profitable, satisfying and long lasting relationship with their customers (Zhang et al., 2020). It refers to the diverse data format which are both structured and unstructured (Nda & Tasmin, 2019; Walker & Brown, 2019).
- iv) Veracity: refers to the quality of generated data, which will bring value for SMMEs to solve their current problems (Marja, 2016; Mai et al., 2020). It is the fourth dimension on data, which discusses the trustworthiness of data because as numerous data source increases unreliability and inaccuracy also increases (Shah et al., 2017; Mai et al., 2020). It covers trust issues, security and uncertainty (Zhu et al., 2018).
- v) Value: offers greater and new vision to generate SMME's value (Rahman, 2016; Kalra & Ranjan, 2021). It refers to the data being used to generate insights value for the benefit of an organisation (Franková et al., 2016; Mai et al., 2020).

SMMEs can also utilise three major sources to collect their structured, unstructured and semistructured data, the three major sources are:

- Social data refers to all data collected from various social media platforms like Twitter, Facebook, Google and LinkedIn (Shinde et al., 2017; Emrouznejad & Charles, 2019).
- ii) **Machine data** to all data collected or created from enterprise planning, global positioning system and weblogs (Moon et al., 2017; Emrouznejad & Charles, 2019).
- iii) Transactional data refers to the data generated from various ecommerce systems like Amazon, Walmart, eBay and Takealot (Shalaginov et al., 2017; Emrouznejad & Charles, 2019).

Once the data has been stored, SMMEs must secured it before and after their analysis process and the structured, unstructured and semi-structured must be processed to transform into valuable insights for the business by means of categorising, summarising, matching-up and performing advanced functions and algorithms (Saleem et al., 2020). In order for SMMEs to get valuable information they must implement or use an appropriate tool to visualise data better and to analyse it quicker than ever before (Moon et al., 2017; Solé-Beteta et al., 2021). There are different BDA tools available from numerous vendors and these tools are promising the same thing to save time from data retrieval to data visualisation (Huang et al., 2018; Walker & Brown, 2019).

SMMEs must require new techniques and technologies ranging from engineering to business, including computer science, networking or analytics if they want to manage large quantities of data effectively (Storey & Song, 2017; Solé-Beteta et al., 2021). Additionally SMMEs also require skilled professionals with the correct IT analytics skills and design skills in order to manage BD effectively (Zhou et al., 2016; Walker & Brown, 2019). According Bhadani and Jothimani (2016) SMMEs can use value chain to produce value from their available data. A value chain is a concept presented by Porter (1980), discussing all set of activities as well as their interactions an organisation needs to perform in order to add value on delivering product(s) or service(s) for their customers (Bhadani & Jothimani, 2016; Faroukhi et al., 2020). According to Soroka et al. (2017) it is a framework that deals with a set of events to create value from available data and has been divided into six phases as illustrated in Figure 2.3.


Figure 0.4: BD Value Chain (Soroka et al., 2017:694)

- Data generation: Generation of data is the first step of the BD value chain for capturing and recording data (Faroukhi et al., 2020). Data is generated from different sources which includes Call Detail Records (CDRs), social media and finance transactions (Bhadani & Jothimani, 2016).
- ii) Data collection: collect, validate and store data (Faroukhi et al., 2020). Data is acquired from all possible data source including Log files, sensors, web crawlers and network monitoring systems. For example SMMEs can use CDRs and also obtain customer's opinions and complaints from social network in order to predict churns from their existing customers (Bhadani & Jothimani, 2016).
- iii) Data transmission: After data being collected, it gets transferred to the storage and processing infrastructure in order to be processed and analysed. It gets passed into two phases namely: Inter-Dynamic Circuit Network transmission and Intra-Dynamic Circuit Network transmission. Inter-Dynamic Circuit Network transmission helps with the removal of data from one data source to the data centres, while latter also helps with the removal of data within data centres, helps in organising data and helps in managing data (Soroka et al., 2017; Faroukhi et al., 2020).
- iv) Data pre-processing: It implies a validation processing, cleaning, reduction and data integration to set up storage (Faroukhi et al., 2020). Data collected from various sources may cause redundancy, noisy and inconsistency hence pre-processing is vital in order to improve data quality for analysis, improve accuracy on analysis and reduce storage expenses. According Bhadani and Jothimani (2016) data can be preprocessed using the following steps:
 - Integration: The data coming from various sources are shared to deliver a unified and uniform view of available data. The two commonly used traditional methods are data federation and data warehouses. Data warehouses fulfils the extraction process, transformation process and loading process. When extract process executes, data must be selected, collected and analysed. The

transformation process is when converting the extracted data into an ordinary format. Loading process is when importing the extracted data and transformed data into a storage infrastructure.

- Cleaning: Data must be tested for correctness, comprehensiveness and consistency. In order to increase data quality, data may be removed or modified. The following are five step processes for cleaning mentioned by Soroka et al. (2017) namely: error types are distinct and determined, errors are recognised from the data, errors are modified, error types and corresponding examples are documented and data entry procedure may be modified to avoid future errors.
- Elimination of redundant data: Data redundancy is when datasets have additional data or data duplications. This has an impact on storage cost, have an increase to data inconsistency and disturbs data quality. To reduce this various data reduction methods such as data filtering and compression must be used.
- v) Data storage: process and analyse data to generate new potential insights (Faroukhi et al., 2020). BD storage applications must offer consistent storage space and powerful access to the data. Factors such as consistency, availability and partition tolerance must be considered by storage systems.
- vi) **Data analysis:** Data analysis utilise analytical methods or tools to model, inspect and mine data to extract value (Faroukhi et al., 2020). When all data have been collected, transformed and stored then data needs to be analysed. Once the data is analysed and the results has been visualised then the decision makers for SMMEs must whether to reward the positive behaviour or to change any negative behaviour. If there is any particular problem, it can be analysed to comprehend the root cause of it and allow SMMEs to take informed decisions and also planned for any necessary actions.

2.8 Characteristics of BD solution for SMMEs

SMMEs must have capabilities to deep understand data characteristics in order for them to be able to collect, process, analyse and transform data coming from different formats into information, insights and meaning for business purposes (El Aboudi & Benhlima, 2018; Ghavami, 2020). Ogbuokiri et al. (2015:340) state that for SMMEs to gain competitive advantage within their markets, they should start looking for the below characteristics when acquiring BD solutions:

2.8.1 Flexibility and choice

Large organisations usually purchase an all or nothing solution that requires their customers to automatically stop using their existing system which contributes to a high cost and time consuming process for their IT departments (Ogbuokiri et al., 2015). With regards to SMMEs there is a different approach because when a specific department wants to automated its current system it works independently with no co-relationship or a team effort with its IT department as a results that specific department proposes their own requirements, justify all the costs for the new system requirement and also choose a solution which only solve their problem (Ogbuokiri et al., 2015). For SMMEs to achieve their objectives with BD they must choose a BD solution that will conform to their existing system and also a solution which has all the capabilities to resolve all their needs (Ogwang, 2016). The solution must be complete with all the features SMMEs might need to resolve their needs and it must not force the SMMEs to replace their existing system (Ogbuokiri et al., 2015; Ahmadi et al., 2016; Ogwang, 2016).

2.8.2 Simplicity

The BD solution must be easy deployed with less than a week for SMMEs to start using it, the BD solution must work seamlessly without the need of a special resource to train the employees and the BD solution must also provide SMMEs with a self-service capability for business users to use (Ogbuokiri et al., 2015).

2.8.3 Cost

The BD solution must be priced to meet SMMEs budget, SMMEs must only pay for features they needed to resolve their needs and licensing must start at a lower cost and grow together when SMMEs increase their scope of work (Ogbuokiri et al., 2015).

2.9 SWOT analysis for BD

The BD SWOT analysis is discussed from a perceptive of SMMEs but there are no limits to discuss or applied SWOT analysis in large organisations because they have more resources and more capacity than SMMEs to implement BD in their organisations (Marja, 2016). According to Ahmadi et al. (2016) BD holds many promises for SMMEs. SMMEs can use BD to benefit new business opportunities, or SMMEs can use BD to understand their strengths and limitations. In order for these SMMEs to have a better understanding, the below SWOT analysis approach will be discussed in more detail:

2.9.1 Strength

The Volume, Velocity and Variety (3Vs) as mentioned by Ahmadi et al. (2016) and Schosser (2020) are the most fundamental strengths of BD for SMMEs. SMMEs can use the 3Vs to get new business opportunities, gain competitive advantage and also making better decisions across all different functional areas either it's from finance, marketing, human resource or operations (Collins, 2016; Schosser, 2020). Some of the other strengths for BD which has been identified by Li (2022:599) are as follows:

- i) BD is generated in volumes meaning the data is available for everyone to use.
- ii) BD can be used SMMEs to limit fraudulent transactions within their organisations.

iii) BD also provides real time tracking for SMMEs which will allow them to respond quickly to any problem.

2.9.2 Weaknesses

The cost to use BD has been mentioned by Collins (2016) and Li (2022) as one of the weaknesses as it is costly for SMMEs to store, collect and analyse large volumes of data. Some of the weaknesses of BD for SMMEs which have been identified by Kaisler et al. (2020:35) are:

- i) Skilled resource SMMEs are facing a challenge of not having a skilled professional to utilise BD to its high potential.
- ii) Infrastructure Some of the SMMEs doesn't have enough infrastructure to store all the data needed to resolve their problems and also are facing a challenge of getting new technologies to utilise BD.
- iii) Data qualify the current data stored is becoming old fast since SMMEs doesn't have a skilled person to utilise it.

2.9.3 Opportunities

As mentioned before Schosser (2020) states BD holds many promises for SMMEs. Some of these promises are also opportunities for SMMEs for using BD. Some of the opportunities for BD to assist SMMEs are to i) gain new business opportunities, and ii) develop quality products and service that will help SMMEs to satisfy their customers' needs (Marja, 2016; Steinhoff et al., 2019). According to Ahmadi et al. (2016) and Lies (2019) using BD by SMMEs will allow them to be able to customise their products and service, improve SMME's reliability, prevent fraudulent activities with their organisations, and also assist them with effective marketing of their products. SMMEs can also use BD to enhance their working environments by improving flows of information and responses, and also increase their business value (Marja, 2016; Li, 2022).

2.9.4 Threats

As much as Volume has been mentioned as one of the most fundament strength for BD, collecting of more and more data has its own risks because some of the collected data will not be used appropriately by SMMEs (Ahmadi et al., 2016; Li, 2022). Some of the other Weakness for BD which has been identified by (Marja, 2016:09) are as follows:

- i) Analysing wrong data by SMMEs will cause a waste of resources.
- ii) Privacy and cyber-security issues has also been identified a major threat for SMMEs.
- iii) SMMEs might suffer a backlash from the public if the community feel that their data are misused.
- iv) Also SMMEs might need to hire a new skilled resource for working with BD applications.

2.10 Innovation strategy to guide successful implementation of BD by SMMEs

IT innovation strategy has been described by La-Olivo et al. (2016) and Bagherzadeh et al. (2020) as an organisation's strategy which uses IT as a main principle to improve innovation performance and facilitate economic activities performed by each organisation. The main drive of an IT strategy is to make sure that development of a project portfolio is implemented effective and efficient, and to increase competitive business environment (La-Olivo et al., 2016; Bagherzadeh et al., 2020). However Rogers (2003:12) has defined innovation as an "idea, practice or object that is perceived as new by an individual or other unit of adoption". SMMEs cannot be competitive if their IT strategies are not implemented carefully.

It is important for SMMEs to response in time and with accuracy to the customer's needs in order to achieve a competitive edge, and provide efficiency and effective solutions to clients (Alwan & Ku-Mahamud, 2020). However, it has been recognised by researchers that the adoption of BD by SMMEs will not only bring opportunities and optimisation, but also challenges to them (Zhao & Yang, 2017). Table 2.2 illustrates opportunities and challenges faced by SMMEs when implementing innovation strategy for BD (Zhao & Yang, 2017:04).

Table 0.2: Some opportunities	and challenges	faced by	SMMEs when	implementing	innovation	strategies
for BD (Zhao & Yang, 2017:04)						

	Domains	Dimensions	Explanation		
Opportunities	Environment	Technical Environment	SMMEs must emphasis on data management tools in order to use more and more data collecting from different sources.		
	Task	Operations Management	SMMEs will have an opportunity to identify its data source, and also have an ability to collect, store and analyse such BD in order to achieve the organisation's strategic and operational goals.		
			Improve their services and customer segmentation.		
			Enhance product design by reducing defects cost for all products.		
			Obtain new business opportunities.		
		Strategic Management	Ability to implement customer segmentation strategies in order to create competitive advantage.		
			Ability to determine strategies to meet customer's needs and make better decisions.		
			According Intezari and Gressel (2019) the analysis of BD will help SMMEs to take immediate actions, adapt business processes and also improve their customer experiences.		
			SMMEs can use BD to improve the usage of existing assets and also use BD to drive an open innovation strategy to increase new business opportunities (Trabucchi & Buganza, 2018).		
		Marketing Management	Improve after-sales services.		

	Domains	Dimensions	Explanation
		Human resource Management	Majority of SMMEs will have a functionality to provide accuracy online recruitment.
Challenges	Technology	Data Management	SMMEs are facing challenges of shortage skilled personnel for analysing and managing data effectively.
		Data Technology	SMMEs are not following to their data opportunities because of either they do not know about BD or are not yet matured enough to handle BD (Revathy & Mukesh, 2017:98).
			BD processing is also a critical challenge faced by SMMEs which needs efficient strategies on how to collect, analyse, store and aggregate huge quantities of data (Boubiche et al., 2018).
			Lack of Business cases – Lack of case studies or success stories is a major concern for SMMEs to successful spread innovation within the organisation (Coleman et al., 2016).
	Task	Human Resource Management	Shortage of in-house data analytics expertise SMMEs are faced with either less or no in-house data analytics to analyse BD (Coleman et al. 2019).
			Lack of executive sponsorship.

2.11 Big data strategic capabilities

BD can transform SMMEs IT functions into a dynamic capabilities offering SMMEs opportunities for performance advantages and assist them in examining customer's behaviours, evaluations and their needs (Green et al., 2018; Ciampi et al., 2021). SMME's market competition is increasing, and as a result decision makers are more dependent on BD to classify performance trends, improve product performance and enhance inventory management (Lamba & Singh, 2018; Ciampi et al., 2021). Exploiting the capabilities and unknown potential of BD can lead SMMEs to make better decisions in order to improve profits, make better customer satisfaction and quick resolution of customer issues (Ciampi et al., 2021). SMMEs can utilise BD to detect if any of their products usage are slowing in order avoid possible churns from the consumers and also assist them to implement value co-creation initiatives (Hofacker et al., 2016; Ciampi et al., 2021).

According to Trabucchi and Buganza (2018) and Olszak and Zurada (2020) BD can help SMMEs to improve on the current usage of the existing assets, increase the level of personalising their offering services, competitiveness and survival. BD can enhance SMMEs to drive open innovation strategy as a mechanism to gain new business opportunities and develop an innovative value proposition where data is either used a central or supportive role (Rajapakse & Kiran, 2019; Ciampi et al., 2021). SMMEs can utilise BD to make decisions on whether to continue or discontinue with a product based on the real movements in the stock

price over the past six month or a year period (Intezari & Gressel, 2019). SMME's management can use BD to understand their business at a more granular level, invent new products or services, respond to business changes at a fast rate and help SMMEs to take complex decisions that are more competitive in the market place (Santos et al., 2017; Dubey et al., 2019).

BD will have a substantial influence on value creation and competitive advantage for SMMEs by improving or creating new ways on interacting with customers or developing new products and services strategies to raise profits (Santos et al., 2017; Line et al., 2020). Furthermore SMMEs can use BD to become smarter and innovative on doing things that were considered not possible before and assist them to implement user centred innovation and user driven innovation (Alharthi et al., 2017; Ciampi et al., 2021). SMMEs operating in a manufacturing industry can use BD to assist on the usage of smart grids in order to monitor the energy supply, usage within the production process and improve supply chain and operational performance (Shah et al., 2017; Dubey et al., 2019). SMMEs can utilise BD to change the creation of new products by limiting the time to the market, reduce cost and optimise product adoption by the customers (Zhan et al., 2019; Line et al., 2020).

2.12 The role of BD in decision-making

It is with no surprise that decision making is the most imperative duty of any manager, and sometimes it has a direct and huge effect on SMMEs (Thore-Olsson et al., 2019). According to Thore-Olsson et al. (2019:1068) the definition of decision making is either a "rational process consisting of several sequential steps or a more chaotic process consisting of dynamic linkages viewed as interwoven network of issues". According Côrte-Real et al. (2016) and Li et al. (2022) using BD for decision making will make an SMME to become an agility organisation when using its current knowledge, results in becoming a more competitive organisation and lead SMMEs to make better decisions to improve profits. SMMEs can utilise BD to predict hardware failures, ensure data integrity to improve risks versus rewards and also improve their current positions on determining potential risks and incidents (Rajapakse & Kiran, 2019).

SMMEs in the financial institutions can use BD to support their core businesses by analysing all the transactions to detect if there was any fraudulent transactions or money laundering (Frisk & Bannister, 2017; Shamim et al., 2019). SMMEs operating in the sector for example financial institutes (FSPs) can also share their client's claims and use BD to detect patterns which can point to the same fraudulent activities among each other (Trabucchi & Buganza, 2018). According to Frisk and Bannister (2017) decision making is made up of three phases namely intelligence, design and choice. Furthermore Frisk and Bannister (2017:2077) described these phases as follows:

i) Intelligence as a logic of gathering decision relevant data and information.

- ii) Design as a scrutinising the options to determine the expected results and also analyse how these results will meet the actual goals.
- iii) Choice is about making a selection between the expected results.

SMMEs can use BD to understand which organisational policy or policies would fail when a certain decision has been taken by its decision maker(s) and also identify policy or polices that will suffer under a range of scenarios (Foley, 2018; Shamim et al., 2019). According to Lamba and Singh (2018) SMMEs can use BD to enhance inventory and productivity, shorter cycle times, increase customer satisfaction, and use it to make real decision to faster resolution for customer issues. Intezari and Gressel (2019:75) categorise organisational decisions into two different angles namely i) structured and ii) unstructured decision making.

2.12.1 Structured decision making

Structured decision making is a defined as a common mathematical models (statistical methods and linear programming), with no standards and overall method to gain the best solution to address the unstructured decision problems and can help in assisting decisions that lead to action (Krell et al., 2017; Robinson et al., 2019). Deciding on whether to carry on or discontinue on a particular product based on its stock price movement over the period of its previous six months is considered by Intezari and Gressel (2019) as a structured decision making. Windheim (2020:99) claims that any effective decision made by SMME's executives has been considered through a systematic process with a strong defined elements in the following diverse order of steps:

- i) Problem classification
- ii) Problem definition
- iii) Specifications satisfying the answer of the problem
- iv) Decision considered to be right, rather than being acceptable to meet all conditions
- v) All the action plans for the decisions
- vi) Results supporting the validity and effectiveness of the decision against the actual problems

BD knowledge management systems are considered by He et al., (2017) and Angelis and Phillips (2021) as the embracing tools for better decision making and quantitative decision analysis. Therefore SMMEs can start investing on BD knowledge management systems initiatives in order to improve their business performance, grow their innovation capabilities, and attain better competitive advantage (Pauleen & Wang, 2017; Angelis & Phillips, 2021).

According to Norman et al. (2018:3590) there are four key functional areas for KM, namely:

- i) Collecting data into KM systems.
- ii) Storing the data into the warehouse for a certain period of time while furthermore allowing users to use it when its required.
- iii) Offering capabilities for analysing and analysis tools such as BDA.

iv) Provide visualisation and reporting.

SMMEs must make sure their KM system are compatible to meet system's wide requirements such as security, reliability, manageability, scalability, redundancy and usability before implementing them (Norman et al., 2018). Furthermore KM system must provide easy to setup and easy to use configurations and must be easily managed without excessive any cost (Shinde et al., 2017; Angelis & Phillips, 2021).

2.12.2 Unstructured decision making

Unstructured decision making do not follow a set of order task, therefore they heavily depended on human judgment, experience, previous knowledge and analysis of the decision context and alternatives (Intezari & Gressel, 2019). Business Intelligence (BI) tools are considered important and very useful in making unstructured decision making because they have a reliable and fast reporting mechanisms (Lugmayr et al., 2017; Lucena & Popadiuk, 2020). According to La Torre et al. (2018) Mark Zuckerberg's decision on purchasing WhatsApp for \$19 billion is an example of an unstructured decision making.

The decision was based from the available data, during that time there were about 450 million active users in December 2013 and with an increase of 1 million users joining WhatsApp every day (La Torre et al., 2018). When making unstructured decisions, business reporting tool such as the balance scorecard can be used by SMMEs to visualise their business performance metrics (Lucena & Popadiuk, 2020).

2.13 The role of BD in customer segmentation

Segmentation has been defined by Thoeni et al. (2016:2195) as the "process of dividing the total market into a number of smaller, more homogeneous submarkets and termed market segments". Customer segmentation is defined by Munusamy and Murugesan (2020:1922) as a "process of dividing a broad customer base into manageable sub-groups of customers". Furthermore Thoeni et al. (2016) and Munusamy and Murugesan (2020) considered segmentation as the most important and broadly practised marketing approach SMMEs can apply to understand their customer's needs. Segmentation can help SMMEs to target customer based on which customer attraction and retention strategies are framed.

Munusamy and Murugesan (2020) argue that segmentation is a process of grouping together similar types of customers with a common characteristics such as buying same iterms, same values and same demography. SMMEs can use BD to customise their products to fit a distinct demographic, psychographic, geographic group or use BD to understand customer's behaviour towards that specific product (Nguyen, 2021). SMMEs can utilise BD to boost their profits, nurture and strengthens customer relationships and also provide a seamless experience across all channels (Pozza et al., 2018; Nguyen, 2021). Customers have different preferences, some prefer different attributes from the same product or service, therefore

SMMEs can utilise BD to create business experience which fulfils and exceeds all the customer expectations (Guillet & Kucukusta, 2016; Nguyen, 2021).

2.14 The role of BD in innovation

Innovation has been indicated Sun et al. (2018) and Ghasemaghaei and Calic (2020) as the most vital driving forces to support economic growth and organisations are investing on BD to discover innovative ways to differentiate themselves from competitors. According to Esterhuizen et al. (2017) and Ghasemaghaei and Calic (2020), innovation is a fundamental requirement SMMEs can utilise to achieve competitiveness. BD is the next stronghold for innovation, competition and productivity. BD can help SMMEs to reduce inefficiencies in product delivery, improve marketing and access on products and services, increase qualities on products and services, and make more personalised products and services for new and existing customers (Hao et al., 2019). Calder et al., (2016) and Hao et al. (2019) emphasised that if SMMEs and their customers are involved in product innovation, both parties will gain value from pricing structures and distribution channels.

Utilising BD during innovation life cycles will help SMMEs to gain better lessons throughout all the phases and at the end of each phase they are opportunities SMMEs can learn from the successes and failures of it (Fortunato et al., 2017; Ghasemaghaei & Calic, 2020). Using BD for digital marketing will allow SMMEs to receive a real time feedback on their marketing difficulties and help SMMEs to manage all the risks efficiently (Kunz et al., 2017; Arthur & Owen, 2019). This approach (digital marketing) will help SMMEs to capture results from customer behaviours, advance SMME's competences and help SMMEs to sustain their establishment over long time (Kunz et al., 2017; Hao et al., 2019). BD will help SMMEs to increase the usage or improve the usage of the existing assets (Trabucchi et al., 2018; Arthur & Owen, 2019).

During product innovation SMMEs can utilise BD in order to get quick answers to achieve the following: reduce costs, reduce time spent for product innovation, helps them to develop products faster and improve their offerings, and help SMMEs to make smarter decisions (Androniceanu, 2019). Furthermore, SMMEs can use and re-use data at any given time to enlarge company's operating domain, increase their level of personalizing services and leverage data as a paradigm (Nunes, 2017; Androniceanu, 2019). Researchers have recently started to point out key success factors (Table 2.3) of BD in innovation (Zhan et al., 2019).

Table 0.3: Key success factors (Zhan et al., 2019:522)

Key success factors	Essential elements
Pre development research	Idea generation Initial screening Preliminary market assessment Detailed market study Financial analysis Well defined product
Accelerated innovation process	Systematic innovation process Autonomy management Cross functions teams Simultaneous development
Customer connection	Market orientation Customer communication Understanding of customers Good relationship with customers
Ecosystem of innovation	Connection with customers and partners Proficiency of marketing test Fast development and launch Quick response to market Market and partner tests

Zhan et al. (2019) emphasised if SMMEs fail to implement these key success factors, they will have an unsuccessful results for their organisations. According to Del-Vecchio et al. (2018) BD will help SMMEs to create value on effective decision making, improves organisational performance, position them better in their industries, achieve high customer satisfaction, produce more innovative products and create innovate business models.

2.15 Big data and cloud computing for SMMEs

Cloud computing is one of the fundamental components for implementing BD in the industrial process and facilities, which can be effective used by SMMEs to monitor all their applications and are capable of providing SMMEs an ability to analyse and store data economically and efficiently (Terezinho, 2017; Bakhouya et al., 2020). However, SMMEs must first collect their raw data, push the data to the cloud, then store the data and analyse it in order to create functional information (Terezinho, 2017; Bakhouya et al., 2020). According to Stergiou and Psannis (2017) and Aceto et al., (2020) SMMEs can utilise cloud computing to provide accessibility to their data from anywhere at any given time with no need for hardware equipment and allow pay-per use billing on a short term basis without any upfront commitment. Cloud computing has been defined by Aceto et al. (2020:03) as a "paradigm that enables utility computing, for example the leasing of computing resources in real time, with minimal interaction with the provider". According to Sutaj (2017:299) cloud computing is a "model in

which computer processing, storage software, software and other services are supplied as a set of virtualised resources over the internet". Xiao et al. (2019) have defined cloud computing as a public computing which is accessible over the internet and offers users a global and suitable access to a shared pool of computing resources consisting of networked servers, storage and software application. The National Institute of Standards and Technology (NIST) has defined cloud computing as a "model for enabling ubiquitous, convenient, on demand network access to a shared pool of configurable computing resource (e.g. networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction" (Yamin & Makrami, 2015:854). According Al-sharafi et al. (2017) and Aceto et al. (2020) implementing cloud computing services will enable SMMEs to avoid high cost on setting up ICT infrastructure, high fees on servicing and maintaining these infrastructures and enable pay-per use billing strategy.

Cloud computing has been argued by Huang et al. (2018) and Bakhouya et al. (2020) as a cost efficient solution to store data economically and efficiently. It has an ability of providing low cost storage solution to cover massive storage requirement for BD data storage. According to Norman et al. (2018) cloud computing is more reliable, more scalable, and more secured than any other local storage system. Furthermore Andrade and Torres (2018) states that cloud computing has an advantage to provide great user management flexibility, maximise speed processing and provide great visualisation of BD.

SMMEs are faced with lower adoption rate of IT innovation compared to large organisation, cloud services has a potential of reducing some challenges (Salleh et al., 2018). The following are some of the challenges faced by SMMEs that can be reduced by cloud service (Gao et al., 2020:203): i) bridging limited funds to employ a skilled technical personnel for information systems, ii) develop a management tasks from information systems and iii) restrictions to invest the capital on information and communication technology. SMMEs can give their service providers an authority to manage cloud services for them on a pay as you use paying terms (Paramita & Wiradinata, 2016; Aceto et al., 2020). According Paramita and Wiradinata (2016) and Gao et al. (2020), SMMEs which are current using cloud services have an additional benefits on focusing on their businesses, their infrastructure is more cost-effective and they have a shared service for problem solving.

Researchers across different industries have concluded that the cost savings and cost reduction are not the most key factors for SMMEs to adopt cloud computing (Xiong et al., 2016; Aceto et al., 2020). Furthermore Xiong et al. (2016) and Bakhouya et al. (2020) mention that ease to use, reliability, security and privacy are the most top priorities for SMMEs to adopt cloud computing followed by cost savings.



Figure 0.5: Factors that influence SMMEs to adopt cloud computing (Suryaputra & Wiradinata, 2016:152)

Figure 2.5 illustrates some of the factors that influence the adoption of cloud computing by SMMEs (Paramita & Wiradinata, 2016). Cloud computing has the potential to help SMMEs in achieving their organisational goals and objectives, to gain competitive advantage and to provide better services to their customers (Sutaj, 2019). Some of the main benefits of cloud computing for SMMES as indicated by Sutaj (2019:300) are listed in Table 2.4.

Benefit	Detail
Storage	They are no physical limits on data storage from the cloud- based networks. SMMEs can easy increase the storage plan without the need of further investing on the hardware.
Cost	SMMEs can negotiate with cloud providers to deliver an enterprise class technology at a very low cost. This service will enable SMMEs to become more active than their competitors and will also enable SMMEs to minimize their IT and sever licensing fees.
Secured and protected data	The data stored at the cloud is mostly encrypted to avoid online criminals in detecting it. To prevent widespread infecting on the network, infected files are normally not stored and archived.
Disaster proof	Cloud computing will help SMMEs to reduce data loss because their data will be stored and backup everywhere and SMMEs will be able to retrieve their data within a short period of time in the event of disaster.
Reliability and availability	During the event of server outage SMME's IT resources will be reinstated quickly because their data will be stored offsite and backed to other locations.
Focus on core competency	SMMEs will not be required to employ any IT expect to their organisations because every aspects of their data will be taken care by the service providers and that will give SMMEs an opportunity to focus more on their core competency.
Flexibility and Remoteness	Cloud computing services only allow an authorise person to access the resource they need using a secured internet connection and a well-suited device without having to be at the SMME's premises.

Table 0.4: The main benefits of cloud computing for SMMES (Sutaj, 2019:300)

Benefit	Detail
Use of updated software	SMMEs will always have the latest software version because all the software installation, updates and licensing will be managed by the cloud service providers.

According to Al-sharafi et al. (2017) and Gai et al. (2017) in the developed countries businesses and governments are using cloud computing to enhance their service delivery, performance, reduce start-up cost for SMMEs and enhance business competitiveness. Furthermore a study conducted by Srinuan (2018) suggests that the fastest growing small business from five European countries are strongly using cloud technology to deliver their products and services globally. The use of cloud computing by SMMEs will strength SMME's capabilities to deliver products and service that are only delivered by large enterprise in the past, and will flatten the competitive arena (Alshamaila et al. 2017; Gai et al., 2017). According Kumar et al. (2017) and Das et al. (2018) various software applications like Enterprise Resource Planning (ERP), Customer Relationship Management (CRM) and Supply Chain Management (SCM) can be utilised by SMMEs at a low cost using cloud computing. Furthermore SMMEs can use cloud computing without worrying about the challenges of installation and maintenance (Kumar et al., 2017; Das et al., 2018). In summary cloud computing for SMMEs is now available, matured, can be used comprehensively and does not only offers storage at a good rate, it also offers infrastructure and services (Basahel et al., 2016; Bakhouya et al., 2020).

2.16 Big data analytics

SMMEs have identify the importance of BD analytics to change the competitive landscape and extract value insights from BD (Sharma et al., 2016; Datta et al., 2021). Many SMMEs do not want to just collect data, they want to understand its meaning and the importance of it so that they can make better decisions (Elgendy & Elragal, 2016; Liu et al., 2021). Big Data Analytics (BDA) provides SMMEs with techniques of getting hidden information within their organisations and provide it to their analysts and program managers (Norman et al., 2018). According to Durhin (2019) more than 50 percent of SMMEs are currently using some form of BD analytics to prevent fraud, forensics and analyse their network traffic. Furthermore, less than 20 percent of SMMEs are currently using BD analytics to detect their information, predict possible hardware failures, confirm data integrity and verify their data classification. However, SMMEs can enhance their balance of risk compared to reward and can be in a better position to prevent possible risks and incidents (Durhin, 2019).

Research conducted by Wamba et al. (2017) and Liu et al. (2021) suggest that 87 percent of the organisations consider BD analytics to change their competitive landscape. However, 89 percent of the enterprises believe that if SMMEs do not include BD analytics in their strategies. As a result they are at high risk of losing their current market share and momentum (Buhalis

& Volchek, 2021). But investing on BD is still a challenge for SMMEs because SMMEs are struggling to understand the link between BD analytics competencies and their organisation's performance (Dubey et al., 2019).

SMMEs must consider on selecting appropriate BDA technologies if they want to benefit from the scalability of increase volume, velocity and variety of their collected BD (Buhalis & Volchek, 2021). BDA technology has a potential of helping SMMEs to manage their entire data management cycle technically and economically from collection prospective and storing of their entire datasets to analysing data so as to provide new and valuable information for decision making (Sekhar & Sekhar 2017; Dubey et al., 2019).

2.17 The competitive advantage of using BD analytics

A competitive firm is a firm or organisation having a greater success compared to its current or potential competitors (Peteraf & Barney, 2003; Côrte-Real et al., 2016; Akter et al., 2020). According to Akter et al. (2020) BD analytics has been seen a game changer to improve efficiency and effectiveness of SMMEs because it has high strategic and operational potential. Small enterprises can create a competitive advantage through the understanding of information and the prediction of the evolution of facts based on the data they collect.

Organisations such as Amazon are currently using BD to monitor, track or secure more than one billion items from their inventory located in more than two hundred locations globally (Clark, 2016; Aceto et al., 2020). They rely on projecting analytics to anticipate shipment by forecasting customer's next purchase so as to pre-ship the item on the nearest depot closer to the customer (Jha et al., 2020). Amazon has been seen as BD's biggest player because they have engineered their e-commerce channel with a personalised recommendation system that has been built from customer's transactional data when they purchase (AI-Emadi et al., 2015; Jha et al., 2020).

Various industries are expecting BD to have a huge impact on their businesses. For example, the retail industry is expecting BD to reduce their fraudulent activities, to provide them with the better customer experience and also assist them in making real-time decisions (Wamba et al., 2017; Line et al., 2020). In the health sector, BD analytics is expected in provide better quality in life, in reducing all the operation costs (Fan et al., 2014; Akter et al., 2016; Faroukhi et al., 2020). According to Fox (2015:89), "the vast quantities of data generated every year provide enormous opportunities for retail businesses to gain from BDA". BDA helps SMMEs in making intelligence business decision which results in identifying new business opportunities to gain a revenue, provide better quality service and also in job creation (Allen, 2016; Faroukhi et al., 2020). SMMEs can use BD to position themselves in their markets (Alsooj, 2015; Line et al., 2020).

2.18 Big data management

Big Data Management (BDM) has been defined by Alhajj et al. (2020:05) as "a vital requirement to secure data while also allowing users to query and retrieve the data". BDM can be defined as "the organisation, administration, storage and management of large volumes of structured and unstructured data sets" (Ghavami, 2020:04). Furthermore BDM refers to data collection which helps to ensure best quality of data is collected and accessible for business intelligence and data analytics (Beri & Ojha, 2016; Ghavami, 2020). According to Nda and Tasmin (2019), for any SMME adopting BD, BDM is a real challenge because the majority of SMME's are facing challenges of performing BD modelling, BD storing, BD retrieving, BD analysing and BD visualisation.

As indicated by Nda and Tasmin (2019), SMMEs can benefit from BDM with proficiently managing BD's characteristics, keep their data so as to improve their customer segmentation, develop better retentions strategies. SMMEs can use BDM to clean data coming from different sources (both structured and unstructured) and to encrypt their data for security and privacy drives. Furthermore, BDM has a responsibility in collecting and understanding organisational data, with a goal to confirm data reliability for SMMEs (Zhu et al., 2018; Kaufmann, 2019).

BD's architecture must be aligned with the organisation's support infrastructure and business services model in order to improve performance, facilities innovative products and assist in quality decision making (Khan et al., 2014; Alzahrani & Sadaoui, 2020). According to Gupta and Rohil (2020) the main objective of BD technology is to reduce the cost of hardware and processing and also examine the data value before committing to any company resource. BD that has been properly managed is easily accessible, very reliable and secure, and also easily manageable (Khan et al., 2014; Paré et al., 2020).

Preparing data for analytics takes time and require highly skilled resource when doing an analytics on large data amounts as it requires an ability to store, filter, transform and to access it (Assuncao et al., 2019). Deploying BDM tools on the cloud environment and also the solutions on performing analytics have been identified as a major challenge on the cloud (Assuncao et al., 2019). Assuncao et al. (2019:129) further define various models for cloud deployment when adopting it to the enterprise cloud analytics as:

- i) Private: A private cloud is suitable for businesses that require the highest level of control of security and data privacy and in such conditions this type of cloud infrastructure can be used to share the services and data more efficiency across the different departments of a larger enterprise".
- ii) Public: Public cloud offers high efficiency and shared resources with low cost. The analytics services and data management are handled by the provider and the equality of services is specified in a contract. organisations can leverage these clouds to carry out analytics with a reduced cost or share insights of public analytics results.

iii) Hybrid: Hybrid combines both (private and public). Customers can develop and deploy analytics applications using a private environment, thus reaping benefits from elasticity and a higher degree of security than using only a public cloud".

2.19 Big data management tools

As indicated by Roy et al. (2018) BD optimisation is an essential feature for managing a database and warehouse. BD optimization are used for fetching data from several sources so that it can be used for multiple purpose (Shalaginov et al., 2017; Aswale & Mukul, 2020). The rising of computing technology has made huge volumes of data manageable without requiring supercomputers and high cost (Khan et al., 2014; Yadav, et al., 2020). There are many tools and techniques currently available to manage data. However, each organisation needs to identify its unique tools and techniques for storing, accessing and analysing organisation's data because BD cannot be stored in a single machine as it is different from traditional data (Khan et al., 2014; Ghavami, 2020). The most common tools and techniques for BD are:

- i) Hadoop: Baskaran (2019:15) has defined Hadoop as an "open source framework which permits substantial sets of data to be prepared to utilise hardware system". Sekhar and Sekhar (2017) and Dubey et al. (2019) state that it is a java based free programming system that can used to store wide range of information in a group. The data structure for group is running parallel has an ability to allow users to process information across all hubs.
- ii) **Map Reduce:** According to Katal et al. (2013:404) Map Reduce is a "programming model that is appropriate for parallel handling of enormous data dispersed over an extended number of machines backed Hadoop.
- iii) Microsoft HDInsight: According to Sekhar and Sekhar (2017) and Ghavami (2020) Microsoft HDInsight has been defined as a data management tool controlled by Apache Hadoop and act as an administration in the cloud. Microsoft HDInsight uses windows Azure Blob stockpiling as a default framework for documentation to gain high accessibility with minimal effort (Jahanshahi & Bhattacharjee, 2020).
- iv) NoSQL: Since normal SQL are used to deal with extensive organised information, NoSQL (Not Only SQL) are mandatory to deal with unstructured information (Nagar et al., 2016; Ghavami, 2020). The unstructured data stored in NoSQL databases have no particular mapping, each line item has its own specific arrangement of sector qualities. NoSQL have functionality to provide better execution on removing away huge measure of information (Sekhar & Sekhar, 2017; Ghavami, 2020).
- v) Hive: Hive are defined as a transporter of information for Hadoop. They are used for data mining and are always kept running on top of the Hadoop (Mercier et al., 2017; Ghavami, 2020).

vi) SQOOP: This is a technical equipment used by Hadoop to connect with different social databases in order to exchange information (Mercier et al., 2017; Yadav et al., 2020). According to Yadav et al. (2020) SQOOP are usable to exchange structured information into Hadoop or Hive.

2.20 Big data and SMMEs

Technology and Innovation has been indicated by Sen et al. (2016) and Matt et al. (2020) as a right of way for SMME's growth and BD is one of the fundamental key drivers for it. Producing products faster, cheaper, and with higher quality than the competitors, is no longer enough for SMMEs to achieve competitive advantage. SMMEs can use BD to predict their markets and scrutinize their customer behaviour in order to increase organisation's flexibility, productivity, quick responsiveness to meet customer's needs and sustain long term risk analyses (Soroka et al., 2017; Wang & Wang, 2020). Sen et al. (2016) and Saleem et al. (2020) state that innovation will increase the opportunity of SMME's growth because SMMEs have the potential to create the knowledge from new technology when using BD as it is a contributing factor to have a competitive advantage and can transform their traditional business into a modern innovative business. Therefore, SMMEs must take the business risk of adopting BD without any fear of failure for innovations towards the future.

Senarathna et al. (2018) and Datta et al. (2021) suggest that SMMEs must consider implementing cost effective software and hardware solution such as cloud technology and open source Hadoop systems if they want to process and analyse structured and unstructured data cheaper. The implementation of cloud computing can also reduce their total number of IT employees within their organisation. Collecting more and detailed data will boost SMME's performance and adopting BD will improve SMME's capability to narrow their customer segmentation and will furthermore enhances SMMEs creativity in tailoring products and services to their customers.

Coleman et al. (2019:14) have proposed that 30.6 percent of SMMEs are currently producing innovative new products, 36.19 percent are producing innovation for marketing, 28.68 percent are innovating in-house, 10.32 percent are collaborating their innovative with others, 14.61 percent are selling their products online, and 22.01 percent are purchasing their products online. Although SMMEs have a huge potential using BD, they also have a fear for data security and privacy (Shah et al., 2017; Coleman et al., 2019).

According to Shah et al. (2017) and Wu and Guan (2018) BD have a potential to help SMMEs in making more competent demand forecasting, supply planning, sensor data driven operations analytics and simultaneous engineering. SMMEs can use BD to create new business opportunities, increase competence and innovation (Sen et al., 2016; Datta et al., 2021). BD will help SMMEs to prevent potential fraud activities within their organisations and helps SMMEs to gain forensics from previous occurred fraudulent activities. The development

of BD and BD analytics infrastructure will help SMMEs to enrich their customer's engagements and furthermore helps SMMEs to improve operational efficiency, innovate or improve their business model. BD will enable an SMME to outperform their competitors and also help SMMEs to improve their businesses by saving cost for them (Nda & Tasmin, 2019).

2.21 Adoption of BD by SMMEs

2.21.1 Introduction

A large number of theories of adopting and diffusion of IT-based innovations have been proposed and tested. Adoption of IT innovation has been proven as a complex process which can be better understood by using multiple models (Shrouf & Miragliotta, 2015). According to Verma and Bhattacharyya (2017) there are number of key theories which need to be considered when adopting IT innovations. Some of the theories are, i) theory of planned behaviour (Ajzen, 1991:190), ii) theory of reasoned action (Karahanna et al., 1999:200), iii) Technology Acceptance Model (TAM) (Davis, 1989:330), iv) unified theory of acceptance and use of technology (Venkatesh et al. 2003:468), v) Technology Organisation Environment (TOE) framework (Tornatzky et al. 1990:25) and vi) the diffusion of innovation (DOI) (Rogers, 2003:26). For the purpose of this research the following three adoption frameworks namely TOE, TAM and DOI are discussed in more detail.

2.21.2 Technology Organisation Environment (TOE) Framework

TOE has been proposed by Tornatzky et al. (1990) and has been tested and validated by many researchers to determine factors that have influence in adopting a new technology by enterprises (Awa et al., 2015; Ogwang, 2016; Malik et al., 2021). These researchers have commonly presented that various technologies, organisational and environmental factors contribute or stop the adoption of new technology in enterprises (Verma & Bhattacharyya, 2017; Malik et al., 2021).

Ogwang (2016) and Na et al. (2022) use TOE to understand the factors that influence the adoption of BDA in the telecommunication industry, and report that there are a significant number of factors which influence the adoption of BDA in telecommunication based on the following:

Based on the relative advantage of the technology, readiness of the technology and competence of the technology. Also based on the organisation's awareness of the technology, innovativeness, human resources expertise, support from top management and IT competence. Based on environmental regulatory support, compliance regulatory, privacy protection and security, fair competition, satisfactory from the customers trust from customers, business value and churn from customers.

According to Gani et al. (2015) and Na et al. (2022) TOE is used to identify organisation's features that influence the adoption or the implementation of an innovative technology. The TOE framework was used in this study to investigate all the factors which influence SMMEs

in adopting BD to meet their customer's needs and also become more competitive towards their competitors.

2.21.3 Technology Acceptance Model (TAM)

TAM is a concept used by many researchers in explaining IT adoption in various fields which also explains the acceptance of using a technology by the users (Ochieng, 2015; Simões et al., 2020). TAM is obtained from Theory of Reasoned Action known as TRA, which also suggests numerous factors in influencing decisions made by users in accepting and when and how to use a new technology (Gani et al., 2015; Malik et al., 2021). However, TAM has been improved by incorporating Roger (1995) DOI framework to include other factors which help in explaining and predicting the adoption of new technology (Na et al., 2022). TAM has been used successfully to predict the acceptance of new technologies by users (Yarbrough & Smith, 2007; Na et al., 2022). TAM is suggested to be used by SMMEs to monitor their employee's behaviour towards the acceptance of BD strategy within their organisation (Turan et al. 2015; Malik et al., 2021).

2.21.4 Diffusion of Innovation Theory (DOI)

Diffusion is a process used by members of the organisation to adopt an innovative initiative and explains organisational adoption and diffusion processes (Hashim, 2007; Simões et al., 2020). DOI is used by many researchers to describe all the factors which affect the spread of innovation in an organisation and is primarily based on technology characteristics and users perception on innovation (Ochieng, 2015; Simões et al., 2020). According to Rogers (1995:18) innovation is mostly determined by the following five elements to decide whether to adopt or not to adopt a new innovation within an organisation:

- i) **Relative advantage:** as defined by Sanson-Fisher (2004) and Rogers (1995) is when an idea has been seen as a new innovation to bring better benefits for an organisation.
- ii) **Trialability:** which is a degree whereby an innovation can be used for an experiment (Chen et al., 2012:1170).
- iii) Complexity: known as a degree whereby the results of innovation is very difficult to use (Rogers et al., 2005).
- iv) **Observability:** which is determined to a degree whereby the innovation results are openly accessed by other colleagues or peers in an organisation (Rogers, 2003).
- v) Compatibility: "is a measure of the degree to which an innovation is perceived as being compatible with existing values, past experiences and the needs of potential adopters" (Sanson-Fisher, 2004:555).

2.22 Summary

BD is one of the fastest growing development tools with the capability of changing the way SMMEs can use their customer behaviours to analyse and transform it into valuable insights. The use of BD by SMMEs will help them to enhance their existing operating process by

transmitting all their data generated from different sources into a single device and use that data to make better decision. Many researchers have confirmed that BD is the most powerful tool for SMMEs to create more and new business opportunities if it's being utilised to its potential capability. These SMMEs can utilise BD for making decisive informed decisions, improve our communities by providing better services and creates new business opportunities. This chapter has provided detailed summary on how SMMEs can utilise BD in this current 21st century. SMMEs are essential to our economic growth because they have an ability to create new job opportunities, and can address all the challenges of skills strategies which could assist in Gross Domestic Product (GDP) of the South African economy. Researchers across the globe have indicated BD as a core topic because it changes the way industries function by providing organisations with opportunities to collect, generate, analyse and process vast amount of data.

SMMEs can utilise BD to predict their markets and examine their customer's behaviour in order to increase organisation's flexibility, productivity, quick responsiveness to meet customer's needs. SMMEs can increase their profits, develop and strengthen customer relationship and provide seamless customer experience across all channels implementing BD strategies. Innovation has an ability to increase SMME's growth because SMMEs will have the potential to create the knowledge from new technology when using BD as it is a contributing factor to have a competitive advantage.

In the following chapter three (3) the research methodology used in this research is discussed.

CHAPTER THREE: RESEARCH METHODOLOGY



Figure 0.1: Layout of Chapter 3

3.1 Introduction

The aim of this study, as proposed in Chapter 1, section 1.5, is to explore how SMMEs can use BD to differentiate their customer segments and to create a competitive advantage. The objectives of the study is to i) determine if SMMEs see the value of BD for their businesses, ii) identify any challenges faced by SMMEs to include BD strategy to their businesses, iii) identify SMMEs strategy to meet their customer's needs and iv) examine the way SMMEs are measuring their success.

SMMEs do not use BD to differentiate their customer segments and are losing their competitive advantage resulting in market share as well as potential profit lose. The challenges face by SMMEs for not using BD are explored through the following research questions:

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage? According to Doody and Bailey (2016:106), if the aim of the research is "to understand and explain, then the questions to ask are 'How' and 'Why'. If the aim is to explore, then the questions to ask are 'What' and 'How'". In this study two questions (what and how) are asked. In this chapter the researcher provides an overview of a i) research philosophy, ii) research approach, iii) research strategy, iv) research aim, v) data collection, vi) data analysis techniques and vii) ethical considerations are presented.

3.2 Research philosophy

3.2.1 Introduction

Research methodology is defined by Jamshed (2017:87) as an "approach or architectural design by which a researcher outlines a proposal to the problem finding or problem solving". According to Creswell and Miller (1997:36) research methodology is a "broad strategy that outlines our choice and use of specific methods relating them to the anticipated outcomes, but the choice of research methodology is based upon the type and features of the research problem". Research philosophy is an explanation of frame of reference for knowledge development and theory against the same belief of knowledge (Cazeaux, 2017). Research philosophy commonly originates from theoretical backgrounds. Therefore conducting a research is a process which entails a substantial supervision from one stage to another but also following different procedures within the paradigms (Crotty, 2016:43).

Cazeaux (2017:82) defined philosophy as the "use of abstract ideas and beliefs that inform our research". There are other philosophies such as pragmatism, positivism and realism. As stated by Vogl et al. (2019) phenomenological / interpretivist approach in research attempt to understand people's views or practices using a qualitative methods such as observations, interviews and document analysis.

The philosophical stance of this study is influenced by Järvensivu and Törnroos (2010) who has recommended four paradigms of analysis of social theory as proposed by Burrell and Morgan (1979). Figure 3.2 illustrates the four paradigms of analysis of social theory for this study and also illustrates the research philosophies for this study namely: the ontology and epistemology approach.

Therefore the four paradigms are similar to a map showing one's personal perception and comprehending of their social surroundings. Burrell and Morgan (2017) states that the four paradigms signify the intellectual territory of the theorist and as well as the meta theoretical expectations on how to consider the reality.



Figure 0.2: Four paradigms of analysis of social theory (adapted by Järvensivu & Törnroos, 2010:102)

The two main pillars or research philosophy namely ontology and epistemology are discussed in the following sub-sections.

3.2.2 Ontology

3.2.2.1 Introduction

Ontology is defined by Hathcoat et al. (2019:10) as "the study of existence, which talk about the nature of social objects and understanding of the nature of reality". Ontology is the opinion of reality (Vogl et al., 2019:620). As stated by Zehnder and Riemer (2017:4869), ontology is influenced by two concepts namely: i) objectivism and ii) subjectivism. Objectivism debates that reality occurs without experience of it, while subjectivism debates that reality occurs only when there is experience (Vogl et al., 2019). The ontological paradigm is presented as the objectivist and subjectivist paradigm. These two paradigms are discussed in the following two sections.

3.2.2.2 Objectivism

According to Saunders et al. (2019:135) objectivism provides an "impression that the existence of social entities is independent and external to the social factors which result in such existence". Objectivism declares that social phenomena and their meanings have a situation that is unrestrained of social actors. Objectivism indicates that social reality survives without the influence of individuals.

3.2.2.3 Subjectivism

Subjectivism is about "social phenomena which is created from the perceptions and consequent actions of those social actors concerned with existence" (Saunders et al.,

2019:135). This study adopted subjective ontological stance since the researcher sought to understand how FSPs can use BD to differentiate their customer segments and create a competitive advantage and the feedback was obtained from the participants. In order to fulfil the aim of this study a subjectivist approach was used to effectively develop best answers on participant experiences and interpretations. The participant answers were seen as a socially constructed and comprehended from the participant's truth points of views.

3.2.3 Epistemology

3.2.3.1 Introduction

Epistemology concerns questions such as "what fit as knowledge?", how are knowledge entitlements declared?" and "what is the connection between a researcher and what's being researched" (Saunders et al., 2019). According to Du Plooy-Cilliera et al. (2014:23) epistemology was initiated from Greek words: episteme which means "knowledge", logia or logos which means "the study or theory of". Wahyuni (2012) states that epistemology is the comprehending and information about the world and what is all about, and the reality developed from its essence.

Epistemology is concerned with methods of getting information in the world. According to Mirhosseini (2018:469) there are mainly three different epistemological types applied when conducting a research namely: i) interpretivism, ii) positivism and iii) critical realism. The epistemology of this research is aimed at exploring on how SMMEs can use BD to differentiate their customer segments and to create a competitive advantage. The researcher gathers qualitative information from respondents on the topic and recognises the different views from interviewees using the interpretive philosophy.

This research was based on an interpretivist principle because the research was conducted among people rather than the objects and the researcher acknowledged all the different views from all the interviewees. In the following sub-sections i) interpretivism, ii) positivism and iii) critical realism are discussed.

3.2.3.2 Interpretivism

According to Bernard and Harvey (2018:53) interpretivism is a "systematic analysis of social meaningful action through the direct detailed observation of people in natural settings in order to understand and interpret how people create and maintain their social world". Mangla (2020) argues that interpretivism is a method of knowing social constructs using ways aimed to understand human behaviour. Interpretivism starts by comprehending how respondents realise the world and then comprehends the reasoning after what was seemed as an extraordinary behaviours (Myers, 2013).

In this approach the importance is to place individual's ability to create a meaning (Burrell & Morgan, 2017). The researcher seeks reality from human knowledge by directly interacting with the participants in a research setting, the collaboration helps the researcher to acquire

detail information in a subjective context from the participants and the data collected is interpreted to formulate a theory (Hosseini et al., 2019). Walsham (2018) declares that an interpretive researchers begins with an assumption that formulates to reality occurs on social constructions such as language, consciousness and shared meaning.

According to Myers (2018:13), there are six principles of interpretive research that can be used to guide researchers on their studies. Table 3.1 illustrates the principles of interpretive research in the study.

Principles of interpretive field research	Explanations
The essential principle of the hermeneutics circle	The principles propose that the understanding of human is accomplished by rehearsing between the independent meaning of parts and the whole that they practice.
The principle of contextualisation	Involves serious reflection of the community group and the past background of the research situation so that the future interviewer can see how the existing situation under investigation began.
The principle of interaction between researchers and subjects	Involves serious reflection on how the research resources or information were created as an idea by people through communication between the researchers and participants.
The principle of abstraction and generalisation	Involves connecting symbols to describe out information through the application of principles to the theoretical, general concepts that explains the understanding nature of human and social action.
The principle of dialogical reasoning	Involves sympathy to possible conflicts between the theoretical insights managing the research design and actual findings with subsequent cycles of revision.
The principle of multiple interpretations	Involves sympathy to possible dissimilarities in interpretations against participants as are usually expressed in multiple narratives.

 Table 0.1: The principles of interpretive research in the study (Myers, 2018:13)

Based on these principles as shown in Table 3.1 and from the perspective of the research using FSPs that have a formal structure where there is a potential of using BDA as a strategic tool in making decisions an interpretivist approach was followed.

3.2.3.3 Positivism

The Positivist approach is "when the reality has been established and can be experiential and defined from an unbiased point of view without interference from the occurrence of the study" (Sanfilippo et al., 2019:188). As stated by Abu-Alhaija (2019:126) the "systematisation of the knowledge generation process with the help of quantification and scientific methods are adopted by positivistic thinkers to enhance precision in the description of parameters and the relationship among them". According to Welman et al. (2005) positivists attempt to find

relationships laws, which reflects to all the people during that particular time. The positivist approach significance is to place values of reasons, certainty and strength of measurements based on accurate information collected through direct observation and experience (Saunders et al., 2019). The approach consist of the influence of reality with differences of only a single independent variable so as to classify regularities and form a relationship between some of the components of the world (Klein & Myers, 1999).

As stated by O'Leary (2021), the positivist research stand accepts that reality is autonomous from the researcher and his / her tools used for this study and is commonly connected with quantitative theory testing research. The process of choosing a research design is very exciting due to the number of approaches currently available (O'Leary, 2021). Table 3.2 illustrates the differences between positivism and interpretivism.

Meta-theory assumptions of:	Positivism	Interpretivism	
Ontology	Researcher and reality are different.	Researcher and reality are invisible.	
Epistemology	Unbiased reality happens beyond the human mind.	Information of the world is deliberately created from a person's life experience.	
Research Object	Research object has built in qualities that occurs from the independently of the researcher.	Research object is clarified from the fundamental meaning of the person's life experience.	
Method	Analysis and statistics	Phenomenology, hermeneutics etc.	
Theory of truth	Participants theory of their truth: One to one representing between the statement of the research and reality.	Truth is implemented: Analysis of the research object matches lived experience of the object.	
Validity	Certainty: Reality is truly measured from data.	Defensible information claims.	
Reliability	Replicability: Results for the research can be reproduced.	Interpretative awareness: researchers acknowledge and speak out about the implications of their subjectivity.	

Table 0.2: Illustrates positivist approach versus interpretive approach (Cepeda-Carrion & Ruiz, 2005:856)

This study did not followed a positivist approach.

3.2.3.4 Critical realism

Critical realism is also known as realism paradigm and post positivism. Critical realism was recognised out of frustration with positivism's non principles of humanistic, high focus and its importance on the unplanned nature of worldwide laws (Hoddy, 2019). According to Denzin and Lincoln (2018) critical theory is when a researcher embraces more transactional and subjectivist epistemology, where the researcher and the study of research are recognised to

be linked together with the principles of the researcher. Its main objective is to deliver resources to people that will assistance them to change and comprehend their world (Sanfilippo et al., 2019). According to Saunders et al. (2009) critical research emphases on the disagreements, conflicts and flaws in existing society.

The philosophy of realism as mention by Saunders et al. (2019) is to demonstrate that certainty is fairly independent of the mind. Furthermore, critical realist uphold that researchers are required to change social relations by assessing, showing and changing any unfair practices within the society (Du Plooy-Cilliera et al., 2014). The main aim for critical realism is to describe traditions, change society and free individuals from all forms of harassment and equip them in building a better world for themselves during the process (Kelemen & Rumens, 2008). Realism in its particular position is aware of the values of human systems, as it recognises that opinions have a certain gentleness and there are dissimilarities between a participant's perceptions of reality and the reality itself (Denzin & Lincoln, 2018). According to Saunders et al. (2009:119) "realism philosophy illustrates that reality is fairly independent of the mind". For these reasons critical realism was not followed as an epistemology.

3.3 Research approach

3.3.1 Introduction

The research approach is concerned about certain procedure followed while investigating a specific topic (Seijger et al., 2021). According to Saunders et al. (2019:152) the three types of approaches coordinating the research path to be followed are i) inductive, ii) deductive and ii) abductive approach (Table 3.3). An inductive approach is when the research starts by gathering data to explore an experience and build theory (Saunders et al., 2019). A deductive approach is when the research starts with the theory, often begin from reading academic literature and then design a research strategy to evaluate the theory (Kelemen & Rumens, 2008). An abductive approach is when a research collects data to investigate a phenomenon, discover themes and explain patterns, to create a new or modify an existing theory which is subsequently tested from data collection.

Table 0.3: Deduction, induction and abduction: From reason to research (Saunders et al., 2019:153)

	Deduction	Induction	Abduction
Logic	In a deductive reasoning, when the premises are true, the conclusion must also be true.	In an induction reasoning, studied premises are used to create untested conclusions.	In an abductive reasoning, studied premises are used to create testable conclusions.
Generalisability	Generalising from the general to the specific.	Generalising from the specific to the general.	Generalising from the interactions between the specific and the general
Use of data	Data collection is used to evaluate propositions or hypotheses related to an existing theory	Data collection is used to explore a phenomenon, identify themes and patterns and create a conceptual framework.	Data collection is used to explore phenomenon, identify themes and patterns, locate these in a conceptual framework and test this through subsequent data collection.
Theory	Theory falsification or verification.	Theory generation and building.	Theory generation or modification, incorporating existing theory where appropriate, to build new theory or modify existing theory.

3.3.2 Deductive approach

According to Saunders et al. (2019) deductive approach entails the development of a theory that is then put through to a rigorous test through a series of suggestions. A person can prepare a hypothesis to demonstrate or disprove a theory within the positivist philosophical paradigm (O'Leary, 2021). It is the ruling research approach in the natural sciences where laws present the basis of clarification, permit the anticipation of phenomena, forecast their occurrence and therefore allow them to be controlled.

3.3.3 Abductive approach

Abductive approach starts with the observation of an amazing fact, it then works out of reasonable theory of how it could have occurred. According to Van Maanen et al. (2007) reasonable theories can account for what is discovered better than others and it is these theories that will help uncover more amazing facts.

3.3.4 Inductive approach

To achieve all the objectives of this research, an inductive research approach was chosen. According to Seijger et al. (2021) a research approach give assistance to draw conclusions and to confirm the truth in the matter. Saunders et al. (2009:126) defines inductive approach as an "approach to get a feel of what is going on, so as to understand better the nature of the problem".

Saunders et al. (2012) state that the research approach can be the inductive theory, which involves the development of a theory as a result of the observation of empirical data. The inductive approach's advantage is that it allows implications to merge from the collected data

and authorises patterns to be recognised (Ishtiaq, 2019). Furthermore, inductive approach assists with the use of existing theoretical underpinnings to construct questions and objectives and to point out ideas that the researcher might discover during the research process. The inductive approach's advantage is to allow values to emerge from the collected data, to identify all the patterns and facilitates the existing use of theoretical underpinnings so as to develop research questions and objectives (Ishtiaq, 2019:36).

This research took an inductive approach to theory development.

3.4 Research strategy

3.4.1 Case study

A research strategy has been defined by Saunders et al. (2012:48) as an "objective to accomplish a specific goal". Research strategy helps the researcher to achieve or fulfil their aims and objectives of the study (Marshall & Rossman, 2016). Furthermore Marshall and Rossman (2016) argue that research strategy guides the researcher with directions to follow in answering all the research questions and guide the researcher with directions in gathering data to carry out the research. Research strategy has been stated by Saunders et al. (2012:173) as "scientifically techniques utilised during the research process to help the researcher in getting the answers to the research questions".

As stated by Yin (2003:25), the scope of a case study is an "empirical inquiry that investigates a contemporary phenomenon with its real life context, especially when the boundaries between phenomenon and context are not clearly evident". Walsham (2018:53) posits that a case study research is "most suitable for an information systems (IS) research since its main objective is to study of information system within the organisation rather than focusing on the technical problems within an organisation". According to Hartley (2019) a case study research is essential on assisting the understating of a difficult phenomenon and also contribute on adding value to the previous with the same issue.

A case study (multiple case study) has been adopted for this study because this research was conducted with multiple SMMEs using interviews with semi structured questionnaires. The research was based on interviewing fifteen (15) SMMEs within the Financial Institutions in the Western Cape and Gauteng region (Chapter 4, section 4.6).

3.4.2 Sampling

Ruxwana et al. (2010:17) describe a sample as "a process whereby certain part of the population is selected by the researcher using a specific technic to produce knowledge about the population and also to uncover multiple realities about such population". When targeting a diverse population using a correct sampling strategy is very necessary. Purposive, non-random and convenient sampling were used to sample the FSPs and participants in this study. Babbie (2013:1) defines purposive sampling as "a judgmental sampling which involves selection of objects or individuals with a purpose of a researcher to make his / her own

judgment about which sample will be useful or representative". As stated by Babbie (2015) to streamline a sampling process, the researcher must divide the research population into manageable groups and the study can be piloted on each group. Sampling techniques are divided into two types namely: i) probability sampling and ii) non-probability sampling. Some of the characteristics of a probability method as defined by Ishak and Abu-Bakar (2014:32) are: i) generalisations is allowed on the populations as it has been defined by the sampling frame, ii) parameters are used for estimating the population, iii) probability sampling reduces bias and iv) probability sampling applies randomly selection for units. Probability sampling refers to a method of knowing exactly the number and the location of the components of a research population can all be reached by the researcher, and where all the components of the research population can be selected equally or can be all disqualified from the sample research (Babbie, 2015).

The proposed study purposefully select South African FSPs in the Western Cape and Gauteng region that have a formal structure where there is a potential of using BDA as a strategic tool in making decisions.

3.4.3 Unit of analysis with the aim to explore

Fifteen (15) FSPs were selected as the unit of analysis (Chapter 4, section 4.4, Table 4.2) with the aim to explore how FSPs can use BD to differentiate their customer segments and create a competitive advantage.

3.4.4 Unit of observation

The unit of observation was the FSPs decision makers i.e. (Chief Information Officers, Chief Executive Officer, IT Managers and Program Managers). A snowball sampling technique was also used on this research to engage with all willing decision maker's. All the participants that were interviewed on this research were all issued with consent letters allowing them to agree on participating on this research.

As stated earlier the unit of observations were two (2) Chief Executive Officers (CEO), one (1) Head of Risk Compliance, one (1) Chief Financial Officer (CFO), two (2) Managing Directors, one (1) Fund Manager, one (1) Financial Advisor Category 2, two (2) Chief Operating Officers, one (1) Head of business / Chief Operating Officer, two (2) Owners and key Individuals, one (1) IT Manager and one (1) Principal Owner of the Business.

3.5 Data collection

3.5.1 Introduction

Interviews were selected as a primary data collection method for this research and were conducted using semi-structured interview questionnaire (Appendix A). The initial intention was to do all interviews face-to-face at the officers of the FSPs. On Monday the 23rd March 2020, our South African President Cyril Ramaphosa made an announcement that people

across South Africa had to go on a 21 days lockdown and extended it to nationwide lockdown for two weeks. The researcher mitigated this change of the environment risk by doing the remaining interviews (5) using MS Teams and Google Meet.

3.5.2 Permission

Acquiring permission from the participants to carry out the research is an essential requirement. The researcher used an email to send out invitations to all the selected SMMEs (Chapter 4, section 4.4; Appendix B). Electronic follow-ups using email were done before interviews took place. During the interview process all the participants were asked to freely accept and signed the consent letter to participate of the research. All the participants signed a letter of consent to allow for the data collection.

3.5.3 Interview process

The method of collecting data and analysing it forms part of the scientific research design. In this study, semi-structured questionnaires were used for data collection. All interviews were guided by an interview guide (Appendix A). All interviews were recorded, and participants were asked before the interview started to grant the interviewer the permission to record the interview. The recordings from the interviews were transcribed and text analysed for additional analysis using the inductive approach. The transcriptions were mailed to the participants to verify and validate the correctness and content of the transcription.

3.6 Data analysis

The main objective for data analysis is to discover meanings from the data (Hartley, 2019). Data analysis in any research is largely depended on the i) research approach, ii) research methods, iii) methods used to collect the data, and iv) instruments utilised to collect the data (Edwards & Talbot, 2014:57). Pandey and Pandey (2015:70) define data analysis as "the introduction of information by carry out a study on a structured material, from a different point of views". Furthermore Hartley (2019:62) defined data analysis as "a research method that defines the collected content in a comprehensive and brief manner."

After the validation of the transcripts keywords were identified, summarised and then categorised (Chapter 4, section 4.12). For further classification a thematic analysis was done, and themes develop which were aligned to the research questions. This was done in an Excel spreadsheet (Appendix E).

3.7 Ethical considerations

Cooper and Schindler (2003), as well as Resnik (2015) define ethics as rules or behaviour standards which distinguish and guide moral principles that govern a person's behaviour and the relationships with other people. According to Myers (2013:34) principles of ethics consist of the following elements: i) honesty, ii) plagiarism, iii) informed consent and iv) permission to publish. Resnik (2015:2) further defines the principles of ethics as:

- i) **Honestly:** All the finds must be reported by the researcher honestly without any madeup data. The transcriptions were mailed to the participant to verify and validate the correctness and content of the transcription.
- ii) **Integrity:** All promises and agreement made to the interviewees or participants must be followed.
- iii) **Openness:** A researchers must be open to the criticism or any ideas raised by the participants.
- iv) **Confidentiality:** The data must be protected and treated confidentially at all times by the researcher.

The study considered several ethical issues and the researcher made sure that all questions asked did not offend the participant in any way. All appointments for interview were made with informed consents. As stated by Welman et al. (2005) the researcher paid attention to the following ethical consideration:

- i) Permissions from the participants were obtained by the researcher and the researcher informed the participant in advance about the purpose of the research (Appendix A).
- ii) A researcher assured all the participants about the rights of their privacy.
- iii) Assurance has been given to the participants that there will be no physical or emotional harm.
- iv) Unethical tactics and techniques were not used by the researcher during the interviews. The study and all the collected data stayed ethical, the researcher's methodology was to put appropriate measures in place to alleviate any possible physical, psychological, cultural, social environmental and financial risk that will negatively impact all participants involved in this study. Sources of data collection was to interview business decision makers using semi-structured interviews which requires all the participants to signed the provided consent letter. From all the consent letters, participants were aware upfront about the nature of this research and the type of data that needs to be collected.

During this research study numerous ethical principles were well-thought-out and consent letters were forwarded and obtained from all FSP's representatives (Appendix B) for the purpose of data collection. Even though the researcher gained consent from all the FSPs representatives no data was circulated without any FSP's permission. All collected data during literature review, interviews, and recorded transcripts had not been counterfeit it's a true reflection of every participant that was interviewed.

3.7 Chapter summary

In this chapter the research methodology was explained on how the research process was accomplished, the methods used throughout the research process, and how the knowledge was obtained. The methodology was chosen based on the study objectives and aim. Therefore, because of the study's objectives and aims a qualitative strategy was chosen. In order to support the selection of a qualitative strategy the research philosophy, research strategy, research aim, data collection methods, data analysis and ethical consideration for this research were discussed. Data was collected by means of semi-structured interview questionnaire.

The researcher selected a multiple (15) case strategy. Non-random and relevant samples were utilised, compromising of 15 participant from various FSPs in the Western Cape Province and Gauteng Province. Semi-structured questionnaire was used for data collection, all interviews were guided by an interview guide. In this study qualitative data analysis approach has been used to analyse all the interviews carried out and to further explore the phenomenon projected by the research questions. All recorded interviews were transcribed, this was the crucial step in this study and all keywords were grouped together and classified consequently. For classification thematic analysis was used, to bring together all the themes to the research questions. Therefore content analysis supported by thematic analysis technique were also used because they have helped the researcher to summarised, organised and categorised data into a comprehensible themes.

In this study, data was analysed to address the research questions and it has also helped the researcher to achieve the aim of study for this particular research. All the findings were interpreted to gain meanings from them. All data pattern were explained and contextualised in a meaningful context.

During interviews several ethical issues were considered and the researcher made sure that all questions asked don't offend the participant in any way. All appointments for qualitative interview were made upfront to secure informed consents. The study and all the collected data stayed ethical, the researcher's methodology was to put appropriate measures in place to alleviate any possible physical, psychological, cultural, social environmental and financial risk that will negatively impact all participants involved in this study.

CHAPTER FOUR: ANALYSIS AND FINDINGS



Figure 0.1: Layout of Chapter 4

4.1 Introduction

This chapter introduces the analysis and findings gained in the exploration study on how FSPs can use BD to create a competitive advantage. This chapter gives information on the context and respondents, and introduces an analysis of the data collected utilising the methodology defined in Chapter Three. The findings extracted and the interpretations are made from the analysis of data stipulated. The findings, categories and theme are presented along with the corresponding research sub-questions.

For benefit of reference, the problem statement, research questions, research, aim of the study and research objectives are once again stated (Table 1.1).

4.2 Problem statement

FSPs do not use BD to differentiate their customer segments and are losing their competitive advantage resulting in market share as well as potential profit loses.

4.3 Research questions

RQ1: What are the challenges FSPs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

RSQ 1.1: What are the perceptions of FSPs towards the value of BD for their businesses?

RSQ 1.2: What are the factors affecting FSPs when adopting BD analytics for creating a competitive advantage?

RQ2: How can FSPs utilise BD analytics in order to create a competitive advantage?

RSQ 2.1: How do FSPs determine the needs of their customers?

RSQ 2.2: How can FSPs incorporate BD in their business strategies?

4.4 Aim of the study

The aim of this study is to explore how FSPs can use BD to differentiate their customer segments and to create a competitive advantage.

4.5 Research objectives

- i) Determine if FSPs sees the value of BD for their businesses.
- ii) Identify any challenges faced by FSPs to include BD strategy to their businesses.
- iii) Identify FSPs strategy to meet their customer's needs.
- iv) Examine the way FSPs are measuring their success.

4.6 Cases of the study

The research was conducted with fifteen (15) FSPs (Chapter 1, section 1.2) of Financial Institutions in the Western Cape and Gauteng provinces, which made up the units of analysis of the study. FSPs from other provinces were not considered due to time and financial constraints. Interviews were administered to the FSP's decision makers' i.e. (Chief Information Officers, Chief Executive Officer, IT Managers & Program Managers etc.), who comprised the units of observation. All names of the FSPs are omitted for privacy sake and replaced by letters in alphabetical order. The number of employees in each FSPs or whether its owner managed, services or products that are currently provided by the FSPs and FSP's background are presented in Table 4.1.
Table 0.1: Description of the FSPs who participated in the study

FSP Number	Number of employees in your organisation	Services or products provided by the SMME
1	150 full time employees and 120 contractors' employees (full time employees are compliance officers and contracting employees are invigilators for their exams).	Compliances Training Regulator Exams
2	30 Employees	All-in-one pension service – (From advice to pension administration and asset management).
3	55 Employees	Financial consultation – (Help clients to organise their finances so that they achieve their goals).
4	9 Employees	Financial advisors – (Provide advice and services across a broad range of financial products such as: pension funds, shares, medical aid, provident fund and group health care)
5	8 Employees	Financial advisor – (Provide advice on fund management).
6	3 Employees	Financial service advice – (Provide 13 of the 16 services that can be performed by a category two licence).
7	1 Employee	Manage Fund.
8	7 Employees	Capital Management Investment Management Unit Trust Pension Fund
9	37 Employees	Manage money and Portfolio on behalf of a lot of independent advisors.
10	6 Employees	Financial services Investments Medical Funds Risk and life Insurance Short-term insurance Strategic planning – (working with investment specialist based in the US).
11	9 Employees	Wealth Management Insurance Tax and Accounting solutions
12	105 Employees	Medical aid brokerage
13	6 Employees	Unit trust
14	1 Employee	Professional Accounting Auditing
15	14 Employees	Investment brokerage

*FSP = Financial Service Providers

The number of employees ranges from 1 to 150. These FSPs are providing various financial service ranging from investment to financial advising.

4.7 The participants

Fifteen (15) FSPs with fifteen (15) participants were interviewed in order to answer the research questions. Twelve (12) FSPs are based in the Western Cape Province between Cape Town and Worcester and three (3) are based in Gauteng Province between Johannesburg and Pretoria. The participant's role, number of years' experience and participant's highest level of education are presented below on Table 4.2.

For reasons of confidentiality, the names of the participant are not publicized. Codes have been used to reference the participants.

Code	Position / Role in the FSP	Years of Experience	Highest Level of Education
P1	Chief Executive Officer (CEO)	20 Years – (Previous was with Old Mutual, doing farming and was a Senior Agriculture Researcher)	Master's degree in Agriculture
P2	Head of risk compliance	4 Years – (Previous with Alexander Forbes for 15 years as Investment solutions)	LLB degree and a post diploma in Compliance Management
P3	Chief Financial Officer (CFO)	4 Years – (Previous with Old Mutual as a Quantitative Analyst)	Master's degree in Finance
P4	Managing Director	20 Years – (Previous with Liberty for 12 years)	Certified Financial Planner (CFP)
P5	Managing Director	10 years – (Previous with Old Mutual)	Certified Financial Planner (CFP)
P6	Fund Manager	14 Years	BCOM degree and Certified Financial Planner (CFP)
P7	Financial Advisor Category 2	10 Years	BCOM degree and Certified Financial Planner (CFP)
P8	Chief Operating Officer	8 Years	MSc degree and MBA specialising in Financial Risk Management
P9	Head of Business/ Chief Operating Officer	8 Years – (Previous with Helderburg Fruit Juice for 12 years and with Sanlam).	B.Com Accounting
P10	Owner and Key Individual	20 Years – (Previous with Old Mutual for 15 years).	Certified Financial Planner (CFP)
P11	Chief Operating Officer	11 Years	Chartered Accountant (SA) Post Graduate diploma in Financial Planning Masters in Financial Management Certified Financial Planner (CFP)
P12	IT Manger	8 Years – (Previous with Lomil Polymers for 3 years as an IT Manager)	B.Sc. (Hons) Computer Science and a degree in Business Science

Table 0.2: Participant's role, number of years' experience and highest level of education

Code	Position / Role in the FSP	Years of Experience	Highest Level of Education
P13	Chief Executive Officer (CEO).	8 Years – (Previous with Green Road Bank and as a CEO of Siphumelele Investment)	Charted accountant
P14	Owner and Key Individual	3 Years	Bachelor degree in Internal Auditing and member of the South African Institution of Professional Accountant and South African Institution of Internal Auditors
P15	Principal owner of the business	37 Years	Three year Diploma in Management. Qualified Company Secretary FSP Certificate: Level 1

*FSP = Financial Service Provider, P = participant

4.8 Analysis

All the recorded interviews were transcribed and read several times. Table 4.3 is an example of an extract from the transcribed interviews.

Table 0.3: An example of a transcribed interview

Appendix	Appendix : Interview Transcription 1		
RQ No.	Research Question	P1	
RQ1:	What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?		
RSQ1.1	What are the perceptions of SMMEs towards the value of BD for their businesses?		
IQ1.1.1	To what extent does your organisation have experience with BD analytics?	We sit on a lot of different data sources, we have not really done anything with the data rather than maybe doing a demographic slicing and dicing getting to understand the trends who is moving in to the industry who is leaving the industry because we have access to that kind of information via the exams and registrations but we haven't really done anything in my view valuable with the datasets we have.	
IQ1.1.2	What are the main obstacles that would prevent your organisation in using BD analytics?	There are no obstacles rather than resource and time. It is only now we are working with a team of data scientist where we are basic pulling all the different datasets together creating the links between individuals in different datasets so that we can start seeing a more comprehensive profile of individuals, people that we have been dealing and working with, so that is in progress as we speak. But the really problem over time we get busy and	

Appendix : Interview Transcription 1		
		start doing urgent things and not looking at the important things
IQ1.1.3	How does your organisation categorise your existing customers?	For us as the business our customers has got a different views because we view our customers differently to that of a FSP, at the same time we need the same kind of information because I do believe that a FSP should understand what the customer value is, they should be able to understand enough about their customers to classify them and they should also start understanding enough about their customers to sort of classify them in terms of their personality standards because it's one of the things I do believe people do not focus enough about it, they look at numbers and they don't look at individuals they try to quantify things but they don't quantify and evaluate the fact that we are in a people's business not so much in numbers business. All my clients that I have deal with the compliance officer, so we transfer knowledge and the same thing happens with FSP, FSP value is the ability to transfer knowledge to you so that you give me information that I can use to help you to provide solutions.

*RQ = research question, IQ = Interview question and P = participant

After the interviews were transcribed and proofread, the key words and concepts were identified. Figure 4.1 is an examples of a summary from responses and extracting data from transcriptions based on phrases and keywords.



Figure 0.2: Creating a summary from the responses

4.9 Findings

The findings were analysed and represented for each question asked during the interview. All interview answers were transcribed and coded, and as a result 26 findings were derived from the interviews. The participants expressed their views on how their organisation have experienced BD analytics.

4.9.1 Research question 1

RQ1: What are the challenges FSPs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

4.9.1.1 RSQ 1.1: What are the perceptions of FSPs towards the value of BD for their businesses?

IQ 1.1.1: To what extent does your organisation have experience with BD analytics?

A number (8 out of 15) of the participants (P) (P1, P2, P3, P5, P8, P9, P11 & P15) confirm that their organisations either have huge data sources, sizeable enough to perform operations, or enough data for their organisations. However none of them use or analyse data for creating innovative opportunities, new businesses opportunities and intelligent business decisions. P1 said that,

"We sit on a lot of different data sources, we have not really done anything with the data rather than maybe doing a demographic slicing and dicing getting to understand the trends who is moving in to the industry who is leaving the industry because we have access to that kind of information via the exams and registrations but we haven't really done anything in my view valuable with the datasets we have" (Appendix D1).

This is supported by P11 who stated that,

"We have enough data that we can use to improve our client products and also we can use it to analyse our client needs but we have not done anything with the data. As much we have the data but the skill within the company to use the data is not available and also it hasn't been our focus" (Appendix D11).

P6 and P7 mention that there are organisations who are aware of BD and are making use of their personal contacts or pay a third party companies to analyse data for them. P6 said that "we don't have experienced with BD analytics. However we make use of personal connections and also we have a company that we pay between R10 000 to R20 000 a month to source information for us" (Appendix D6 & D7). Furthermore, P8 mentioned that "they have used BD to analyse their current stock (asset stock) on hand and have used data extensively on investments but haven't used it to analyse their customer's needs, however it is still a challenge for them because their data source is not always available" (Appendix D8).

Contrary to P6, P7 and P8, P10 argued that "there are small companies with no experience on BD analytics. Companies such as Discovery or Old Mutual assist them in any data analysis. Some of their data they get through connections. These contacts are also available for their future data needs. P11 confirmed that they do have enough data that they can use to improve their client's products. They use the data to analyse their client's needs. However, they don't have the skills within their organisation to analyse or use the data and have not prioritise BD as focus point. P12 pointed out that "Their organisation are fully aware of BD but their only challenge is time, they have a list of an entire projects prioritised / scheduled for year 2020

and some of them have carried over from previous year (2019) and because of that BD hasn't been prioritized for this year" (Appendix D12).

In the case of P13, the organisation has started working with an organisation from Stellenbosch helping them to use BD to analyse, extract, and enhance their performance and to reduce the emotive elements on their selection and allocation. According to P14 their organisation have no real experience with BD analytics. Their data is stored electronically as they are a paperless environment. Currently they do not analyse their existing data, and only store it as per directive of the Financial Sector Conduct Authority (FSCA). The Protection of Personal Information Act (no.4 of 2013) (POPIA) in South Africa requires FSPs to safely store all their records (financial, administrative, communications and correspondence) for five years. P15 stated that,

"Their organisation have a sizeable experience on BD. To give an indication on what they do, they are an investment brokerage company and they specialize in investment and do not spend time on short-term insurance, medical aid and others, instead they have a specialised areas of where they do their operations. So what they have done per client: they have a whole variety of investment and obviously that comes in a form of a statistic data and from time to time analyse it and make it their business to ensure that all their clients various investments on their brokerage code. They enjoy analysing the data and segmenting it to various forms like investment, retirement annuity and etc. Yes, they have devoted a lot of time on analysing data and it has been working for them tremendously" (Appendix D15).

Finding 1: Most of the FSPs are aware of BD.

Finding 2: FSPs do not have the knowledge and experience to utilise BD for their organisations.

Finding 3: FSPs do not use BD to analyse client's needs.

Finding 4: FSPs lack time and reliable resources as a major constraints to implement BD.

Finding 5: FSPs have little or no experience with BD.

IQ 1.1.2: What are the main obstacles that would prevent your organisation using big data analytics?

This question was asked to determine the obstacles facing FSPs when considering BD strategies and implementation. Only P1 stated that they don't have any obstacles preventing their organisation using BDA other than time and resource. Furthermore, their organisation has started working with a team of data scientists who help them to pull the different datasets together. The data scientists also create the links between individuals in different datasets so that FSP can see a more comprehensive profile of individuals, as well as people they are dealing and working with.

P2 believed that cost and the skill to handle BD are the main obstacles preventing their organisation in using BD analytics, since their head office is situated outside South Africa.

Providing money to their South African branch has always been a major challenge. They don't have the necessary skills or understanding of BD and there is no specific focus for the skill. As a results they don't have the necessary systems or tools implemented.

P3 was of the opinion that collecting the right data from a reliable source has been identified as their major obstacle preventing their organisation from using BD analytics. Their organisational data is more about people's personal life which gets collected by their consultants and feeding their system. They look at various things about the person lives of their clients. The data is qualitative data in nature, getting it into the right structure to be usable is quite difficult for them. A further challenge is to get consistent data. P3 said that,

"I think here (Lifecheq) the main thing is collecting the right data and collecting it reliable. Our information for example is a lot about someone's personal life so our consultants feed that information in our system for example this is how much I earn, this is how much money I have left, this is how many people I am supporting in my family and we look at various things about the person and a lot of that is qualitative information getting that into the right structure and become usable is quite difficult for us at the moment and also getting that information from people in a consistency way is hard. For me getting the process to get that right is tricky" (Appendix D3).

P4 stated that they don't have any obstacle(s) that prevent their organisation from using BD analytics other than collecting and analysing data.P4 explained that "They don't know how to collect and analyse any data because according to them if they would like to use the BD, they have to source their own BD and they don't know how they would use that data to gain more clients and sell more products" (Appendix D4). P5 believes that they don't have any obstacle(s) that prevent their organisation from using BD analytics however in the same breath said that "They don't have time since their organisation is focusing more on making sales to keep the organisation moving" (Appendix D5).

P6 and P7 stated that their organisations are currently not using BD, but indicated that they recently attended a course on behavioural change in financial management. At the course it was pressed on the delegates that gone are the days where FSPs just sell products to clients. Instead of just selling FSPs need to provide services based on client's needs. P7 indicated that,

"The only obstacle they have is to segment the data to get correct information out of it. Furthermore, one of their organisation is running BD for them. Running the process to extract good data require about 6 hours to overnight to download such data from Bloomberg. Importantly they do see cost as an issue. They are willing to pay for quality data and have it analysed for them" (Appendix D7).

P8 did not mention any obstacle(s) preventing their organisation from using BDA. However, the P8 has indicated that they are in a big drive currently with ESG investment (Environmental, Social and Governance). According to P9 cost is their biggest obstacle. They have between

8 000 and 10 000 individual clients. If there was a cheaper way for them to extract information and better ways to apply BDA they would consider it. They are currently contracted with almost 100 FSPs to outsource their client's investment management for them. P9 stated that,

"Cost is our biggest obstacle. We have relationship of between 8 000 to 10 000 individual clients. However, if there was an easier and cheaper way to extract and utilise client's information then we can consider using BDA. We are currently contracted with almost 100 of FSPs that outsource client's investment management. We have employed a specialist in the financial advisor field to learn about financial investment, funeral cover, gap cover, and medical cover. It becomes more complicated such that the regulator keeps on changing the rules. One needs to have a category 2 which is sub categorised into two: where category 2a requires to run a fund with a minimum amount of 3 000 000 (running from their account not the companies money) or thirteen weeks of working capital in their banking account at any stage of the financial year. This is to avoid any adverse risk such as Ponzi scheme where clients lost lots of money. In a nutshell they manage funds on behalf of the financial advisors" (Appendix D9).

P10 mentioned that they haven't come across any obstacle and it is difficult for them to respond to this question. However, on further probing P10 said that the ease of getting access to BD and reliability of the data is an issue. According to P11 their organisation lacks the relevant skill and do not have the time to implement BD strategies such as collecting and analysing the data.

According to P12:

"Their main obstacles preventing them from using BDA are as follows: time and handling of an unstructured data. Talking about emails, they have thousands of emails that goes out of their server per day, under number of unstructured data sitting there and it is difficult to search. However they have a project setup for next year planning on making their emails far more structured so that they can have them system based, such that they can be able to search through them and indexed, and with the whole point of BD they can be able to get information that can assist their business" (Appendix D12).

P14 did not have any obstacle(s) preventing their organisation from using BDA. P14 said that they don't have time for it and it's not in a line of their profession. They had been approached by other insurers asking if they can provide them advises on the analysis to push or increase their business. The FSP did not do this as yet and doesn't think there is a current need for that beyond what their current business process which is data storage.

P15 indicates that their specific problem is to get information pertaining people which are retiring for an example at the University of Stellenbosch, furthermore that exercise to access that information of people retiring is an obstacle for them.

Finding 6: Time, cost, resources and skills are the main obstacles preventing FSPs from using BDA.

Finding 7: Collecting reliable data and analysing the data are seen as main challenges.

IQ 1.1.3: How does your organisation categorise your existing customers?

This question was asked to examine the challenges faced by FSPs when using BDA in building and differentiation customer segmentation to create a competitive advantage. P1 states that they view their customers differently compared to other FSPs. However, they need the same kind of information because a FSP must understand their customer's value, must understand enough to classify customers with their personality standards. P2 indicated that,

"We use a very simple model. We look at various types of products we sell or provide (A, B and C) and our existing customers. We arrange our products into different portfolios using the following criteria: how big the portfolio is and how long we have stored a customer in that portfolio before it gets matured. Eventually all of these criteria impact our bottom line eventually and our retention of existing customers. However, it is based on how long a customer has been paying premiums in order to retain them" (Appendix D2).

P3 mentioned that their customers are categorised based on location. P3 said that,

"Cape Town and Johannesburg clients are not the same. Johannesburg clients are more keen to try new things. When looking at their finances they often have side hustle, unlike Cape Town clients generally don't have any side hustle, very little on engaging and they don't have lot of questions compare to Johannesburg clients" (Appendix D3).
P4 stated that they categorise their client as good or bad customers. They regard as a good customer if the customer pays their product(s) or account and a bad customer doesn't pay their product(s) or account. P4 followed by saying that,

"We decide whether to sign up with a client based on the ease of doing business with the client. This was substantiated by an example of client A with 10 million rand as a difficult client to work with versus with client B who has 1 million rand as an easy client to do business with. Furthermore customers are also categories based on ease doing business with them" (Appendix D4).

P5 shared that they categorise customers based on the timeframe having business relationship and said that "Customers with less than five years relationship are categories as "Bronze" customers, customers with more than five years relationship but less than 10 years relationship are categories as 'Gold' customers and customers with more than 10 years relationship are all categories as 'Platinum' customers" (Appendix D5). P6 and P7 mention that ,

"Our clients are categorised as: 'A', 'B', and 'C' which is based on the income they generate. An 'A' client generates more than R30 000 a year for the organisation, a 'B' client generates more than R15 000 a year for the organisation and a 'C' client

generates any amount less than R15 000 a year for the organisation" (Appendix D6 & D7).

P9 indicated that they categorise their existing customers based on the timeframe they have a relationship with the customer and stated that,

"Our business started in 2001, we categorise our customers based on the number years with had relationship with them and also on the size of investment. For example, an R100 000 amount is regarded as a small investment and R10 000 000 amount is regarded as a big investment. We also value and manage a relationship of a customer who has an R10 000 000 amount, with a son saving R500 towards his retirement annuity, a wife with R30 000 tax free savings account" (Appendix D9).

P12 stated that they categories their customers based on the services they offer to them. P12 further said,

"They categories their customers based on services they offer versus what they can get in return from their customers. The complementary services are for specific customers but it boils down to the specific needs of the customer. For example, Medshield scheme have a lot of co-payment cost involved and their Gap cover product covers it. On the other hand you will find other medical scheme does not offer copayment as much as Medshield does. Therefore, they are prime to market gap cover product with customers using Medshield scheme" (Appendix D12).

P15 said they sub-divide their clients as "A", "B", "C" and "D" in terms of the amount money that they have investment with them.

Finding 8: FSPs categorise their customers based on service they offer to the customers.

Finding 9: FSPs categorise their customers based on their age, affordability, location, timeframe they had business relationship with them and income they generate from them.

IQ 1.1.4: How do you manage your competitiveness against your competitors?

This question was asked to gain better understanding of any factors affecting SMMEs to adopt BDA for creating competitive advantages and to determine how FSPs can utilise BDA in order to create a competitive advantage. P1 stated that their customers are different from other FSPs. For them the more they understand FSPs the better they service their needs. P1 stated that,

"With all their competitors they have the most data about FSPs. They have the most data about FSP's level of education, and number of years' experience. They know in advance how long it takes for a FSP to be complaint or meet certain criteria for being complaint. They are aware that very few of their competitors will actually have that data, and even if they have, they wouldn't know about it or know how to utilise it. They are busy developing a software application that will calculates all the requirements a FSP needs to be compliant for example, number of experience, any short term training

required, qualifications and continuous professional development. The system will also provide a functionality to indicate what qualification a FSP would need and indicates when a FSP is required to start the qualification" (Appendix D1).

P2 said that managing competitiveness against their competitors is a big issue for them. They don't manage competitiveness against competitors because their company is huge at UAE, UK and US, which makes them unworried about South Africa market. P3 mentioned that they don't think there is another company in South Africa currently doing what they doing. P3 mentioned that they do have competitors because every company does have competitors. P3 preferred not to say much about their business since it's quite a key for them right now.

P5 stated that they use a pricing analyses model to manage competitiveness against their competitors. They strive to provide cheaper advice fees compared to their competitors.

P8 mentioned that on the institutional space they do not focus mainly on the competitiveness against their competitors saying that,

"They consider which asset classes is most effective to their business and focus mainly on the equity investor. They analyse the life style of the investor. In essence they are quality style investors, others focuses on value growth and momentum is more indexing and they compete among index investors. However, there is a limited data on various styles because everybody has their own interpretation and it is largely driven by the data provider body like MSCI who has defined indices based on various equity styles and as an organisation they have their own products, a quality investment fund, some competitors may define it differently but it all depends on what type of data it is in your disposal. For their funds they have done a research and valuation to package it to suite their style. It is difficult to do a comparison with other competitors because everybody wants to be unique in their styles" (Appendix D8).

P9 argued that everything in the world is competitive, but all depends on good services in the most realistic way. P9 stated that,

"Their business runs on 25 percent base the investment company makes annually. For every rand invested they got 0.25 return on investment. It takes time to build this kind of business and relationships as well. Performance is key, if you are sour in performance you run a risk to lose a customer the minute the market don't perform. Their priority is to be the best performing organisation, by taking a volatility risk and making it into a package that they can sell to advisors" (Appendix D9).

P11 argued that the more they know their customers the better they can service all their dissimilar needs. The service is based on the data they have about the customer. So, their best strategy for now is to collaborate with other organisations so that they can better service their customers. P12 stated that they manage their competitiveness against their competitors by comparing their services and products to similar competitive products and services to see

where a gap can be obtained. P12 mentioned that they certainly use their gap product cover and compare it with other gap products out there. P12 supported the argument by saying that,

"They have their own gap cover and therefore they are in charge of its benefit and cost. Fundamentally they compare gap cover benefits yearly, with all types of claims came through, customer's needs and compare it with their competitors gap cover in order to adjust their benefits accordingly. This is done yearly exercise to make their product competitive" (Appendix D12).

P13 mentioned that they outperform the competitors. It depends on how to they manage their performance and cost structure plays a role.

P14 stated that they believe in the way they do their business. The operation of their business is conducted electronically and overall they manage their competitiveness based on the service they provide. Their correspondence are mostly electronically and are on record. Sometimes they have brokers that go visit their customer and all engagements are well documented. P15 stated that this is something they don't really have a problem with. The quality of advice they provide serve as a sort of a divider between them and their competitors. Finding 10: Most of FSPs believe they don't have competitors.

Finding 11: FSPs don't manage competitiveness.

Finding 12: Some of the FSPs manage their competitors by: attending roadshows and presentations, lowering advisory fees and provide good quality services.

IQ 1.1.5: How would big data add value in the organisation?

This question was asked to determine if FSPs sees the value of BD for their businesses. P1 stated that BD would add an immense value to their business. P1 mentioned that,

"They know for instance by using BD they would be able to indicate how many repetitions or datasets they would need to improve their reliability scores in their predictive modelling or information management. They are keen to understand why a particular customer is battling with a qualification. They will be able to help and assist customers who are battling to finish any qualifications. Depending on what angle people are involved in their business, BD will help them to understand that person so well. They will be to manage and coach that person before that person realize his / her needs or problems" (Appendix D1).

BD adds value to both internal and external processes. Internal process means BD will bring more staff satisfaction, staff happiness and help them to retain their staff as P2 said, "BD will add value in their organisation by helping them to determine system and process upgrades, helping their employees to operate better, provide better services to their customers and help their organisation to keep their customers and be better competitive in their market" (Appendix D2).

P3 as well as P4 argued that BD adds value in their organisation because they are dealing with number of aspects on customer's financial life. They have an insurance product which requires lot of customer's personal medical history. P3 mentioned that,

"There are lot of BD applications in the market that contain customer's medical history that they can use upfront to determine customer's monthly premium. New start-up insurance companies will start using the medical aid data history to improve pricing. They know a start-up company who is going to use that data and they know a big insurance company who has been trying to use that data but unfortunately it's been struggling. They understand a good system interface will help them to collect data from their customers" (Appendix D3).

P5 and P13 said that BD adds an immense value in their organisation by identify new clients because according to them getting new clients and creates the potential to earn more revenue by increasing performance and profit margins. P8 mentioned that "The more they have good quality data access the better the analysis outcomes because it will be based on clean data. BD will add value in their organisation to perform risk analysis" (Appendix D8). P9 stated that,

"They do not use BD since their main focus is on investment. They require an upfront fee to their customers but in most cases they do not want it. To operate a business the same way they operate, a FSP will require to have at least 10 years of relationship with its customers before it starts gaining a profit income. Online investment may be an approach but in South Africa is not yet forthcoming. Their approach was to then manage big pocket of fund online. They don't necessary develop a products, and most of the money they manage is discretionary fund, which different clients purchase in a form of unit trust, equity, commercial properties, bonds, money market what they blend into a package for a client. That is not really a product but investment money lying around from different companies e.g. Alan Grey, Investec. The only product that they have is retirement annuity and tax free savings other that they do see indulgent policies or trade indulgent products as products but those as well they do not create them but adopt them from big investment companies. They tailor-made these products according to the customer's need. Some of these products are showing profits, some are showing declines and loses. When they consult with a customer for advisory, they record all their sessions. After the session they conduct an objective analysis that determines the customer's appetite and benchmark for the portfolio investment. They measure their performance using rolling terms which is CPI +3 over a three year period" (Appendix D9).

P10 indicated that the more data they have the more they can do business. P11 further mentioned that BD will help their organisation to better understand their client needs and help them identify their peak periods so that they can sufficiently cater their services.

FSPs need to obtain information which they cannot just get it by looking at raw data. They need to analyse it and get trends from it. It became evident that there is a space for BDA to push/take their business to the next level. P14 is of the opinion that their FSP do not have the capacity at this moment to drive the BD opportunity. BD also provides FSPs to build a competitive services according to P15.

Finding 13: FSPs believe that BD will add value to their organisations.

Finding 14: FSPs have indicate that BD will add value by:

- i) Improve their reliability score in predictive modelling.
- ii) Bring more happiness and satisfaction to their internal staff.
- iii) Identify customer's assets and income in order to get new customers.
- iv) Identify customer's needs and help them to provide better services.
- v) Identify any gaps in their current products.
- vi) Increase FSP's performance and profit margins.

4.9.1.2 RSQ 1.2: What are the factors affecting FSPs when adopting BD analytics for creating a competitive advantage?

IQ 1.2.1: Do you think FSPs are having challenges in adopting BD?

This question was asked to determine if FSPs are having any challenges in adopting BD.

P1 stated that FSPs are experiencing huge challenges in adopting big data. P1 mentions that, "Some of the FSPs don't have data. Some of the FSPs do have data but they don't realize they have it. Some of FSPs have a diverse set of data which has become meaningless to them unless they go the BD process or program. Very few of the FSPs really understand the bigger picture of what data can do for them" (Appendix D1).

P2 argued that FSPs haven't really explore BD however they think there is a lot of data floating around which gets used on an unstructured basis.

P3 mentioned that FSPs are having challenges in adopting big data and said that,

"When they want to record data, they don't know how clean the data is and that is the hardest part for them. They always think it's their systems. If their system can perform or behave like most website, provides customers a functionality to review their medical aid for example that would be great and easy. The challenge is they record a lot of customer's information manually. Customers sometimes capture their information on their system or our consultants call customers capture their information" (Appendix D3).

"FSPs don't understand BD, don't know how to use it or implement it to their advantage and to them BD is foreign" according to P4 (Appendix D4). P5 mentioned that collecting the right data is one of their biggest challenge FSPs are having in adopting BD and their organisation does not understand BD or have no any experience dealing with BD. P6, P7 and P10 were of the opinion that no they don't have any challenges in adopting BD and are currently storing

most of their data in files and backup system. P8 stated that accessing good quality data it's an obstacle for FSPs and good quality data comes at a ridiculous price. P9 mentions that FSPs don't have enough data and said that,

"They have started the business in 2001 and they might be sitting with lots of information or trends that they do not realise that they have or in a position to extract them. At the moment they are using three various systems, for different purposes: for reporting, fees, and CRM perspective and the systems don't necessary talk to each other" (Appendix D9).

P11, P12, P13 and P14 mentioned that they are having challenges in adopting BD because they do not understand BD and they are only aware of it. FSPs are spending large amounts of money in updating their computers and systems specifically their databases.

Finding 15: Most of the FSPs have a huge challenges in adopting BD.

Finding 16: The following are some of the challenges faced by FSPs in adopting BD:

- i) FSPs don't have any data or are not aware if they have it.
- ii) FSPs haven't really explore BD and don't know how to use it or implement it to their advantage.
- iii) FSPs are having challenges in collecting good quality data.

IQ 1.2.2: Do you think FSPs have the right analytical tools or skills to handle unstructured data?

This question was asked to explore if FSPs have the right analytical skills and tools to handle BD. P1 stated that,

"Very few of FSPs are really focusing on BD. They are focusing more on the client's needs or client's solutions without understanding the value of BD. FSPs don't have systems, they have lot of information with them and the information is stored in the spreadsheets. Some of the FSPs have a lot of data however they don't necessary see the value of it because they don't understand BD" (Appendix D1).

P2 mentioned that their organisations doesn't have the right analytical tools or skills to handle unstructured data. P2 said that "They will have to employee or get the external people to assist them with BD. They need both the skills and tools to handle unstructured data" (Appendix D2). P3 stated that in their organisation they do have people with experience to handle unstructured data and have experience to analyse data because they have done it before.

P4, P5, P6, P7, P11, P12, P13 and P14 indicated that they don't have the right analytical tools or skills to handle unstructured data. For example, P8 mentions that in this industry FSPs use excel, stating that,

"Some of the FSPs have internally systems. Other FSPS use python data set. Access to the tools is too much of an obstacle for FSPs unless if there are tools out there that they are not aware of. The challenge FSPs are facing they don't have a skilled personae to analyse the data and use it meaningful for them. There is always a shortage of skills when it comes to that. They do have some guys with the right skills however they still need to train them and they have hired some graduates which they will train them to do data analysis. Most of their employees have skills in data management, quantitative management and mathematical skills BDA. It is a matter of bringing those skills together" (Appendix D8).

P9 indicated that they have the right analytical tools or skills to handle unstructured data. However, they don't think their employee's skills are being used appropriately to generate more investment and more income at the end they need to make money. P12 stated that currently as it stands they do have analytical tools to handle unstructured data. P12 mentioned that,

"They haven't or decided to have analytical tools or systems. However if they decided to have analytic tools or systems, they will make it as project so that they can put more time and focus on it. They will definitely do through research on it to understand all the different tools that are out there in the market and choose specific the one that is tailored or carted to kind of product and services offered by the company" (Appendix D12).

- Finding 17: Many FSPs (9) don't have the right analytical tools or skills to handle unstructured data.
- Finding 18: Some FSPs (4) do have the right analytical tools or skills to handle unstructured data.
- Finding 19: Two FSPs do not focus on BD.

IQ 1.2.3: Do they have a desire to know about big data?

This question was asked to determine if FSPs are willing to understand and know more about BD. P1 mentioned that all the FSPs have a desire to know about big data. P1 claims that "All FSPs think they want more data. They want more knowledge about their current customers. Very few of FSPs would say they don't want more data about their customers, they always want more data about them but they do necessary understands what to do with that data" (Appendix D1). P2 mentioned that their current management do have the desire to know about BD. Their model is more on selling their products. P5 states that FSPs have a huge desire to know about BD and they think it will help them a lot to gain more new customers. P10, P11, P12, P13, P14 and P15 were also of the opinion that FSPs wants to know more about BD. Contrary to P2 and P5, P3 stated that FSPs don't really have a desire to know about BD because the majority of their employees know BD and have dealt with it. P4 mentioned that majority of their employees are on average 50's years of age and stated,

"FSPs are comfortable with operating model and their business going is going well. They don't see the need to expand further. However, they do have younger generation which needs to pursue and make use of BD to grow the business. The younger generation is much isolated and unlike the older generation whom grew their business from social contacts and activities such as playing golf. They need to look into other ways to find customers" (Appendix D4).

FSPs do not know where to start as it probably seems it is something complex and needs more time. No one is ever willing to spend time and money on something they do not know how it will help them. More especially when they don't even know the problem is going to solve.

Finding 20: Some FSPs (4 out of 15) don't have the desire to know about BD.

Finding 21: FSPs have a desire to know about BD.

4.9.2 Research Question 2

RQ2: How can FSPs utilise BD analytics in order to create a competitive advantage?

4.9.2.1 RSQ 2.1: How do FSPs determine the needs of their customers?

IQ 2.1.1: Do you see any benefit in using big data to create competitive advantage in your organisation?

This question was asked to determine if FSPs can use BD to create competitive advantage. Eight FSPs (P1, P2, P4, P8, P9, P10, P12 and p14) perceived some benefit in relation to the benefits that can be realised when using BD. P5 and P11 stated that there is a huge benefit in using BD to create competitive advantage in their organisation. This is because the more they understand their clients the better the service they provide to them. P13 was of the opinion that BD definitely yield to performance. P15 concluded by saying that if FSPs can use BD positively it will enhance growth of their companies. However, P3, P6 and P7 did not see any

short-term benefit on using BD to create a competitive advantage to their organisation.

Findings 22: FSPs are seeing benefits in BD to create competitive advantages for their organisations.

IQ 2.1.2: In your own view, what can BD analytics do to improve your customer's needs? This question was asked to determine if FSPs are seeing the value in using BD. P1 stated that,

"BDA can improve customer's needs by improving their set of solutions noting that products being developed in South Africa becomes extremely complex with multiple numbers of variations. If machines can be in a better position to ultimately match a product, combines it with the customer's needs and be able to put a solution together. Advisors within the FSPs belief that they understand all the products very well. 95 percent of the products they understand very well but there is still a 5 percent of products that are extremely hidden or complex for them, especially if they start looking on things like disability cover, income replacement they are so many implementations that they can actually look at" (Appendix D1).

P2 stated that BDA may help FSPs to be able to identify and allocate tailor made products to their customers and also help them to measure customers' needs accordingly.P3, P5 and P11

mentioned that BDA can be used on product pricing, which will results in cheaper products for customers. P4 stated that in answering this question from time point of view they do not waste their time to market the unnecessary services. P6 and P7 indicated that BDA may not be used for profiling. P6 was of the opinion that,

"BDA can be used to advise or notify customers with the new products and services, noting that FSPs need to have the latest knowledge on the international markets trends and that information is retrieved on the internet through Bloomberg. Example was also provided on how Discovery analyse the national needs and identify the gaps and capitalise on it by developing a product that they make money out of it. But it is a little different in the financial industry based on the regulations" (Appendix D6).

P8 opinioned that BD analytics maybe used to determine the customers really needs as well as customised solutions noting that not all 25 years old have the same objectives or act in the same manner and BD will identify the real nuance of every customer segment. P9 mentioned that a happy customer is for life. Getting more information from customers will be of good advantage, for example what are FSP doing well, where to improve, and identify all the unnecessary things. P10, P12, P14 and P15 responded in agreement to the fact that BDA can improve their customer's needs, determine customer's needs to improve services. P13 said that customers comes to FSPs because they want performance. What basically happen is that, FSP sells to friend or families therefore they have to deliver performance. As long as they perform their customers are happy.

- Finding 23: FSPs agreed that BDA can help them to improve in identifying their customers' needs.
- Finding 24: FSPs mentioned that BDA can be used to identify gaps, assist in terms of identify customers' needs and also help them to inform and price product to remove redundancy on products markets.

4.9.2.2 RSQ 2.2: How can FSPs incorporate BD in their business strategies?

IQ 2.2.1: Does your organisation have a strategy on big data or data analytics?

This question was asked to identify any challenges faced by FSPs to include BD strategies into their organisations. P1 mentioned that they do have a strategy which they are busy working on it. P1 further stated that "There is a different strategy used by their organisation to understand the diversity of the information they had on customers in different datasets and they also understands the value of combining those datasets and starts finding information about their customers that they normal won't make leads on them" (Appendix D1). P15 indicated that they do have a strategy on BDA and mentioned that they analyse all the information that they have and devote lot of time on the information to ensure that whatever the customers' investments they are would be under their management.

FSPs have a premature BD strategy since they are mostly using CRM systems. P2 stated that "They got a system which store all their customers' data on it. All the documentations are stored on it for the purpose of registration and FICA. They also keep their policy documents, variations and personal data" (Appendix D2). P4, P5, P6, P7, P10, P12 and P13 have on the other hand indicated that they do not have BD strategy or BDA strategy. P8 and P9 mentioned that although they do not have a clear defined strategy at this moment. They would like to move towards a centralised data hub, to access more sensible data. P11 and P14 state that they don't have a strategy yet but considering it because technology advances every day and as an organisation they would not want to be left behind in the technology evolution.

Finding 25: FSPs do not have a strategies on BD.

Finding 26: FSPs want to use BD strategy in developing a systems based approach to enhance their growth.

IQ 2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

This question was asked to determine the perceptive view of FSPs towards Strength, Weakness, Opportunity and Threats (SWOT) approach and how the FSPs asses SWOT analysis in order to understand their business climate. FSPs do in some way or another do SWOT analysis at least once a year. It may differ from every three months (P1, P3 & P4). P1 indicated that they assess their SWOT every three months. P2, P8 and P9 said that they do the analysis twice a year. P9, P12 and P15 stated that they conduct their SWOT analysis at an annual basis which is convened every year towards the middle of January and they also have a strategic session.

P6 and P7 mentioned that their SWOT analysis is not based on BD. They do conduct personal introspection. They've indicated that to understand the current business climate they attend webinars with international role players. P10 didn't provide periodicity in terms of conducting their SWOT analysis saying "They do know number of customers they are losing and tare trying to find out why these customers are cancelling their services with them. For any service provided to the customer they always seek for feedback from the customer. All the feedbacks are consolidated and discussed on their meetings" (Appendix D10). P11 mentioned that they do conduct SWOT analysis and it is a continuous assessment they do weekly as things change frequently. They considers this as a less costly approach because they address situation before it expands or becomes costly.

Finding 27: FSPs do perform SWOT analysis.

IQ 2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

This question asked to see how FSPs look at their competition and to see where they need to improve. FSPs evaluate on a regular basis their competition (P1, P2, P3, P5, P6, P7, P9, P10,

P11, P12, P13, & P15). P4 mentioned that they don't have much involvement with other colleagues in the same industry. However, periodically they look at the outputs from other financial advisors and compares it with their own. In 90 percent of the cases they offer better services and are confident on that. P8 said that they don't put their focus too much on the competition side but on their own offering, stating that it may be a draw back. P14 on the other hand felt that there is no need to assess competitiveness because there are not many financial consultancy out there.

Finding 28: FSPs frequently analyse competition and have demonstrate varies approach to the process with some doing it often depending on the organisation.

4.10 Summary of findings

All the findings listed below have been extracted from the research questions and themes are obtained from the findings.

4.10.1 Findings RQ1:RSQ 1.1

Table 4.4: Shows fourteen (14) findings from the RSQ 1.1.

Table 0.4: Fin	dings for	RSQ 1.1
----------------	-----------	---------

Finding No.	Description
Finding 1	Most of the FSPs are aware of BD.
Finding 2	FSPs do not have the knowledge and experience to utilise BD for their organisations.
Finding 3	FSPs do not use BD to analyse customer's needs.
Finding 4	FSPs lack time and reliable resources as a major constraints to implement BD.
Finding 5	FSPs have little or no experience with BD.
Finding 6	Time, cost, resources and skills are the main obstacles preventing FSPs from using BDA.
Finding 7	Collecting reliable data and analysing the data are seen as main challenges.
Finding 8	FSPs categorise their customers based on service they offer to the customers.
Finding 9	FSPs categorise their customers based on their age, affordability, location, timeframe they had business relationship with them and income they generate from them.
Finding 10	Most of FSPs believe they don't have competitors.
Finding 11	FSPs don't manage competitiveness.
Finding 12	Some of the FSPs manage their competitors by: attending roadshows and presentations, lowering advisory fees and provide good quality services.
Finding 13	FSPs believe that BD will add enormous value to their organisations.
Finding 14	FSPs have indicate that BD will add value by helping them to:
	Improve their reliability score in predictive modelling.
	Bring more happiness and satisfaction to their internal staff.
	Identify customer's assets and income in order to get new customers.
	Identify customer's needs and help them to provide better services.
	Identify any gaps in their current products.
	Increase FSP's performance and profit margins.

*RSQ = research sub-question

4.10.2 Findings RQ1: RSQ 1.2

Table 4.5: Shows seven (7) findings from the RSQ 1.2.

Table 0.5: Findings for RSQ 1.2

Finding No.	Description
Finding 15	Most of the FSPs have a huge challenges in adopting BD.
Finding 16	The following are some of the challenges faced by FSPs in adopting BD: FSPs don't have any data or are not aware if they have it. FSPs haven't really explore BD and don't know how to use it or implement it to their advantage. FSPs are having challenges in collecting good quality data.
Finding 17	Many FSPs (9) don't have the right analytical tools or skills to handle unstructured data
Finding 18	Some FSPs (4) have indicated that they do have the right analytical tools or skills to handle unstructured data.
Finding 19	Two FSPs do not focus on BD.
Finding 20	Some FSPs (4 out of 15) don't have the desire to know about BD.
Finding 21	FSPs have a desire to know about BD.

*RSQ = research sub-question

4.10.3 Findings RQ2: RSQ 2.1

Table 4.6 shows three (3) findings from the RSQ 2.1.

Table 0.6: Findings for RSQ 2.1

Finding Number	Description
Finding 22	FSPs are seeing benefits in BD to create competitive advantages for their organisations.
Finding 23	FSPs agreed that BDA can help them to improve in identifying their customers' needs.
Finding 24	FSPs mentioned that BDA can be used to identify gaps, assist in terms of identify customers' needs and also help them to inform and price product to remove redundancy on products markets.

*RSQ=research sub-question

RSQ 2.2: How can FSPs incorporate BD in their business strategies?

Table 4.7 shows here findings of RSQ 2.2.

Table 0.7: The findings for RSQ 2.2

Finding No.	Description
Finding 22	FSPs are seeing benefits in BD to create competitive advantages for their organisations.
Finding 23	FSP agreed that BDA can help them to improve in identifying their customers' needs.
Finding 24	FSPs mentioned that BDA can be used to identify gaps, assist in terms of identify customers' needs and also help them to inform and price product to remove redundancy on products markets.

*RSQ = research sub-question

4.10.4 Findings RQ 2: RSQ 2.2

Table 4.7 shows the findings of RQ2: RSQ 2.2.

Table 0.8: The findings for RSQ 2.2

Finding No.	Description
Finding 25	FSPs do not have a strategies on BD.
Finding 26	FSPs want to use BD strategy in developing a systems based approach to enhance their growth.
Finding 27	FSPs do perform SWOT analysis.
Finding 28	FSPs frequently analyse competition and have demonstrate varies approach to the process with some doing it often depending on the organisation.

*RSQ = research sub-question

4.11 Headline findings

From a total of twenty-eight (28) findings, five (5) headline findings were identified.

The following headline findings have been identified from the findings;

Headline finding 1:	Most of the FSPs are aware of BD
Headline finding 2:	FSPS have challenges in adopting BD, namely:
	the lack of Knowledge and experience
	the lack of time and reliable resources
Headline finding 3:	FSPs are seeing benefits in BD to create competitive advantages for
	their organisations
Headline finding 4:	FSPs mentioned that BDA can be used to identify gaps, assist in terms
	of identify customers' needs and also help them to inform a specific
	product
Headling finding 5:	ESDs do not have a strategy on PD

Headline finding 5: FSPs do not have a strategy on BD

4.12 Categories and themes development

Twenty-eight (28) findings were derived from the interviews. These findings were summarised and then categories were developed on Table 4.9.

Table 4.9 illustrates the findings, and categories and themes. Findings are grouped together and therefore arranged into themes.

RSQ	Finding No.	Finding	Categories	Themes
RSQ 1.1	Finding 1	Most of the FSPs are aware of BD	Adoption	Management
RSQ 1.1	Finding 2	FSPs do not have the knowledge and experience to utilise BD for their organisations	Lack of skill	Training
RSQ 1.1	Finding 3	FSPs do not use BD to analyse client's needs.	Lack of Skill	Training

Table 0.9: Summary of RSQs, 28 findings (28), categories (11) and themes (5)

RSQ	Finding No.	Finding	Categories	Themes
RSQ 1.1	Finding 4	FSPs lack time and reliable resources as a major constraints to implement	Reliable data source and insufficient time	Data Management
RSQ 1.1	Finding 5	FSPs have little or no experience with BD.	Lack of skill	Training
RSQ 1.1	Finding 6	Time, cost, resources and skills are the main obstacles preventing FSPs from using BDA.	Insufficient time, cost effectiveness and lack of skill	Data management, Finance and Training
RSQ 1.1	Finding 7	Collecting reliable data and analysing the data are seen as main challenges.	Reliable data source and lack of skill	Data management and Training
RSQ 1.1	Finding 8	FSPs categorise their customers based on service they offer to the customers.	Customer experience	Marketing
RSQ 1.1	Finding 9	FSPs categorise their customers based on their age, affordability, location, timeframe they had business relationship with them and income they generate from them.	Customer experience	Marketing
RSQ 1.1	Finding 10	Most of FSPs believe they don't have competitors.	Strategic analysis	Management
RSQ 1.1	Finding 11	FSPs don't manage competitiveness.	Strategic analysis	Management
RSQ 1.1	Finding 12	Some of the FSPs manage their competitors by: attending roadshows and presentations, lowering advisory fees and provide good quality services.	Strategic analysis	Management
RSQ 1.1	Finding 13	FSPs believe that BD will add enormous value to their organisations	Adoption	Management
RSQ 1.1	Finding 14	FSPs have indicate that BD will add value by helping them to: Improve their reliability score in predictive modelling. Bring more happiness and satisfaction to their internal staff. Identify customer's assets and income in order to get new customers. Identify customer's needs and help them to provide better services. Identify any gaps in their current products. Increase FSP's performance and profit margins.	Adoption	Management
RSQ 1.2	Finding 15	Most of the FSP have a huge challenges in adopting BD.	Adoption	Management

RSQ	Finding No.	Finding	Categories	Themes
RSQ 1.2	Finding 16	The following are some of the challenges faced by FSPs in adopting BD:	Adoption	Management
		FSPs don't have any data or are not aware if they have it.		
		FSPs haven't really explore BD and don't know how to use it or implement it to their advantage. FSPs are having challenges in collecting good quality data.		
RSQ 1.2	Finding 17	Many FSPs (9) don't have the right analytical tools or skills to handle unstructured data	Lack of skill	Training
RSQ 1.2	Finding 18	Some FSPs (4) have indicated that they do have the right analytical tools or skills to handle unstructured data.	Skilled Personnel	Training
RSQ 1.2	Finding 19	Two FSPs do not focus on BD.	Adoption	Management
RSQ 1.2	Finding 20	Some FSPs (4 out of 15) don't have the desire to know about BD.	Adoption	Management
RSQ 1.2	Finding 21	FSPs have a desire to know about BD.	Adoption	Management
RSQ 2.1	Finding 22	FSP are seeing benefits in BD to create competitive advantages for their organisations.	Strategic analysis	Management
RSQ 2.1	Finding 23	FSPs agreed that BDA can help them to improve in identifying their customer's needs.	Strategic analysis	Management
RSQ 2.1	Finding 24	FSPs mentioned that BDA can be used to identify gaps, assist in terms of identify customers' needs and also help them to inform and price product to remove redundancy on products markets.	Customer experience	Marketing
RSQ 2.1	Finding 25	FSPs do not have strategies on BD.	Strategic analysis	Management
RSQ 2.1	Finding 26	FSPs want to use BD strategy in developing a systems based approach to enhance their growth.	Strategic analysis	Management
RSQ 2.2	Finding 27	FSPs do perform SWOT analysis.	Strategic analysis	Management
RSQ 2.2	Finding 28	FSPs frequently analyse competition and have demonstrate varies approach to the process with some doing it often depending on the organisation.	Market research	Marketing

*RSQ = research sub-question

4.13 Themes

Data collected from fifteen (15) participants that were interviewed have been analysed, which resulted in a summary of twenty-eight (28) findings, eleven (11) categories and five (5) themes. Table 4.10 represents the link between RQs and the themes developed.

Table 0.10: Represents the link between research questions and the themes developed

Research Questions	Themes	
RSQ 1.1, RSQ 1.2, RSQ 2.1 and RSQ 2.2	Management	
RSQ 1.1, and RSQ 1.2	Training	
RSQ 1.1	Data Management	
RSQ 1.1	Finance	
RSQ 1.1, RSQ 2.1 and RSQ 2.2	Marketing	

*RSQ = research sub-question

The above mentioned five themes represented in Table 4.10 will discussed furthermore on the next coming chapter (Chapter 5). The five themes are: i) Management, ii) Training, iii) Data Management, iv) Finance and v) Marketing.

4.14 Chapter summary

This chapter supports a background of the BD on how FSP can utilise it to differentiate their customer segments and use it to create competitive advantage. The chapter represents all the interviewee responses and findings of the research study. All the finds are linked to the research questions and are grouped together in a tabulated form.

The participant included are: two Chief Executive Officers (CEO), Head of Risk Compliance, Chief Financial Officer (CFO), two Managing Directors, Fund Manager, Financial Advisor Category 2, two Chief Operating Officers, Head of business / Chief Operating Officer, two Owners and key Individuals, IT Manager and Principal Owner of the Business.

This chapter concludes with information from 15 participants' response, 14 interview questions, a summary of 28 findings, 11 categories and five themes and seven headline findings.

Themes are: Management, Training, Data Management, Finance and Marketing.

The headline findings are:

Headline finding 1: Most of the FSPs are aware of BD.

Headline finding 2: FSPS have challenges in adopting BD, namely:

- i) Lack of Knowledge and experience
- ii) Lack of time and reliable resources
- Headline finding 3: FSPs are seeing benefits in BD to create competitive advantages for their organisations.

Headline finding 4: FSPs mentioned that BDA can be used to identify gaps, assist in terms of identify customers' needs and also help them to inform a specific product.

Headline finding 5: FSPs do not have a strategy on BD.

The next chapter five (5) discusses the findings in terms of the themes developed.

CHAPTER FIVE: DISCUSSION



Figure 0.1: Layout of Chapter 5

5.1 Introduction

The chapter discusses the themes developed from the findings and linked to the research questions. The discussion arises against the background of the aim of the study, namely to explore how FSPs can use BD to differentiate their customer segments and to create a competitive advantage. Research questions together with the sub-questions are addressed within each theme. Themes are addressed and considered in the following order: i) management, ii) training, iii) data management, iv) finance and finally, v) marketing. The chapter ends with a summary.

5.2 Theme 1: Management

Management is an earliest practice in governmental, religious, military and other types of early organisations (Bowden et al., 2020). Wang (2020:38) defines management as "planning, organising, leading and controlling". For this study the definition of management include four characteristics as mentioned by Bowden et al. (2020:33) namely:

- i) It pays attention to costs and labour value.
- ii) Management includes maximising competitive advantage.
- iii) Focuses towards numerous competitive markets for sustainability.
- iv) Management is supported by legal frameworks that guarantee contracts and protect property and individual rights.

Theme 1 relates to the following research questions and research sub questions: RQ1: "What are the challenges FSPS face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?", RSQ 1.1: "What are the perceptions of

FSPs towards the value of BD for their businesses?", RSQ 1.2: "What are the factors affecting FSPs when adopting BD analytics for creating a competitive advantage?", RQ2: "How can FSPs utilise BD analytics in order to create a competitive advantage?", RSQ 2.1: "How do FSPs determine the needs of their customers?" and RSQ 2.2: "How can FSPs incorporate BD in their business strategies?"

Finding 1 indicates that most of the FSPs are aware of BD. IT provides extensive opportunities to FSPs for automating, informing and business transformation (Paré et al., 2020). Endorsed by the increasing importance of IT to business performance, literature has been mainly done on management practices that affect the quality and range of IT's impact (Rahimi et al., 2016). When attempting to deliver value from the organisation's IT investment, it is a must to comprehend top management's perceptions of the centrality or strategic significance of IT in their organisation (Paré et al., 2020). According to Paré et al. (2020:01), IT centrality refers to "the idea that IT assets might not be viewed as strategic by top executive's goals and implementing its strategies".

A positive outcome of this study is that FSPs are aware of BD. BD offers significant advantages for FSPs in order to gain competitive advantages and assist FSPs in making better recommendations and decisions for the customers. FSPs can utilise BD to analyse customer's needs and help in developing products that meet customers' needs. BD can be utilised by FSPs to seek innovative opportunities and new business opportunities. BD has been identified by Rahman (2017) as powerful competitive tool for FSPs to become more competitive and making better decisions. BD will not just bring opportunities for FSPs but will address challenges faced by FSPs to acquire knowledge in creating and implementing BD strategies. FSPs can utilise BD to differentiate their customer segmentation in order to customise their products and services offering.

IT management is recognised by Alt et al. (2020:610) as an enterprise function for planning, organising, and controlling IT resources such as people, processes, technology and data. Furthermore, Alt et al. (2020) state that software and IT services are the most important areas with IT management, and have a fundamental responsibilities for development, operations and managing third party software providers. In order for management to be fully functional, managers must step up, lead their teams efficiently with lot of confidence and maintain motivation (Ng et al., 2020). Finding 10 states that "most of FSPs believe they don't have competitors" (Appendix D3). IT management can assist FSPs to be more competitive in a more demanding institutional environment that is both service based and knowledge based (Paré et al., 2020). According to Bowden et al. (2020) management entails maximising competitors from all angles and use different types of resources and capabilities to create competitive advantage (Jahanshahi & Bhattacharjee, 2020).

In today's evolving and competitive world, most organisations strive towards better ways to increase their competitiveness and leadership compared with their competitors (Jahanshahi & Bhattacharjee, 2020). According Mukunthan and Selvakumar (2019) FSPs ability to succeed over the competitors is a way to describe why some of the FSPs are more successful than others, and it is fundamental important to the survival in a highly competitive business world. There are positive results in better performance than competitors, if FSPs can identity and analyse their competitors. Finding 11 states that "FSPs don't manage competitiveness" (Appendix D8). Findings 12, 22, 23, 25, 26 and 27 indicate that BD can create competitors may results in FSPs lacking competitiveness improvement. Jahanshahi and Bhattacharjee (2020:04) have identified the advantages of competitiveness improvement namely:

- FSPs can provide better quality of products or services compared to their competitor's products and services.
- ii) FSP's managerial capability will be much better than their competitors.
- iii) FSP's profitability is better.

manage threats.

iv) FSP's company image is better than that of the competitors.

v) The competitors are struggling to take the place of the FSP's competitive advantage. Enhancing competitiveness empowers FSPs to produce some superior services and products to their customers (Rahimi et al., 2016). Rahimi et al. (2016) refer to enhancing competitiveness as a degree to which FSPs can deduct cost, exploit opportunities and

Finding 13 and Finding 14 state that FSPs believe that BD will add enormous value to their organisations. BD has been identified by Kaisler et al. (2020) as a developing phenomenon with a huge implications and effects on business strategy, profitability and process improvements. According to Faroukhi et al. (2020) value is the coherent goal of any organisation. BD can add value to FSPs by identifying potential competitive advantage and organise processes to create added value in breaking down strategically importance activities to comprehend all the impacts (Faroukhi et al., 2020). Line et al. (2020) mention that FSPs can utilise BD to generate an extra value from advertising, customer relationship management, marketing research, resale and trading of their products. Furthermore BD can be utilised to develop or improve the efficiency of loyalty program platforms and BD has value when utilised to improve advertising from targeted customers (Line et al., 2020).

Findings 15, 16, 19, 20 and 21 indicate challenges in adopting BD by FSPs. Chuah and Thurusamry (2021) note that data protection and data privacy are the biggest challenges faced by FSPs when adopting BD. The protection of data and security issues are lowering BD adoption. Another challenge with BD adoption by FSPs is organisational culture. FSPs do not want to share information towards each other but they need to start building a non-hierarchical culture of trust (Walker & Brown, 2019). Another challenge faced by FSPs with BD adoption

is infrastructure. FSPs must start investing in proper server equipment, high processors or middleware in order to manage BD (Chuah & Thurusamry, 2021).

Furthermore Chuah and Thurusamry (2021) mention that FSPs are struggling to interpret data quickly enough due to improper infrastructure and are faced with challenges on responding to complex queries with immense columns of data. According to Faroukhi et al. (2020) another challenge faced by FSPs is the limited existence of analytical software tools on the market. This leads to a struggle to discover analytical solutions with powerful and friendly interactive analytical competences tools. IT management and a complete execution of BD also need a team effort that entails lean and agile practices (Ghavami, 2020). A team effort is needed because no vendor solution, individual employee fulfils all the BD analytics requirements of the organisation and learn approach must be implemented to avoid duplication of work, process waste and discordant systems (Ghavami, 2020).

5.3 Theme 2: Training

Training is regularly linked with large effect sizes for outcomes in the areas of knowledge, skills and behaviours (Cooper et al., 2021). Sicari (2021:56) defines training as "key word and learned personnel keep learning from the way they shape the exam on the basis of a tailored approach to clinical non-invasive diagnosis". According to Cooper et al. (2021) training is used to improve individual competence and improved team's performance. For this study the definition of training is a technique designed to make a measurable difference to the employee performance and the entire business success.

Theme 2 relates to the following research questions and research sub questions: RQ1: "What are the challenges FSPS face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?", RSQ 1.1: "What are the perceptions of FSPs towards the value of BD for their businesses?", RSQ 1.2: "What are the factors affecting FSPs when adopting BD analytics for creating a competitive advantage?".

Finding 2 has indicated that FSPs do not have the knowledge and experience to utilise BD for their organisations, and Finding 5 FSPs have little or no experience with BD. However, in order for FSPs to gain knowledge and experience in utilising BD, FSPs must start investing in BD trainings to gain knowledge and develops talents for their organisations. FSPs must start recruiting fresh people or graduates and provide them with experience, training, skills and knowledge. Having experience and knowledge on BD will results in FSPs becoming a more competitive advantage organisation. Furthermore FSPs will improve their business performance, grow their innovation capabilities and achieve better competitive advantage. Gaining knowledge and experience in BD will assist FSPs to identify systems compatible with the following requirements such as security, reliability, manageability, scalability, redundancy and usability before implementing them.

An appropriate BD training for FSP's employees provide them with a horizontal skill set while also providing them the soft skills associated with the current business changes (Solé-Beteta et al., 2021). Furthermore Solé-Beteta et al. (2021) mention that training of FSP's employees on BD, calls attention to the importance of encouraging the ability of their employees to perform core delivery objectives. According to Line et al. (2020) providing training to employees is all about educating employees to achieve better performance, empower employees to prepare for promotion and enable them to improve performance and productivity. Rothwell et al. (2020) distinguish training to be a push strategy meant to improve job performance and upturn production with a problem-centred strategy.

FSPs do not use BD to analyse client's needs, FSPs must start implementing BD strategies. If they don't, FSPs are facing a high risk of not be able to identify fraudulent activities within their organisations. BD will assist FSPs to customise their products and services. FSPs can utilise BD to improve their reliability. Furthermore BD can assist FSPs with effective marketing of their products. FSPs can utilise BD to identify any trends of their customer's needs and assist them to provide products and services suitable for their customer's needs. Using BD will assists FSPs to predict the potential churns from their existing customers.

Time, cost, resources and skills are the main obstacles preventing FSPs from using BDA. FSPs are looking for better ways to make organisational sponsored training programmes more attractive to their employees (Rothwell et al., 2020). In order to achieve this, FSPs must pursue accreditation equivalent to some training initiatives for university credit and make sure that their training are provided with highest standards of quality (Cooper et al., 2021). For better preparation on accreditation, FSPs must establish number of policies and procedures to administrate their training functions (Rothwell et al., 2020). According to Solé-Beteta et al. (2021) accreditation means that training and development programs are meeting the research based standards and are provided in good quality. Furthermore Rothwell et al. (2020:02) defined accreditation as a "process of independent attestation to the adherence of standards that meet the requirements of a profession".

Collecting reliable data and analysing the data are seen as main challenges. Finding 17 indicates that many FSPs (9) don't have the right analytical tools or skills to handle unstructured data. Finding 18 has indicated that some FSPs (4) do have the right analytical tools or skills to handle unstructured data. Buying data from other organisations can lower the cost of collecting data and FSPs can utilise that data to achieve their business objective. FSPs can collect data from various sources, use it to improve their capability to narrow their customer segmentation and also use it to tailor down products and services to their customers. The challenge faced by FSPs is that they don't have skilled personae to analyse the data and utilise it meaningful. FSPs must train their existing employee to do data analysis. An accredited training instils and prepares employees for future marketplace requirements, and builds a competent talents by setting learning goals for their performance (Sicari, 2021). The constantly

increasing amount and fast development of tools, techniques and technologies linked to BD require a wide skill set and training in order to deep understand BD domains ranging from engineering to business (Solé-Beteta et al., 2021).

According to Line et al. (2020) providing training to BD professionals is a way of equipping them with horizontal skill set and providing them with soft skills linked to the current society. Overall, training employees in the organisation to accept and solve complex problems or encounters in a multidisciplinary situation utilising collaborative strategies encompasses a wide skill set which results of broad interest in the field of developing technologies such BD (Solé-Beteta et al., 2021). Education innovation has been mentioned by various researchers as an efficient, effective and successful way to solve teaching and learning challenges in the modern education (Kaufmann, 2019; Sicari, 2021; Solé-Beteta et al., 2021). According to Cooper et al. (2021) there is a growing interest in FSPs to explore and consider new strategies to effectively train and shape its employees to master inaccurately connected disciplines, add value to their evolution industry and interact with different radically different profiles. On the other hand providing training to your employees, even utilising modern active learning strategies and project based training might take a considerable amount of time off for employees, which is typically not available from FSPs (Chuah & Thurusamry, 2021). When FSPs consider training of employees, it is necessary to consider training environments that best fit the needs of their employees. The environment should be open to the presence of all kinds of innovative technologies, methodologies and have a capacity to adapt employees coming from technologically, social, and physical perspective (Paré et al., 2020).

FSPs that provide regular quality training to their employees are naturally more scalable than FSPs that do not (Rothwell et al., 2020). As a result, FSPs that provide quality training to their employees become more attractive employer of choice because training practices have definable and lifelong benefits (Bowden et al., 2020). Training guarantees an investment which signifies the value proposition for an employee, an organisation and a customer (Kaufmann, 2019). It is important for FSPs managers to make sure the employee's training is fundamental aligned with the organisation's strategic goals, by doing so FSP will challenge its competitors and succeed its mission (Wang, 2020). Therefore, training plans that are aligned to the organisational strategies guide managers to design training modules that are beneficial to their employees (Rothwell et al., 2020).

For better operational compliance achievement FSPs must incorporate training (Hosseini et al., 2019). Furthermore Hosseini et al. (2019) state that training employees is regarded as mandatory educational exercise which must be conducted in order for an organisation to be compliant. FSPs must utilise training as an opportunity to improve and gain knowledge and talents essential in the people presently employed in the industry (Rothwell et al., 2020). FSPs must use training as a method to maintain and develop talents within their organisations (Paré et al., 2020). According to Cooper et al. (2021) training helps employees to meet organisation's

performance requirements, help them to continue meet requirements as work conditions keep on changing and prepare them for to meet requirements for future higher levels of responsibility.

In executing of strategy, offering training assists employees to be ready for tomorrow's jobs and to predict a new future (Hosseini et al., 2019). According to Rothwell et al. (2020:38) training can be structure using three basic models namely:

- Centralised training organisation: Where all programs, resources and professionals are housed as one unit, and all curriculum training decisions, programs and methods of delivering trainings are made.
- ii) **Decentralised training organisation:** Where all learning and training methods are housed from different operating departments throughout the organisation.
- iii) **Embedded training organisation:** Distinguish by organisational strategic alignment of learning, either some of training are housed centralised or decentralised.

Therefore organisational success is depend on having structured training department that is accountable for both internal and external training efforts (Solé-Beteta et al., 2021).

5.4 Theme 3: Data Management

DM defined by Alhajj et al. (2020:05) as "a vital requirement to secure data while also allowing users to query and retrieve the data". Nda and Tasmin (2019:5009) describe DM as a "reality that represents a set of challenges involving BD modelling, storage, retrieval analysis and visualisation for several areas in organisations". Several techniques has been suggested in the literature about BD management, analysis and data representation but very few articles are dealing with unstructured data in FSP (Kaufmann, 2019). FSPs can utilise DM to comprehend valuable information in BD set that are both structured and semi-structured coming from different sources (Kaufmann, 2019).

Theme 3 relates to the following research questions and research sub-questions: RQ1: "What are the challenges FSPS face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?" and RSQ 1.1: "What are the perceptions of FSPs towards the value of BD for their businesses?".

FSPs lack time and reliable resources as a major constraints to implement. Finding 7 states that collecting reliable data and analysing the data are seen as main challenges. DM is used to manage and store massive amounts of heterogeneous data coming from multiple sources, analysed and served to end users and external applications (Davoudian & Liu, 2020). The growing number of BD technologies and affordable infrastructures have encouraged FSPs to strengthen their business value by launching BD systems (Kalra & Ranjan, 2021). The widespread adoption of smartphones, social media platforms and wearable technologies does not only increase the amount of available data but multiplying new data sources for FSPs (Horita et al., 2020). According to Horita et al. (2020) BD can be utilised by FSPs to transform

the entire business process and assist them to provide greater support for decision making in different contexts such as business management and marketing. Figure 5.1: are some of data sources for BD adopted by (Emrouznejad & Charles, 2019).



Figure 0.2: Sources of BD (Emrouznejad & Charles, 2019:02)

The range of valuable data coming from different veracity can be easily collected at a high speed in numerous real-life application (Mai et al., 2020). Social networking sites (e.g., Twitter, Facebook & LinkedIn) are identified by Mai et al. (2020) as rich source of complex BD. Conducting manual analysis on data can be time consuming, impractical in the widespread of BD source and computerised data analytics tools are desirable (Kalra & Ranjan, 2021:12221). There is a need to highlight some of the data science solution for BDA, namely:

- Data extraction: FSPs can utilise a standard-tier search API platform to search all social networks published over a certain period and then collect that data. Alternatively FSPs can utilise the below enterprise-tier API platform for collecting any published data:
- ii) **Amazon Web Services (AWS):** which is an online server that gives cloud computing with numerous CPU's memory, storage and networking ability.
- iii) **RStudio:** is an open source programming software used for statistical computing and graphics.
- iv) **Data filtering:** FSPs can utilise any preference to filter their data. For better filtering and reduction on workloads, data science solution are providing the following features:
- v) Generating acronyms: e.g., "Doctor" can be generating into "Dr".
- vi) Enumerating variants: "Doctors" enumerated as a plural form of "Doctor".
- vii) Generating synonyms: e.g., "GP" generated as a "General practitioner".
- viii) **Data analysis:** Once FSPs have collected and filtered their data, they can analyse their filtered data in order to extract interesting knowledge and valuable information

that was previous unknown by the organisation. FSP can utilised the following techniques:

- ix) Data mining
- x) Sentimental analysis

xi) Data and result visualisation

BD processing require data collection, management, analysis and display (Gupta & Rohil, 2020). BD processing can be utilised to gather data from the data source and pre-process it to produce high quality data set for successive processes (Alzahrani & Sadaoui, 2020). Since there are different data source for BD, there may be different models contradicting to each other, therefore it is fundamental important to clean the data when performing data integration process (Gupta & Rohil, 2020). Figure 5.2 is an example of BD technology architecture.



Figure 0.3: BD technology architecture (Gupta & Rohil, 2020:288)

Finding 6 state that time, cost, resources and skills are the main obstacles preventing FSPs from using BDA. BD require real-time applications with strong computational functionality so that FSPs can process large amount of data in less than amount of time (Alzahrani & Sadaoui, 2020). Therefore stream processing mode are appropriate for real-time applications (Mai et al., 2020). When using distributed architecture the cost of storage gets reduced and there is an increase on data quantity (Paré et al., 2020). FSPs can lower their processing cost by implementing virtual clusters since they have an advantage of low cost, flexible construction, easy to manage and are more convenient for completing various processing tasks when performing BD analysis (Adnan et al., 2020). FSP are faced with a challenge of BD scarce skilled resources to meet their business needs and these challenges have an indirect or direct impact on one or more factors in their organisations (Yadav et al., 2020). In order for FSPs to overcome this change they must recruit fresh people and provide them experience, skills, knowledge and train them for their future needs (Aswale & Mukul, 2020).

5.5 Theme 4: Finance

The term finance is mostly used in boardrooms, and FSPs managers' needs to understand the subject and finance language if they want to communicate with authorities within their businesses (Anurag, 2020). Purcell et al. (2020:444) define finance as "ways of breaking down physical basis of earnings that become central to securitised flows of funds generated from different asset classes". For this study it is imperative to also define accounting since finance is interrelated to accounting. Accounting is defined as "a system to document all the information about the business transactions and events" (Chen et al., 2020:1197). Furthermore, Chen et al. (2020) indicate that accounting can be utilised by FSPs to breakdown their life into arbitrary periods (quarters and years), summarise the statements of the organisation's financial position and provide performance statements to their stakeholders.

This theme relates to the following research questions and research sub questions: RQ1: "What are the challenges FSPS face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?" and RSQ 1.1: "What are the perceptions of FSPs towards the value of BD for their businesses?".

According to Anurag (2020:65) financial statements consist of three key components, namely:

- i) They assist on getting standardised reports for external stakeholders.
- ii) Are used for tax accounting in compliance with Internal Revenue Service (IRS) rules for computing taxes payable.
- iii) Are utilised for managerial accounting to provide custom reports for internal decision making.

Financial statements have a major role of providing information about the organisation's economic events ranging from acquiring funds from investors to run business operations in order to generate profit which can be either reinvested to the business or distributed to the investors as dividends (Purcell et al., 2020). On the other hand, financial analysis assists organisations to take investments and management decisions (Anurag, 2020). According to Yadav et al. (2020) both financial statements and financial analysis have a common goal of identifying organisations strengths and weaknesses. For decision making and decision taking processes organisations are using management accounts, it helps in management record, plan and control the activities of the business (Castelli et al., 2021).

Time, cost, resources and skills are the main obstacles preventing FSPs from using BDA. FSPs must face their current substantial challenges when using data and must be aware of them before BD can be utilised to produce cost effectiveness in their organisations (Wordsworth et al., 2018). BD generated from different sources can be utilised by FSPs to provide evidence, benefits of decision making and can be utilised to produce cost-effective models on detailed analysis (Wordsworth et al., 2018). Moving data to the cloud for operational processing should guarantee performance and availability from various locations while minimizing the cost (Xiao et al., 2019). Cloud technologies provide powerful and cost-effective
solutions to handle the increasingly high velocity of BD generated from various data sources (Castelli et al., 2021). FSPs can rent the storage from public cloud while taking advantage of cloud computing benefits such flexibility and pay-as-you go model (Xiao et al., 2019). According to Huang et al. (2018) Cloud computing has an ability of providing low cost storage solution to cover massive storage requirement for BD data storage. FSPs can lower their cost of collecting data if they buy the existing data from other organisations and use it to achieve their business objective or if they can obtain their available data sets.

5.6 Theme 5: Marketing

Using data to gain insights for marketing is not a new concepts, for more than 20 years database marketing and analytical CRM has been existing (Gentsch, 2019). According to Steinhoff et al. (2019) marketing is a learning technique for keeping certain ideas and concepts in mind continually and by asking certain dynamic questions regularly. Lamb et al. (2020:04) define marketing as "an activity, set of institutions, and process for creating, communicating, delivering and exchanging offerings that have value for customers, clients, partners and society at large". Furthermore, majority of people thinks marketing is all about making products available in shops, arranging the display of products and maintaining inventories for future sales, therefore marketing means expecting and sustaining customer's needs by means of mutually beneficial exchange processes, and gaining profit margins and becoming more effectively than competitors by means of efficient managerial process (Lamb et al., 2020).

This theme relates to the following research questions and research sub questions: RQ1: "What are the challenges FSPS face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?", RSQ 1.1: "What are the perceptions of FSPs towards the value of BD for their businesses?, RQ2: "How can FSPs utilise BD analytics in order to create a competitive advantage?", RSQ: 2.1: "How do FSPs determine the needs of their customers?" and RSQ 2.2: "How can FSPs incorporate BD in their business strategies?".

According to Andronie et al. (2019) FSPs are always looking forward to improve their service offerings, and majority of these FSPs do know data driven marketing technologies are vital to make it happen. Furthermore, Andronie et al. (2019) state that 78 percent of marketers are working under pressure to become more data driven. There is a huge need for FSPs to start working on data driven strategies and FSPs must invest more on IT technologies because marketing is becoming technology based and mastering BD is vital for FSPs if they want to achieve competitive advantage (Liu et al., 2021). However it is important for FSPs to to their consumers and how they can benefit from utilising social media to obtain more understanding on customer engagement (Liu et al., 2021). FSPs need to start identifying sustainable

marketing strategies that will result them to generate a competitive advantage and innovation in all their products (Gentsch, 2019).

FSPs can customised marketing by offering services to individual customers or by direct sending products to only targeted customers (Lies, 2019). For better implementing marketing strategies the use of marketing intelligence must be adopted by FSPs, this can assist FSPs to develop all the insights gained from data for use of marketing decision (Lisa, 2021). According to Buhalis and Volchek (2021) integrating technology into FSP's strategy increase marketing communications and organisation performance. Marketing performance analytics need a comprehensive approach to identify customer's decision making changes which are influenced by marketing communication (Buhalis & Volchek, 2021). Collecting data from various sources have developed substantial opportunities for FSPs to utilise BDA in developing programmatic marketing techniques for online displays of their products and services (Jabbar et al., 2020).

To challenge the current expectations of BD and programmatic marketing FSPs must use an approach known as problematisation (Buhalis & Volchek, 2021). Problematisation is described by Jabbar et al. (2020:560) as "a considered calculated approach to communicate existing assumptions and test them with unique ideas". FSPs can implement the problematisation approach to challenge the current marketing things and create new ways marketing thinking. Furthermore, FSPs can utilise problematisation approach to spot the gaps for BD and programmatic marketing in their organisations.

Finding 8 states that FSPs categorise their customers based on service they offer to the customers. FSPs categorise their customers based on their age, affordability, location, timeframe they had business relationship with them and income they generate from them. In this research, FSPs are segmenting their customers based on the service they offer to them which propose a good quality of clustering analysis. Clustering is a group of algorithms utilised to divide records in a database according to an extent similarities (Collica, 2019). Clustering analysis is utilised by most researchers to segment the customer markets, and it is conducted and the influencing elements, with subjective norm, attitudes, face consciousness, brand extension and observed risks are categorised into isolated groups (Zhang et al., 2020). Utilising clustering analysis FSPs can gain better return on investment because they can tailor their services to customers that are mostly likely to purchase the offered service (Yang et al., 2019). To achieve a better return on investment, FSPs need to start mass customisation than mass marketing. Customisation is commonly known as one to one marketing, whereby each single customer, product offering and service offering is tailored to a specific customer's needs (Yang et al., 2019).

The question FSPs might be asking themselves is, how to define similar customers? They are various ways to define similar customers in an organisations, FSPs can define similar customers based of the type of services purchased by customers, numbers of policies have

96

taken by customer or the amount of revenue generated from the customer (Zhang et al., 2020). Other metrics can be, the number responds made by a customer to a specific marketing campaign over a time period, offers they purchased on their Web site, amount customers spent on their online platform and the number of information search made by the customer about their services or products (Collica, 2019).

BDA can be used to identify gaps, assist in terms of identify customers' needs and also help them to inform and price product to remove redundancy on products markets. Mining and analysing BD can give FSPs a better understanding of their business patterns and developments, this can help them to drive operations efficiently and gain competitive advantage across their different departments such as sales and marketing, product designs and IT (Schosser, 2020). The 3V's are most commonly characteristics of BD and FSPs can utilised them to develop marketing strategies.

5.6.1 Volume

Volume is the amount of data and it contains data from traditional and non-traditional sources. FSPs must analysis all their transactional data for example the products purchased by their customers, the frequency of each product sale and all of that is BD. Then FSPs can couple their traditional data with the digital data from marketing platforms, add them all and that will results to amount of data that can be used for marketing.

5.6.2 Velocity

Velocity is the speed of purchase information generated and the flow of data into the FSPs sales journeys. FSPs can analysis the growth of data coming to their organisation. Driven data marketing can be utilised by FSPs to collect, analyse and process insights from BD and also drive customer engagement to improve marketing (Schosser, 2020).

5.6.3 Variety

Variety will be the amount of data currently available for FSPs to utilise it for marketing. Marketing strategies can be implemented for both end user and data sources. FSPs must combine both their transactional data and unstructured data coming from different sources so that marketing can use them effectively. For better customer engagement FSPs need to connect both their analytics analysis and marketing communication features in return they will gain more profitable, satisfying and long lasting relationship with their customers (Zhang et al., 2020).

FSPs frequently analyse competition and have demonstrate varies approach to the process with some doing it often depending on the organisation. BD has been identified by researchers as one of the breakthrough technological developments, techniques such as text analytics, machine learning, predictive analytics and data mining can be utilised by FSPs to gain new insights resulting in better and quicker decisions, and yield to both economic and social value

97

(Emrouznejad & Charles, 2019). By using BD FSPs will gain a better impact on employment growth, productivity, and development of new products and services (Lamb et al., 2020).

5.7 Chapter Summary

The themes are discussed in perspective of the findings and categories derived from the study. Five themes emerged on this study, along with references from numerous sources and some these references are participants interviewed for this study. As per the findings and categories, all themes have been mapped per research question. The following are all the themes discussed on this chapter: i) management, ii) training, iii) data management, iv) finance and finally, v) marketing. The respective themes had a significant role in addressing the research questions. Theme 1 (Management) revealed that IT management can assist FSPs to be more competitive in a more demanding institutional environment that is both service based and knowledge based. Theme 2 (Training) shows that providing training to employees is all about educating employees to achieve better performance, empower employees to prepare for promotion and enable them to improve performance and productivity. Theme 3 (Data Management) indicates that BD can be utilised by FSPs to transform the entire business process and assist them to provide greater support for decision making in different contexts such as business management and marketing. Theme 4 (Finance) reveals that BD generated from different sources can be utilised by FSPs to provide evidence, benefits of decision making and can be utilised to produce cost-effective models on detailed analysis. Furthermore Theme 5 (Marketing) revealed that FSPs must start working on data driven strategies and FSPs must invest more on IT technologies because marketing is becoming technology based and mastering BD is vital for FSPs if they want to achieve competitive advantage.

CHAPTER SIX: CONCLUSIONS AND RECOMMENDATIONS



Figure 0.1: Layout of Chapter 6

6.1 Introduction

Chapter six (Figure 6.1) gives a i) conclusion, ii) addresses the research questions, iii) suggests recommendations, iv) addresses research reflections and v) consider future research.

6.2 Conclusion

FSPs are mostly aware of BD. BD offers a significant advantages for FSPs in order to gain competitive advantages and assist FSPs in making better recommendations and decisions for the customers. Collecting more and detailed data can boost FSP's performance and adopting BD can improve FSP's capability to narrow their customer segmentation and can furthermore enhance FSP's creativity in tailoring products and services to their customers. FSPs can benefit from BD to develop better retentions strategies and assist in quality decision making. Utilising BD during innovation life cycles can assist FSPs to gain better lessons throughout all the phases. Furthermore at the end of each phase FSPs can utilise opportunities learn from the successes and failures of it.

BD is the next stronghold for innovation, competition and productivity. BD can help FSPs to reduce inefficiencies in product delivery, improve marketing and access on products and services, increase qualities on products and services, and make more personalised products and services for new and existing customers. FSPs can use BD to support their core businesses by analysing all the transactions to detect if there was any fraudulent transactions or money laundering. BD can have a substantial influence on value creation and competitive

advantage for SMMEs. This can be done by improving or creating new ways on interacting with customers or developing new products and services strategies to raise profits.

6.3 Research questions answered

6.3.1 RQ1: What are the challenges FSPs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

The challenges faced by FSPs when using BD analytics in building and differentiation customer segmentation to create a competitive advantage are listed below:

- i) FSPs do not have knowledge and experience to utilise BD for their organisations.
- ii) FSPs do not use BD to analyse client's needs.
- iii) FSPs lack time and reliable resources as major constraints to implement BD.
- iv) FSPs have little or no experience with BD.
- v) Time, cost, resources and skills are the main obstacles preventing FSPs from using BDA.
- vi) Collecting reliable data and analysing the data are seen as main challenges.
- vii) FSPs don't manage competitiveness.
- viii) Most of the FSPs have a huge challenges in adopting BD.
- ix) The following are some of the challenges faced by FSPs in adopting BD:
- x) FSPs don't have any data or are not aware if they have it.
- xi) FSPs haven't really explore BD and don't know how to use it or implement it to their advantage.
- xii) FSPs are having challenges in collecting good quality data.
- xiii) Many FSPs (9) don't have the right analytical tools or skills to handle unstructured data.

6.3.2 RQ2: How can FSPs utilise BD analytics in order to create a competitive advantage?

FSPS can utilise BD analytics to create a competitive advantage by following the below considerations:

- i) FSPs agreed that BDA can help them to improve in identifying their customers' needs.
- ii) FSPs mentioned that BDA can be used to identify gaps, assist in terms of identify customers' needs and also help them to inform and price product to remove redundancy on products markets.
- iii) FSPs want to use BD strategy in developing a systems based approach to enhance their growth.
- iv) FSPs frequently analyse competition and have demonstrate varies approach to the process with some doing it often depending on the organisation.

In conclusion, there is a need for FSPs to be innovative, create new business opportunities and business models to be competitive since most corporates have already adopted digitisation strategies in their quest to create a competitive advantage in the industry they are functioning in. BD is a powerful competitive tool for any organisation to become more competitive and for better decision making.

6.4 Aim of the study

The aim of this study was to explore how FSPs can use BD to differentiate their customer segments and to create a competitive advantage. During the data collection and analysis it was identified that FSPs don't have BD knowledge, experience and are lacking time, cost and reliable resources to implement BD. BD can help FSP's to narrow their customer segmentation and enhances FSP's creativity in tailoring products and services to their customers. This was achieved by answering the two main research questions stated in section 6.2.

6.5 Recommendations

Literature reporting on BD has primarily focused on larger businesses. FSPs are lacking the financial strength to invest in technology. However, the critical role of FSPs in the local and national economic development means that FSPs also need to understand and implement BD strategies into their businesses. Implementation of BD strategies by FSPs can revolutionise management. With BD dominate FSPs can take advantage on the market share, customers and profits away from those that are not utilising BD. FSPs will gain a competitive advantage, make more driven BD decisions making.

To manage BD, FSPs must establish a strategic plan for computing infrastructure, organisational procedures, policies and rules relevant or related to BD. FSPs must construct innovative solutions for BD rapidly. The study outcomes have pointed various opportunities and challenges FSPs can use BD to differentiate their customer segments and to create a competitive advantage. Therefore recommendations for creating a competitive advantage are summarised as follows:

- i) FSPs can improve their customer services through customer's purchase analyses.
- ii) FSPs can reduce costs and increase profit through cost and profit analyses.
- iii) FSPs can enhance marketing strategies through marketing campaign analyses.
- iv) FSPs can achieve sustainability through long term risk analyses.

6.6 Reflection

Fifteen (15) SMMEs (Chapter 1, section 1.2) of Financial Institutions in the Western Cape and Gauteng were interviewed. Twelve (12) FSPs are based in the Western Cape Province between Cape Town and Worcester and three (3) are based in Gauteng Province between Johannesburg and Pretoria. FSPs from other provinces were not considered due to time and financial constraints. Interviews were administered to the FSP's decision makers' i.e. (Chief Information Officers, Chief Executive Officer, IT Managers & Program Managers etc.).

Acquiring permission from the participants to carry out the research was an essential requirement. The researcher used an email to send out invitations to all the FSPs (Appendix B) but in most times physical visits to the FSPs was vital. Electronic follow-ups via the use of email were done before interviews took place. During the interview process all the participants were asked to freely accept and signed the consent letter to participate of the research. All the participants signed a letter of consent to participant on the research. In this study, semi-structured questionnaire was used for data collection, all interviews were guided by an interview guide (Appendix A). Some of the participants requested interview questions prior to interview so answers may have been thought out which may have mitigated the richness of answers ascertained. All interviews were recorded, and participants were asked upfront before the interview starts to grant the interviewer the permission to record the interview. The records taken from the interviews were transcribed and texted analysed for additional analysis using inductive approach.

Preliminary to the interview process, some challenges were encountered such as participants cancelling the interviews at the last minute after months of effort to secure interview with them. Zoom Meeting technology was utilised for all the 3 participant based in Gauteng, since interview couldn't be conducted in person due to some financial constraints and COVID-19 restrictions rules. In general this was a good phase of the project, it has taught the researcher the following principles accountability, drive and commitment. Some aspects that could have been executed better include improving more on time management and multi-tasking skill, this would have assisted the turnaround time of this study. The researcher have achieved hands on experience in the field of study and stronger with more confidant on the subject matter.

6.7 Future research

Since the research was only limited to 15 FSPs around Western Cape and Gauteng provinces, further research need to be conducted for large sample size and potential FSPs operating in other provinces. The outcomes of this study has indicated that FSPs are lacking the financial strength to invest in technology, future research could be conducted on addressing factors faced by FSP on adopting BD technology. Furthermore, future research can be conducted in addressing strategic plans for computing infrastructure, organisational procedures, policies and rules relevant or related to BD.

REFERENCES

Abazi, B. 2016. An approach to the impact of transformation from the traditional use of ICT to the Internet of Things: How smart solutions can transform SMEs. *IFAC Papersonline* 49(29):148-151. doi:10.1016/j.ifacol.2016.11.091.

Abdulla, M.S.S.H. 2015. Exploring the benefits of using big data when creating mobile applications. *Journal of Information Systems and Technology Management*. 4(1):2-26.

Abu-Alhaija, A.S. 2019. From epistemology to structural equation modeling: An essential guide in understanding the principles of research philosophy in selecting the appropriate methodology. *Australian Journal of Basic and Applied Sciences*. 13(9):122-128. doi:10.22587/ajbas.2019.13.9.12.

Abu-Salih, B, Wongthongtham, P., Zhu, D., Chan, K.Y. & Rudra, A. 2021. Social big data analytics: Practices, techniques, and applications. Springer. doi:10.1007/978-981-33-6652-7.

Aceto, G., Persico, V. & Pescapé, A. 2020. Industry 4.0 and health: Internet of Things, big data, and cloud computing for healthcare 4.0. *Journal of Industrial Information Integration*.18:100129. doi.org/10.1016/j.jii.2020.100129.

Adnan, K., Akbar, R., Khor, S.W. & Ali, A. 2020. Role and challenges of unstructured big data in healthcare. *Data Management, Analytics and Innovation*. pp. 301-323. doi.org/10.1007/978-981-32-9949-8_22.

Ahmadi, M., Dileepan, P.. & Wheatley, K.K. 2016. A SWOT analysis of big data. *Journal of Education for Business.* 91(5):289-294.

Ajzen, I. 1991. The theory of planned behavior: Organisational behavior and human decision processes. *Journal of High Technology Management Research*. 50(2):179-211. doi.org/10.1016/0749-5978(91)90020-T.

Akhtar, P., Frynas, J.G., Mellahi, K. & Ullah, S. 2019. Big data-savvy teams' skills. Big datadriven actions and business performance. *British Journal of Management.* 30:252-271. doi:10.1111/1467-8551.12333.

Akter, S., Wamba, S.F., Gunasekaran, A., Dubey, R. & Childe, S.J. 2016. How to improve firm performance using big data analytics capability and business strategy alignment? *International Journal of Production Economics.* 182:113-131. doi.org/10.1016/j.ijpe.2016.08.018.

Akter, S., Gunasekaran, A., Wamba, S.F., Babu, M.M. & Hani, U. 2020. Reshaping competitive advantages with analytics capabilities in service systems. *Technological Forecasting and Social Change*. 159:120180. doi:10.1016/j.techfore.2020.120180.

Al-Emadi, H., Zakir, J. & Seymour, T. 2015. Big data analytics. *Journal of Enterprise Information Management*. 16(2):81-90.

Al-sharafi, M.A., Arshah, R.A. & Abu-Shanab, E.A. 2017. Factors affecting the continuous use of cloud computing services from expert's perspective. IEEE region 10 conference (TENCON), Malaysia. 2017:986-991.

Alhajj, R., Moshirpour, M. & Far, B. 2020. *Data Management and Analysis*. Springer International Publishing.

Alharthi, A., Krotov, V. & Bowman, M. 2017. Addressing barriers to big data. Business Horizons. Elsevier. 60(3):285-292. doi:10.1016/j.bushor.2017.01.002.

Aljaroodi, H.M., Chiong, R. & Adam, M.T.P. 2020. Exploring the design of avatars for users from Arabian culture through a hybrid approach of deductive and inductive reasoning. *Computers in Human Behavior.* 106:106246. doi.org/10.1016/j.chb.2020.106246.

Allen, G.J. 2016. Big data analytics: What a small to medium enterprise should know. *Journal of Big Data*. 2(2):1-6.

Alshahri, M. 2016. Big data analytics for SMEs: Questions of opportunities, challenges, benefits, operations and challenges solutions. *Journal of Big Data*. 6(1):3-267.

Alshamaila, Y., Papagiannidis, S. & Li, F. 2017. Cloud computing adoption by SMEs in the north east of England. *Journal of Enterprise Information Management.* 26(3). doi:10.1108/17410391311325225.

Alsooj, A. 2015. Big data application in supermarkets: Opportunists for supermarkets. *Journal of Science & Technology Policy Management*. 6(1):3-267.

Alsheikh, M.A., Niyato, D., Lin, S., Tan, H.P & Han, Z. 2016. Mobile big data analytics using deep learning and apache spark. IEEE Network. 30(3):22-29. doi:10.1109/MNET.2016.7474340.

Alt, R., Leimeister, J.M., Priemuth, T., Sachse, S., Urbach, N. & Wunderlich, N. 2020. Software-defined business: Implications for IT management. *Business and Information Systems Engineering.* 62(6):609-621. doi.org/10.1007/s12599-020-00669-6.

Alwan, H.B. & Ku-Mahamud, K.R. 2020. Big data: definition, characteristics, life cycle, applications, and challenges. IOP Conferense Series: Materials Science and Engineering. doi:10.1088/1757-899X/769/1/012007.

Alzahrani, A. & Sadaoui, S. 2020. Clustering and labeling auction fraud data. Data Management, Analytics and Innovation. Springer. pp. 269-283. doi.org/10.1007/978-981-32-9949-8_20.

Andrade, R. & Torres, J. 2018. Enhancing intelligence SOC with big data tools, 2018 IEEE 9th Annual information technology, electronics and mobile communication conference (IEMCON). Vancouver. 2018:1076-1080.

Androniceanu, A. 2019. The social sustainability of smart cities: Urban technological innovation, big data management, and the cognitive Internet of Things. *Geopolitics, History, and International Relations.* 11(1):110-115. doi:10.22381/GHIR11120197.

Andronie, M. 2019. Integrating the principles of green marketing by using big data. Good practices. *Amfiteatru Economic Journal.* 21(50):258-269. doi.org10.24818/EA/2019/50/258.

Angelis, A. & Phillips, L.D. 2021. Advancing structured decision-making in drug regulation at the FDA and EMA. *British Journal of Clinical Pharmacology*. 87(2):395-405. doi:10.1111/bcp.14425.

Anurag, S. 2020. Finance for non-finance executives, in Business Expert Press (BEP). ProQuest Ebook Central. doi:6282180.

Arthur, K.N.A. & Owen, R. 2019. A Micro-ethnographic study of big data-based innovation in the financial services sector: Governance, ethics and organisational practices. *Journal of Business Ethics*. 160(2):363-375. doi.org/10.1007/s10551-019-04203-x.

Assuncao, M.D., Calheiros, R.N.,Bianchi, S., Netto, M.A.S. & Buyya, R. 2019. Big data computing and clouds: Trends and future directions. *Journal of Parallel and Distributed Computing*. 5(3):79-80. doi.org/10.1016/j.jpdc.2014.08.003.

Aswale, N. & Mukul, K. 2020. Role of data analytics in human resource management for prediction of attrition using job satisfaction, in Data management, analytics and innovation. Springer. pp. 57-67.

Awa, H.O., Ojiabo, O.U. & Emecheta, B.C. 2015. Integrating TAM, TPB and TOE frameworks and expanding their characteristic constructs for e-commerce adoption by SMEs. *Journal of Science & Technology Policy Management*. 6(1):76-94. doi.org/10.1108/JSTPM-04-2014-0012.

Baaziz, A. & Quoniam, L. 2013. How to use big data technologies to optimize operations in upstream petroleum industry. *International Journal of Innovation*. 1(1):19-25. doi:10.5585/iji.v1i1.4.

Babbie, E.R. 2013. The basics of social research: Issues in information systems. Cengage Learning. Boston.

Babbie, E.R. 2015. The practice of social research. Cengage Learning, Boston..

Baesens, B., Bapna, R., Marsdan, J.R. & Vanthienen, J. 2018. Transformational issues of big data and analytics in networked business, *MIS Quarterly*. 38(2):629-631. doi.org/10.25300/MISQ/2016/40:4.03.

Bagherzadeh, M., Markovic, S., Cheng, J. & Vanhaverbeke, W. 2020. How does outside-in open innovation influence innovation performance? Analyzing the mediating roles of knowledge sharing and innovation strategy. IEEE Transactions on Engineering Management. 67(3):740-753. doi:10.1109/TEM.2018.2889538.

Bakhouya, M., Zbakh, M., Essaaidi, M. & Manneback, P. 2020. Cloud computing, IoT, and big data: Technologies and applications, *Concurrency and computation: Practice and experience*. 32(17):10-13. doi.org/10.1002/cpe.5896.

Bizconnect. 2014. Agribusiness price updates. Standard bank biz connect.

Basahel, A., Yamin, M. & Drijan, A. 2016. Barriers to cloud computing adoption for SMEs in Saudi Arabia. *International Journal of Information Tehnology*. 8(2):1044-1049.

Baskaran, R. 2019. Impact of big data on enterprise content management. *Journal of Science & Technology Policy Management*. 2(1):15.

Beri, P. & Ojha, S. 2016. Comparative analysis of big data management for social networking sites. 3rd International conference on computing for sustainable global development (INDIACom). *Computer Science.* 2(16):1196-1200.

Bernard, H.R. & Harvey, R. 2018. Social research methods: Qualitative and quantitative approaches. Sage.

Bhadani, A.K. & Jothimani, D. 2016. Big data: Challenges, opportunities and realities.. Effective Big Data Management and Opportunities for Implementation. doi:10.4018/978-1-5225-0182-4.ch001.

Blasiak, K. 2014. Big Data; A Management Revolution: The emerging role of big data in businesses.pp.1-44

[https://www.theseus.fi/bitstream/handle/10024/74701/Big%20Data%20%20A%20Mana gement%20Revolution_Kevin%20Blasiak.pdf?sequence=1]

Boubiche, S.,Boubiche, D.E., Bilami, D.E. & Toral-Cruz, H. 2018. Big data challenges and data aggregation strategies in wireless sensor networks, IEEE Access.*Special Section on Real-Time Edge Analytics for ~Big Data in Internet of Things.* 6:20558-20571. doi:10.1109/ACCESS.2018.2821445.

Bowden, B., Muldoon, J., Gould, A. & McMurray, A. 2020. *The Palgrave handbook of management history*. Springer. doi:10.1007/978-3-319-62348-1.

Buhalis, D. & Volchek, K. 2021. Bridging marketing theory and big data analytics: The taxonomy of marketing attribution. *International Journal of Information Management*. 56:102253. doi.org10.1016/j.ijinfomgt.2020.102253.

Burrell, G. & Morgan, G. 1979. Part I: In search of a framework, Sociological paradigms and organisational analysis: Elements of sociology of corporate life. London: Heinemann.

Burrell, G. & Morgan, G. 2017. Sociological paradigms and organisational analysis: *Elements of the sociology of corporate life*. Routledge.

Calder, B.J., Malthouse, E.C. & Maslowska, E. 2016. Brand marketing, big data and social innovation as future research directions for engagement. *Journal of Marketing Managemet.* 32(5-6):579-585. doi.org/10.1080/0267257X.2016.1144326.

Cant, M. 2020. Legal factors affecting the establishment of township SMMEs in South Africa: Are they killing small businesses? *The Journal of Private Enterprise*. 35(4):61-70.

Carbonell, I.M. 2016. The ethics of big data in big agriculture. *Journal of Big Data*. 5(1):1-13. doi:10.14763/2016.1.405.

Castelli, I.E., Arismendi-Arrieta, D.J., Bhowmik, A., Cekic-Laskovic, I., Clark, S., Dominko, R., Flores, E., Flowers, J., Frederiksen, K.U., Friis, J., Grimaud, A., Hansen, K.V., Hardwick, L.J., Hermansson, K., Königer, L., Lauritzen, H., Le Cras, F., Li, H., Lyonnard, S., Lorrman, H., Marzari, N. Niedzicki, L., Pizzi, G., Rahmanian, F., Stein, H., Uhrin, M., Wenzel, W., Winter, M., Wölke, C. & Vegge, T. 2021. Data management plans: The importance of data management in the BIG-MAP project. Batteries & Supercaps. *European Chemical Societies Publishing*. 4(12):1-11. doi.org/10.1002/batt.202100117.

Cazeaux, C. 2017. Art, research, philosophy. Routledge.

Cepeda-Carrion, G.A., & Ruiz, D.M. 2005... A review of case studies publishing in management decision 2003 - 2004: Guides and criteria for achieving quality in qualitative research. *Management Decision*. 43.(6):851-876. doi.org/10.1108/00251740510603600.

Charles, V. & Gherman, T. 2013. Achieving competitive advantage through big data. Strategic implications. *Middle East Journal of Scientific Research.* 16(8):1069-1074. doi:10.5829/idosi.mejsr.2013.16.08.11811.

Chehbi-Gamoura, S., Derrouiche, R., Damand, D. & Barth, M. 2020. Insights from big data analytics in supply chain management: an all-inclusive literature review using the SCOR mode. *Production Planning & Control.* 31(5):355-382. doi.org/10.1080/09537287.2019.1639839.

Chen, H., Chiang, R.H.L. & Storey, V.C. 2012. Business intelligence and analytics: From big data to big impact. *MIS Quarterly*. 36(4):1165-1188. doi.org/10.1145/2463676.2463712.

Chen, R., Chen, H., Jin, C., Wei, B. & Yu, L. 2020. Linkages and spillovers between internet finance and traditional finance: Evidence from China. *Engineering Markets Finance and Trade 56(6:* 1196-1210. doi.org/10.1080/1540496X.2019.1658069.

Chong, W., Man, K. and Rho, S., 2015. Big data technology adoption in Chinese small and medium-sized enterprises. In *Proceedings of the International MultiConference of Engineers and Computer Scientists* (Vol. 2, pp. 18-20).

Chuah, M.H. & Thurusamry, R. 2021. Challenges of big data adoption in Malaysia SMEs based on Lessig's modalities: A systematic review.. *Cogent Business & Management.* 8(1). doi.org/10.1080/23311975.2021.1968191.

Ciampi, F., Demi, S., Magrini, A., Marzi, G. & Papa, A.. 2021. Exploring the impact of big data analytics capabilities on business model innovation: The mediating role of entrepreneurial orientation. *Journal of Business Research*. 123:1-13. doi:10.1016/j.jbusres.2020.09.023.

Clark, A. 2016. Big data in the music industry: Is the risk worth the reward? *Journal of Science & Technology Policy Management.* 6(1):3-267.

Coleman, S., Göb, R., Manco, G., Pievatolo, A., Tort-Martorell, X. & Reis, M.S.. 2019. How can SMEs benefit from big data? Challenges and a path forward. *Quality and Reliability Engineering International.* 32(6):2151-2164. doi.org10.1002/qre.2008.

Collica, R.S. 2019. Customer segmentation and clustering using SAS Enterprise miner. Sas Institute.

Collins, B. 2016. Big data and health economics strengths, weaknesses, opportunities and threats. *Pharmaco Economics*. 34(2):101-106. doi:10.1007/s40273-015-0306-7.

Cooper, D. & Schindler, P. 2003. Business research methods. *Journal of Business Research.* 16(6):4-12.

Cooper, N., Millela, F., Cant, I., White, M.D. & Meyer, G. 2021. Transfer of training - Virtual reality training with augmented multisensory cues improves user experience during training and task performance in the real world. PLOS ONE. *Chemical Tools & Probes.* 16:1-22. doi:10.1371/journal.pone.0248225.

Côrte-Real, N., Oliveira, T. & Ruivo, P. 2016. Assessing business value of big data analytics in European firms. *Journal of Business Research.* 70(25):379-390. doi.org10.1016/j.jbusres.2016.08.011.

Creswell, J.W. & Miller, G.A. 1997. Research methodologies and the doctoral process. *New Directions for Higher Education.* 99:33-46 doi.org/10.1002/he.9903.

Crotty, M. 2016. The foundations of social research: Meaning and perspective in the research process. Sage.

Das, H., Barik, R.K. & Dubey, H. 2018. Cloud computing for geospatial big data analytics: Intelligent edge, fog and mist computing:. *Studies in Big Data*. https://dl.acm.org/citation.cfm?id=3350446.

Datta, S.P.A., Saleem, T.J., Barati, M., Lopez, M.V.L., Furgala, M.L., Vengas, D.C., Santucci, G., Khargonekar, P.P., & McLamore, E.S. 2021. Data, analytics and interoperability between systems (IoT) is incongruous with the economics of technology: Evolution of Porous Pareto Partition (P3), *Big Data Analytics for Internet of Things*. 2:7-88. doi.org/10.1002/9781119740780.ch2.Dautov, R. & Distefano, S. 2017. Quantifying volume, velocity, and variety to support (Big) data-Intensive application development. *Computer Science*. 2843-2852. doi.org/10.1109/BigData.2017.8258252.

Davis, F.D. 1989. Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS Quarterly*. 13(3):319-340. doi.org/10.2307/249008.

Davoudian, A. & Liu, M. 2020. Big data systems : A software engineering perspective. *ACM Computing Surveys.* 53(5):1-39. doi.org/10.1145/3408314.

Del-Vecchio, P., Di Minin, A., Petruzzelli, A.M., Panniello, U. & Pirri, S. 2018. Big data for open innovation in SMEs and large corporations: Trends, opportunities, and challenges. *Creativity and Innovation Management*. 27(1):6-22. doi.org/10.1111/caim.12224.

Denzin, N. & Lincoln, Y.S. 2018. Handbook of qualitative research . Review by: Gary Alan Fine Source : Contemporary Sociology. 24(3):416-418 American Sociological Association. 24(3):416-418.

Doody, O. & Bailey, M.E. 2016. Setting a research question, aim and objective. *Nurse Researcher.* 23(4):19-23. doi: 10.7748/nr.23.4.19.s5.

Dubey, R., Gunasekaran, A., Childe, S.J., Blome, C. & Papadopoulus, T. 2019. Big data and predictive analytics and manufacturing performance: Integrating institutional theory. Resource-based view and big data culture. *British Journal of Management*. 30(2):341-361. doi:10.1111/1467-8551.12355.

Du Plooy-Cilliera, F., Davis, C. & Bezuidenhout, R. 2014. Research matters., Human Resource Management. Paarl Media South Africa.

Durhin, S. 2019. Big data analytics can help improve information security. *Journal of Big Data*. 11(4).

Edwards, A. & Talbot, R. 2014. *The hard-pressed researcher: A research handbook for the caring professions.* Routledge.

El Aboudi, N. & Benhlima, L. 2018. Big data management for healthcare systems: Architecture, requirements, and implementation. *Advances in Bioinformatics*. 2018:10. doi.org/10.1155/2018/4059018.

Elgendy, N. & Elragal, A. 2016. Advances in data mining. Applications and theoretical aspects. *Journal of Knowledge Management.* 61(71):505-517. doi:10.1007/978-3-319-41561-1.

Emrouznejad, A. & Charles, V. 2019. *Studies in big data. Big data for the greater good.* Volume 42, Springer.

Esterhuizen, D., Schutte, C. & Du Toit, A. 2017. A knowledge management framework to grow innovation capability maturity, *SA Journal of Information Management*. 14(1):1-10. doi:10.4102/sajim.v14i1.495.

Falahat, M., Ramayah, T., Soto-Acosta, P. & Lee, Y.Y. 2020. SMEs internationalization: The role of product innovation, market intelligence, pricing and marketing communication capabilities as drivers of SMEs' international performance. *Technological Forecasting and Social Change*. 152:119908. doi:10.1016/j.techfore.2020.119908.

Falkena, H, Abedian, I, Von Blottnitz, M., Coovadia, C., Davel, G., Madungandaba, J., Masilela, E. & Rees, S. 2021. SME's access to finance in South Africa. Financial Services and Regulations.

Fan, J., Han, F. & Liu, H. 2014. Challenges of big data analysis. *Journal of Big Data*. 1(2):293-314. doi.org10.1093/nsr/nwt032.Challenges.

Faroukhi, A.Z., Alaoui, I.E., Gahi, Y. & Amine, A. 2020. Big data monetization throughout big data value chain: A comprehensive review. *Journal of Big Data.* 7(1). doi:10.1186/s40537-019-0281-5.

Fernandez-Vidal, J.,Gonzalez, R., Gasco, J. & Llopis, J. 2022. Digitalization and corporate transformation: The case of European oil & gas firms. *Technological Forecasting & Social Change*. 174:121293. doi:10.1016/j.techfore.2021.121293.

Foley, A.M. 2018. Climate impact assessment and islandness. Challenges and opportunities of knowledge production and decision-making for small island devoloping states. 10(2):299-302. *International Journal of Climate Change Strategies and Management.* doi:10.1108/IJCCSM-06-2017-0142.

Fortunato, A., Gorgoglione, M., Petruzzelli, A.M. & Panniello, U. 2017. Leveraging big data for sustaining open innovation: The Case of Social TV. *Information Systems Management*. 34(3):238-249. doi.org/10.1080/10580530.2017.1330000.

Fox, O. 2015. Big data for retail SMEs limitations and challenges with recruiting data scientists. *International Journal of Computer Applications*. 6(2):56-99.

Fuster, G.G. & Scherrer, A. 2016. Big data and smart devices and their impact on privacy, Directorate-General for internal policies. Policy department C, Justice Freedom and Security. pp. 10-42. http://www.europarl.europa.eu/studies.

Franková, P. ., Drahošová, M. & Balco, P. 2016. Agile project management approach and its use in big data management. *Procedia Computer Science*. 83:576-583. doi:10.1016/j.procs.2016.04.272.

Frisk, J.E. & Bannister, F. 2017. Improving the use of analytics and big data by changing the decision-making culture: A design approach. *Management Decision*. 55(10):2074-2088. doi:10.1108/MD-07-2016-0460.

Gai, K., Qiu, M. & Zhao, H. 2017. Privacy-preserving data encryption strategy for big data in mobile cloud computing. IEEE Transactions on Big Data. *Computer Science*. 7(4):678-688. doi:10.1109/TBDATA.2017.2705807.

Gani, A., Siddiqa, A., Band, S.S. & Nasaruddin, F.H. 2015. A survey on indexing techniques for big data: Taxonomy and performance evaluation.. *International Journal of Emerging Research in Management & Technology*. 5(2):68. doi:10.1007/s10115-015-0830-y.

Gao, X., Li, J., Miao, D. & Liu, X. 2020. Recognising the tractability in big data computing. *Theoretical Computer Science*. 838:195-207. doi.org/10.1016/j.tcs.2020.07.026.

Gentsch, P. 2019. *Ai in Marketing, Sales and service*. How marketers without a data science degree can use Ai, Big data and bots. doi:10.1007/978-3-319-89957-2_3.

Ghasemaghaei, M. & Calic, G. 2020. Assessing the impact of big data on firm innovation performance: Big data is not always better data. *Journal of Business Research*. 108:147-162. doi:10.1016/j.jbusres.2019.09.062.

Ghavami, P. 2020. *Big Data Management: Data governance principles for big data analytics.* De Gruyter.

Giacalone, M. & Scippacercola, S. 2016. Quantitative methods inquires. Big data: Issues and an overview in some strategic sectors. *Journal of Applied Quantitive Methods*. 11(3):1-18.

Green, S., McKinney Jr, E., Heppard, K. & Garcia, L., 2018. Big data, digital demand and decision-making. *International Journal of Accounting & Information Management*.

Guillet, B.D. & Kucukusta, D. 2016. Spa market segmentation according to customer preference. *International Journal of Contemporary Hospitality Management.* 28(2):418-434. doi:10.1108/IJCHM-07-2014-0374.

Gupta, N.K. & Rohil, M.K. 2020. Big data security challenges and preventive solutions, in Data management, Analytics and innovation. *Advances in Intelligent Systems and Computing*. Springer Singapore.

Gupta, S., Modgil, S. & Gunasekaran, A. 2020. Big data in lean six sigma: A review and further research directions. *International Journal of Production Research*. 58(3):947-969. doi:10.1080/00207543.2019.1598599.

Han, H. & Trimi, S. 2022. Towards a data science platform for improving SME collaboration through Industry 4.0 technologies. *Technological Forecasting and Social Change*. 174:121242. doi:10.1016/j.techfore.2021.121242.

Hao, S., Zhang, H. & Song, M. 2019. Bid data, Big data analytics capability, and sustainable innovation performance. *Sustainability* (Switzerland). 11(24):1-15. doi:10.3390/su11247145.

Hartley, J. 2019. *Essential guide to qualitative methods in organisational research*. London, Sage. pp. 323.

Hashim, J. 2007. Information Communication Technology (ICT) Adoption among SME owners in Malaysia. *The International Journal of Business Information.* 2(2):221-240.

Hathcoat, J.D., Meixner, C. & Nicholas, M.C. 2019. Ontology and epistemology. *Handbook of Research Methods in Health Social Sciences*. Springer, Singapore.

He, W., Wang, F.K. & Akula, V. 2017. Managing extracted knowledge from big social media data for business decision making. *Journal of Knowledge Management.* 21(2):275-294. doi:10.1108/JKM-07-2015-0296.

Ho, J.K.K. 2016. A literature review on employability with diagramming techniques. *American Research Thoughts*. 2(6):3771-3784.

Hoddy, E.T. 2019. Critical realism in empirical research: employing techniques from grounded theory methodology. *International Journal of Social Research Methodology*. 22(1):111-124. doi.org/10.1080/13645579.2018.1503400.

Hofacker, C.F., Malthouse, E.C. & Sultan, F. 2016. Big data and consumer behavior : imminent opportunities. *Journal of Consumer Marketing.* 33(2:89-97. doi:10.1108/JCM-04-2015-1399.

Holden, M.T. & Lynch, P. 2019. Choosing the appropriate methodology: Understanding research philosophy. *The Marketing Review*. 4(4):397-409. http://repository.wit.ie/id/eprint/1466.

Horita, F.E.A., De Albuquerque, J.P., Marchenzini, V. & Mendiondo, E.M. 2020. Bridging the gap between decision-making and emerging big data sources : An application of a model-based framework to disaster management in Brazil. *Decision Support Systems*. 97:2-22. doi:10.1016/j.dss.2017.03.001.

Hosseini, S., Fallon, G., Weerakkody, V. & Sivarajah, U. 2019. Cloud computing utilisation and mitigation of informational and marketing barriers of the SMEs from the emerging markets: Evidence from Iran and Turkey. *International Journal of Information Management.* 46:54-69. doi:10.1016/j.ijinfomgt.2018.11.011.

Huang, Y., Gao, P., Zhang, Y. & Zhang, J. 2018, September. A cloud computing solution for big imagery data analytics. In *2018 International Workshop on Big Geospatial Data and Data Science (BGDDS)* pp. 1-4. IEEE.

Ikram, A., Su, Q., Fiaz, M. & Khadim, S.W. 2017. Big data in enterprise management: Transformation of traditional recruitment strategy. IEEE 2ND International Conference on Big Data Analysis. 2017:414-419. doi.org/10.1109/ICBDA.2017.8078853. Ikram, A., Su, Q., Fiaz, M. & Khadim, S., 2017, March. Big data in enterprise management: Transformation of traditional recruitment strategy. In *2017 IEEE 2nd International Conference on Big Data Analysis (ICBDA)* pp. 414-419. IEEE.

Intezari, A. & Gressel, S. 2019. Information and reformation in KM systems: big data and strategic decision-making. *Journal of Knowledge Management*. 21(1):71-91. doi:10.1108/JKM-07-2015-0293.

Ishak, N.M. & Abu-Bakar, A.Y.A. 2014. Developing sampling frame for case study: Challenges and conditions. *World Journal of Education.* 4(3):29-35. doi.org/10.5430/wje.v4n3p29.

Ishtiaq, M. 2019. Book Review Creswell, J.W. 2014. *Research design: Qualitative, quantitative and mixed methods approaches. English Language Teaching.* Thousand Oaks. Sage. 12(5):40. doi.org/10.5539/elt.v12n5p40.

Jabbar, A., Akhtar, P.& Dani, S. 2020. Real-time big data processing for instantaneous marketing decisions: A problematisation approach. *Industrial Marketing Management*. 90:558-569. doi.org/10.1016/j.indmarman.2019.09.001.

Jahanshahi, A.A. & Bhattacharjee, A. 2020. Competitiveness improvement in public sector organisations: What they need? *Journal of Public Affairs* 20(2). doi:10.1002/pa.2011.

Jamshed, S. 2017. Qualitative research method-interviewing and observation. *Journal of Basic and Clinical Pharmacy*. 5(4):87-100. doi:10.4103/0976-0105.141942.

Järvensivu, T. & Törnroos, J. 2010. Case study research with moderate constructionism: conceptualization and practical illustration. *Industrial Marketing Management. 39(1):100-108.* doi.org/10.1016/j.indmarman.2008.05.005.

Jha, A.K., Agi, M.A.N. & Ngai, E.W.T. 2020. A note on big data analytics capability development in supply chain. Decision support systems. *Humanities and Social Sciences/ Business Administration*.138:113382. doi:10.1016/j.dss.2020.113382.

Joubert, C.S., Schoeman, N.J. & Blignaut, N. 1999. Small, medium and micro-sized enterprises (SMMEs) and the housing construction industry : A possible solution to South Africa's socio-economic problems. *Journal of Economic and Management Sciences*. 2(1):21-33. doi.org/10.4102/sajems.v2i1.2561.

Kache, F. & Seuring, S. 2017. Challenges and opportunities of digital information at the intersection of big data analytics and supply chain management. *International Journal of Operations & Production Management.* 37(1):10-36. doi.org/10.1108/IJOPM-02-2015-0078.

Kaisler, S.H., Armour, F., Espinosa, J.A. & Money, W.H. 2020. *Obtaining value from big data for service delivery. Services Systems and Innovations in Business ands Society Collection.* Business Expert Press. 2020: 202.

Kalra, D. & Ranjan, M. 2021. Enduring data analytics for reliable data management in handling smart city services. *Soft Computing.* 25(18):12213-12225. doi:10.1007/s00500-021-05892-1.

Kamble, S.S., Gunasekaran, A., Ghadge, A. & Raut, R. 2020. A performance measurement system for industry 4.0 enabled smart manufacturing system in SMMEs- A review and empirical investigation, *International Journal of Production Economics*. 229(C):107853. doi:10.1016/j.ijpe.2020.107853.

Karahanna, E., Straub, D.W. & Chervany, N.L. 1999. Information technology adoption across time: A cross-sectional comparison of pre-adoption and post-adoption beliefs, *MIS Quarterly.* 23(2):183-213. doi.org/10.2307/249751.

Katal, A., Wazid, M. & Goudar, R.H. 2013. Big data: Issues, challenges, tools and good practices, in contemporary computing (IC3). 2013 *Sixth international conference on Contemporary Computing (IC3)*. 2013:404-409. doi: 10.1109/IC3.2013.6612229.

Kaufmann, M. 2019. Big data management canvas: A reference model for value creation from data. *Big Data and Cognitive Computing.* doi.org10.3390/bdcc3010019.

Kelemen, M.L. & Rumens, N. 2008. An introduction to critical management research. Sage.

Khan, N., Yaqoob, I. & Hashem, I.A.T., Inayat, Z., Ali, W.K.M., Alam, M., Shiraz, M. & Gani, A. 2014. Big data: Survey, technologies, opportunities, and challenges. *Scientific World Journal*. 6(2):1-18. doi.org/10.1155/2014/712826.

Khoase, R.G., Derera, E., McArthur, B. & Ndayizigamiye, P. 2020. Barriers to start-up and sustainable growth of SMMEs: A comparative study between South Africa and Lesotho.. *African Journal of Business and Economic Research.* 15(2):135-155. doi.org/10.31920/1750-4562%2F2020%2Fv15n2a7.

Khoase, R.G., Mutinta, G. and McArthur, B. 2018. A Comparative Approach to Assess the Impact of Supporting Institutions' Training on SMMEs. *Acta Universitatis Danubius. Œconomica*, *14*(5):783-85.

Klein, H.K. & Myers, M.D. 1999. A set of principles for conducting and evaluating interpretive field studies in information systems. *MIS quarterly.* 23(1):67-94. doi.org/10.2307/249410.

Kossaï, M. & Piget, P. 2014. Adoption of information and communication technology and firm profitability: Empirical evidence from Tunisian SMEs. *Journal of High Technology Management Research*. 25(1):9-20. doi.org/10.1016/j.hitech.2013.12.003.

Krell, M.M., Bernd, J., Li, Y., Ma, D. Choi, J., Ellsworth, M., Borth, D. & Friedland, G. 2017. Field studies with multimedia big data: Opportunities and challenges (Extended Version). *Comouter Science*. http://arxiv.org/abs/1712.09915.

Kumar, A., Palvadi, S.K., Sairam, T.V.M. & Rathore, P.S. 2019. *Impact of big data in social networking through various domains, Handbook of IoT and Big Data.* CRC Press.

Kumar, D., Samalia, H.V. & Verma, P. 2017. Exploring suitability of cloud computing for small and medium- sized enterprises in India. *Journal of Small Business and Enterprise Development.* 24(4):814-832. doi.org10.1108/JSBED-01-2017-0002.

Kunz, W., Aksoy, L., Bart, Y., Heinonen, K., Kabadayi, S., Ordenes, F.V., Sigala, M., Diaz, D. & Theodoulidis, B. 2017. Customer engagement in a big data world. *Journal of Services Marketing*. 31(2):161-171. doi.org/10.1108/JSM-10-2016-0352.

Ladzani, M.W. & Netswera, F.G. 2016. Support for rural small businesses in Limpopo province. South Africa. *Development Southern Africa*. 26(1):226-238. doi:10.1080/03768350902899512.

Lamb, C.W., Hair, J.F., McDaniel, C., Boshoff, C., Terblanche, N., Elliott, R. & Klopper, R.E. 2020. *Marketing.* 6 edition. South Africa: Oxford University Press.

Lamba, K. & Singh, S.P. 2018. Modeling big data enablers for operations and supply chain management. *The International Journal of Logistics Management*. 29(2):629-658. doi:10.1108/IJLM-07-2017-0183.

LaOlivo, J.F.L., Guzmán, J.G., Colomo-Palacios, R. & Stantchev, V. 2016. IT innovation strategy: managing the implementation communication and its generated knowledge through the use of an ICT tool. *Journal of Knowledge Management*. 20(3):512-533. doi.org10.1108/JKM-06-2015-0217.

La Torre, M., Dumay, J.C. & Rea, M. 2018. Breaching intellectual capital : Critical reflections on big data security. *Meditari Accountancy Research*, 26(3):463-482.doi:10.1108/MEDAR-06-2017-0154.

Lesejane, L.A., 2021. An assessment of tax compliance costs among Small Medium and *Micro Enterprise in South Africa* (Doctoral dissertation, North-West University (South Africa)). <u>http://hdl.handle.net/10394/37839</u>.

Li, C., Chen, Y. & Shang, Y. 2022. A review of industrial big data for decision making in intelligent manufacturing, *Engineering Science and Technology, an International Journal.* 29:101021. doi.org/10.1016/j.jestch.2021.06.001.

Li, J., Tao, F., Cheng, Y. & Zhao, L. 2015. Big data in product lifecycle management. *The International Journal of Advanced Manufacturing Technology*.81:667-684. doi:10.1007/s00170-015-7151-x.

Li, J., Lu, M., Dou, G. & Wang, S. 2017. Big data application framework and its feasibility analysis in library. *Information Discovery & Delivery*. *45(4):161-168*. doi:10.1108/IDD-03-2017-0024.

Li, K. 2022. SWOT analysis of e-commerce development of rural tourism farmers' professional cooperatives in the era of big data. *IET communications*. 16(5):592-603. doi:10.1049/cmu2.12358.

Lies, H.J. 2019. Marketing intelligence and big data: Digital marketing techniques on their way to becoming social engineering techniques in marketing. *Intyernational Journal of Interactive Multimedia and Artificial Intelligence*. 5:134-144. doi:10.9781/ijimai.2019.05.002.

Line, N.D., Dogru, T., El-Manstrly, D., Buoye, A., Malthouse, E. & Kandampully, J. 2020. Control, use and ownership of big data : A reciprocal view of customer big data value in the hospitality and tourism industry. *Tourism Management.* 80:104106. doi:10.1016/j.tourman.2020.104106.

Lisa, A. 2021. *Big data marketing: Engage your customers more effectively and drive value.* Wiley. 2019:41-50.

Liu, X., Shin, H. & Burns, A.C. 2021. Examining the impact of luxury brand's social media marketing on customer engagement : Using big data analytics and natural language processing. *Journal of Business Research*. 125::815-826. doi:10.1016/j.jbusres.2019.04.042.

Liu, Y., Soroka, A., Han, L., Jian, J. & Tang, M. 2020. Cloud-based big data analytics for customer insight-driven design innovation in SMEs. *International Journal of Information Management*. 51:102034. doi:10.1016/j.ijinfomgt.2019.11.002.

Lucena, F. D'O. & Popadiuk, S. 2020. Tacit knowledge in unstructured decision process, RAUSP *Management Journal*. 55(1):22-39. doi:10.1108/RAUSP-05-2018-0021.

Lugmayr, A., Stockleben, B., Scheib, C. & Mailaparampil, M.A. 2017. Cognitive big data: survey and review on big data research and its implications. What is really "new" in big data? *Journal of Knowledge Management*. 21(1):197-212. doi.org/10.1108/JKM-07-2016-0307.

Maduku, H.& Kaseeram, I. 2021. Success indicators among black owned informal small micro and medium enterprises' (SMMEs) in South Africa. *Development Southern Africa*. 38(4):664-682. doi:10.1080/0376835X.2021.1913997.

Mai, M., Leung, C., Choi, J.M.C. & KwaN, L.K.R. 2020. Big data analytics of Twitter data and its application for physician assistants: who is talking about your profession in Twitter? in Data Management and Analysis. *Computer Science*. 202:17-32. doi.org/10.1007/978-3-030-32587-9_2.

Malik, S., Chadhar, M. & Chetty, M. 2021. Factors affecting the organisational adoption of blockchain technology: An Australian perspective, *Proceedings of the 54th Annual Hawaii International Conference on System Sciences*, 2020: 5597-5606. doi:10.24251/hicss.2021.680.

Maloka, C.M. 2013. The contribution of small, medium and micro enterprises towards local economic development in Mankweng Township. Limpopo Province. *Journal of Management*. 4(2):1-126. <u>http://hdl.handle.net/10386/970</u>.

Mamman, A., Kanu, A.M., Alharbi, A. & Baydoun, N. 2015. Small and medium-sized enterprises (SMEs) and poverty reduction in Africa. *Strategic Management Perspective*. London: Cambridge Scholars Publishing.

Mangla, S.K. 2020. *Develop academic writing skills for a high impact research project. Scientific Methods Used in Research and Writing.* CRC Press. pp. 1-8.

Mantje, M. & Rambe, P. 2021. The effects of knowledge management on smme competitiveness: An emerging economy context view. Proceedings of the European Conference on Knowledge Management (ECKM). *Academic Conferences International Limited.* 2021:534-544. doi:10.34190/EKM.21.063.

Marja, S.B. 2016. Review of big data in Finland. Journal of Big Data. 9(3):41-89.

Marnewick, C. & Bvuma, S. 2020. An information and communication technology adoption framework for small, medium and micro-enterprises operating in townships South Africa. *Communication Technology*. 12:1-12. doi.org/10.4102/sajesbm.v12i1.318.

Marshall, C. & Rossman, G. 2016. *Designing qualitative research.* Sixth Edition. 95(4):652-654. Sage, Thousand Oaks.

Matekenya, W. & Moyo, C. 2022. Innovation as a driver of SMME performance in South Africa: a quantile regression approach. *African Journal of Economic and Management Studies*.13(3). doi:10.1108/AJEMS-06-2021-0306.

Mati, M. 2016. Rationalisation of business processes to create a unified information systems portfolio in a merger: A case study of a financial institution. *Journal of Information Systems and Technology Management*. 6(1):1-107.

Matt, D.T., Modrák, V. & Zsifkovits, H. 2020. *Industry 4.0 for SMEs: Challenges, opportunities and requirements.* Springer doi:10.1007/978-3-030-25425-4.

Mattera, M. 2018. SMEs transformation through usage and understanding of big data case study: Spanish restaurant industry, 2018. *IEEE 3rd International Conference on Big Data Analysis (ICBDA).* 2018:186-189. doi:10.1109/ICBDA.2018.8367674.

Mbinda, B. & Spencer, J.P. 2016. Risks connected to the work force at the small, medium and micro enterprises. *Risk Governance and Control Financial Markets & Institutions.* 6(4):161-166. doi.org/10.22495/rcgv6i4c1art7.

Mercier, M., Glesser, D., Georgiou, Y. & Richard, O. 2017. *Big data and HPC collocation: Using HPC idle resources for big data analytics.* IEEE Big data 2017. Boston. pp. 347-352. https://hal.archives-ouvertes.fr/hal-01633507.

Mirhosseini, S.A. 2018. Mixed methods research in TESOL: Procedures combined or epistemology confused? *TESOL quarterly*. 52(2):468-478. doi:10.1002/tesq.427.

Moon, H., Kim, K., Kim, S. & Park, J. 2017. Ecosystem design of big data through previous study analysis in the world: Policy design for big data as public goods. *Computer Science*. doi:10.1109/BigDataCongress.2017.79.

Moyo, I. & Gumbo, T. 2021. *Urban informality policies and regulatory frameworks in South African and Zimbabwean cities, in Urban informality in South Africa and Zimbabwe*. Springer. page123-133. doi.org/10.1007/978-3-030-65485-6_9.

Mukunthan, M.A. & Selvakumar, S. 2019. Multilevel Petri net-based ticket assignment and IT management for improved IT organisation support. *Concurrency and Computation: Practice and Experience*. 31(14):1-7. doi:10.1002/cpe.5297.

Munusamy, S. & Murugesan, P. 2020. Modified dynamic fuzzy c-means clustering algorithm – Application in dynamic customer segmentation. *Applied* Intelligence. 50(6):1922-1942. doi.org/10.1007/s10489-019-01626-x.

Myers, M.D. 2013. Qualitative research in business and management. *Journal of Business Research*. 18(6):34.

Myers, M.D. 2018. The philosopher's corner: The value of philosophical debate: Paul Feyerabend and his relevance for IS research. ACM SIGMIS database: *Database for Advances in Information Systems*. 49(4):11-14. doi.org/10.1145/3290768.3290771.

Na, S., Heo, S., Han, S. Shin, Y. & Roh, Y. 2022. Acceptance model of artificial intelligence (AI)-based technologies in construction firms: Applying the technology acceptance model (TAM) in combination with the technology–organisation–environment (TOE) framework. *Buildings*. 12(2). doi:10.3390/buildings12020090.

Nagar, P., Atriwal, L., Mehra, H. & Tayal, S. 2016. Comparison of generalized and big data business intelligence tools. 2016 3rd International Conference on Computing for Sustainable Global Development (INDIAComn). 2016:3585-3588.

Nair, S.S.K. & Ganesh, N. 2016. An exploratory study on big data processing : A case study from a biomedical informatics. *3rd MEC International Conference on Big Data and Smart* City (ICBDSC). doi:10.1109/ICBDSC.2016.7460338.

Nasrollahi, M., Ramezani, J. & Sadraei, M. 2021. The impact of big data adoption on SMES' performance. *Big Data and CognitiveComputing*. 5(4):68. doi:10.3390/bdcc5040068.

Nda, R.M. & Tasmin, R.B. 2019. Big data management in education sector: An overview. 5(6):5009-5014. *Path of Science*. doi:10.22178/pos.47-6.

Ndambo, D. 2016. Big data analytics and competitive advantage of commercial banks and insurance companies in Nairobi. Kenya. http://hdl.handle.net/11295/98835

Ndayizigamiye, P.& Khoase, R.G. 2021. *Analysing the relationship between SMME geographic coverage and e-commerce adoption. In Perspectives on ICT4D and Socio-Economic Growth Opportunities in Developing Countries.* IGI Global. pp. 212-223. doi:10.4018/978-1-7998-2983-6.ch008.

Ng, J.J., Navaretnam, S. & Wei, J.L.Q. 2020. Considerations for IT management in a COVID-19 World. *IEEE Engineering Management Review*. 48(3):16-18. doi:10.1109/EMR.2020.3014777.

Nguyen, S.P. 2021. Deep customer segmentation with applications to a Vietnamese supermarkets' data. *Soft Computing- A Fusion of Foundations, Methodologies and Applications.*. 25(12):7785-7793. doi:10.1007/s00500-021-05796-0.

Norman, R., Bolin, J. & Powell, E.T. 2018. Using big data analytics to create a predictive model for joint strike fighter. *2018 IEEE International Conference on Big Data (Big Data)*. pp. 3590-3596. doi:10.1109/BigData.2018.8622388.

Nunes, M.B. 2017. Understanding big data for industrial innovation and design: The missing Information Systems Perspective. *Journal of Data Information Sciences*. 2:1-6. doi:10.1515/jdis-2017-0017.

Ochieng, G. 2015. The adoption of big data analytics. *Journal of Big Data* 9(4):35-43.

O'Connor, C. & Kelly, S. 2017. Facilitating knowledge management through filtered big data: SME competitiveness in an agri-food sector. *Journal of Knowledge Management*. 21(1):156-179. doi:10.1108/JKM-08-2016-0357.Ogbuokiri, B., Udanor, C. & Agu, M. 2015. Implementing big data analytics for small and medium enterprise (SME) regional growth. *Journal of Computer Engineering*. 17(6):340-480. doi:10.9790/0661-17643543.

Ogwang, K.P. 2016. Strategic role of big data analytics on the competitive advantage of supermarket chains: A case study of Nakumatt Holdings Limited. *Journal of Knowledge Management Practice*. 4(1):400-675.

Oji, O.N.E., Iwu, C.G. & Haydam, N. 2017. The use of social media marketing strategies by SMMEs in the hospitality sector in Cape metropole. South Africa. *African Journal of Hospitality, Tourism and Leisure.* 6(1):1-16. http://: www.ajhtl.com.

O'Leary, Z. 2021. The essential guide to doing your research project. 4th edition. Sage.

Olszak, C.M. & Zurada, J. 2020. Big data in capturing business value. *Information Systems Management*. 37(3):240-254. doi:10.1080/10580530.2020.1696551.

Pandey, P.& Pandey, M.M. 2015. *Research methodology: Tools and techniques*. Romania: Bridge Center.

Paramita, A.S. & Wiradinata, T. 2016. Point of sales application based on cloud computing adoption for Indonesian small medium enterprise. *Qualitative Study.* (1):151-155. doi.org/10.1145/12345.67890.

Paré, G., Guillemette, M.G. & Raymond, L. 2020. IT centrality. IT management model, and contribution of the IT function to organisational performance: A study in Canadian hospitals. *Information and Management.* 57(3):103198. doi.org/10.1016/j.im.2019.103198.

Pauleen, D.J. & Wang, W.Y.C. 2017. Does big data mean big knowledge? KM perspectives on big data and analytics. *Journal of Knowledge Management*. 21(1):1-6. doi.org10.1108/JKM-08-2016-0339.

Peteraf, M.A. & Barney, J.B. 2003. Unraveling the resource-based tangle, *Journal of Global Information Management.* 24(4):309–323. doi.org/10.1002/mde.1126.

Porter, M.E. 1980. *Competitive strategy: Techniques for analyzing industries and competitors*. New York, Free Press.

Pozza, I.D., Brochado, A., Texier, L. & Najar, D. 2018. Multichannel segmentation in the after-sales stage in the insurance industry, *International Journal of Bank Marketing*. 36(6):1055-1072. doi.org10.1108/IJBM-11-2016-0174.

Purcell, T.F., Loftus, A. & March, H. 2020. Value–rent–finance. *Progress in Human Geography.* 44(3):437-456. doi.org10.1177/0309132519838064.

Rahimi, F., Møller, C. & Hvam, L. 2016. Business process management and IT management: The missing integration. *International Journal of Information Management*. 36(1):142-154. doi:10.1016/j.ijinfomgt.2015.10.004.

Rahman, M.M. 2022. Ethics and epistemology in qualitative social psychology. A Comparative Discussion. *Sociology and Politics – Islamic Perspective.*

Rahman, N. 2016. Factors affecting big data technology adoption. *Department of Engineering and Technology*. 2016:0-29.

Rahman, N. 2017. Big data analytics for a sustained competitive advantage. *Journal of Computer Engineering.* 18(8):4-22.

Rahman, N. and Aldhaban, F. 2015. August. Assessing the effectiveness of big data initiatives. In *2015 Portland International Conference on Management of Engineering and Technology (PICMET)* (pp. 478-484). IEEE.

Rajapakse, G.S. & Kiran, K. 2019. The library succession planner's decision-making style. *Library Management.* 38(8-9):497-510. doi:10.1108/LM-02-2017-0017.

Rehman, M.H.U., Chang, V., Batool, A. & Wah, T.Y. 2016. Big data reduction framework for value creation in sustainable enterprises. *International Journal of Information Management*. 36(6):917-928. doi.org/10.1016/j.ijinfomgt.2016.05.013.

Resnik, D.B. 2015. What is ethics in research & why is it important? *Journal of Science & Technology Policy Management*. 12(5):40-80.

Revathy, P.t. & Mukesh, R. 2017 Analysis of big data security practices. 2017. *3rd International Conference on Applied and Theoretical Computing and Communication Technology* (iCATccT). 2017:264-267.

Rialti, R. & Marzi, G. 2020. *Ambidextrous organisations in the big data era. The Role of Information Systems.* Palgrave Pivot, Switzerland. doi:10.1007/978-3-030-36584-4.

Robinson, L.M.. Marzloff, M.P., Van Putten, L., Pecl. G., Jennings, S., Nicol, S. Hobday, A.J., Tracey, S. Hartmann, K., Haward, M. & Frusher, S. 2019. Structured decision-making identifies effective strategies and potential barriers for ecosystem-based management of a range-extending species in a global marine hotspot. *Ecosystems*. 22(7):1573-1591. doi:10.1007/s10021-019-00358-w.

Rogers, E., Medina, U.E., Rivera, M. & Wiley, C. 2005. Complex adaptive systems and the diffusion of innovations. *Journal of Information Systems and Technology Management*. 10(3):1-26.

Rogers, E.M. 1995. Diffusion of innovations. 4th edition. New York: Free Press. pp. 14-25.

Rogers, E.M. 2003. Diffusion of Innovations. 5th edition. New York: Free Press. pp. 24-27.

Rothwell, W.J., Williams, S.L. & Zaballero, A.G. 2020. *Increasing learning & development's impact through accreditation*. Springer. doi:10.1007/978-3-030-14004-5.

Roy, C., Pandey, M. & Rautaray, S.S. 2018. A proposal for optimization of data node by horizontal scaling of name node using big data tools. *2018 3rd International Conference for Convergence in Technology* (I2CT). 2018:1-6.

Rungani, E.C. & Potgieter, M. 2018. The impact of financial support on the success of small, medium and micro enterprises in the Eastern Cape province. *Management information systems quarterly.* Forthcoming. *Acta Commercii.*18(1):1-12. doi.org/10.4102/ac.v18i1.591.

Ruxwana, N.L., Herselman, M.E. & Conradie, D.P. 2010. ICT applications as e-health solutions in rural healthcare in the Eastern Cape province of South Africa. *Health Information Management Journal*. 39(1):17-29. doi:10.1177/183335831003900104.

Sahu, S.K., Jacintha, M.M. & Singh, A.P. 2017. Comparative study of tools for big data analytics. An analytical study. *International conference on computing, communication and automation (ICCCA2017) Comparative*. 2017:37-41. doi.org/10.1109/CCAA.2017.8229827.

Saleem, H., Li, Y., Ali, Z., Mehreen, A. & Mansoor, M.S. 2020. An empirical investigation on how big data analytics influence China SMEs performance: Do product and process innovation matter? *Asia Pacific Business Review*. 26(5):537-562. doi:10.1080/13602381.2020.1759300.

Salleh, N.A. Hussin, H., Suhaimi, M. & Ali, A. 2018. A systematic literature review of cloud computing adoption and impacts among small, medium enterprises (SMEs). 2018 *International Conference on Information and Communication Technology for the Muslim World (ICT4M*), 2018:278-284. doi:10.1109/ICT4M.2018.00058.

Sanfilippo, E.M., Belkadi, F. & Bernard, A. 2019. Ontology-based knowledge representation for additive manufacturing. *Computers in Industry*. 109:182-194. doi.org/10.1016/j.compind.2019.03.006.

Sanson-Fisher, R.W. 2004. Diffusion of innovation theory for clinical change. *Journal of Science & Technology Policy Management*. 180(S6):S55-S56. doi.org/10.5694/j.1326-5377.2004.tb05947.x.

Santos, M.Y. 2017. A big data system supporting Bosch Braga Industry 4.0 strategy, *International Journal of Information Management.* 37(6):750-760. doi:10.1016/j.ijinfomgt.2017.07.012.

Santos, M.Y. & Costa, C. 2020. *Big data concepts, warehousing, and analytics, technology*. River

http://www.riverpublishers.com/journal_read_html_article.php?j=JMMC%2F1%2F3%2F2.

Saunders, M., Lewis, P. & Thornhill, A. 2009. *Research methods for business students.* Pearson. New York.

Saunders, M., Lewis, P.& Thornhill, A. 2012. *Research methods for business Students*, 6th edn, s'. SI. Pearson, Harlow.

Saunders, M., Lewis, P., Thornhill, A. & Bristow, A. 2019. Research methods for business students' Chapter 4: Understanding research philosophy and approaches to theory development. *Research Methods for Business Students*. Pearson.

Schosser, M. 2020. Big data to improve strategic network planning in airlines. Springer.

Scotland, J. 2012. Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms. *Journal of Business Research*. 5(9):9-16. doi:10.5539/elt.v5n9p9.

Seijger, C., Kleinschmit, D., Schmidt-Vogt, D., Mehmood-UI-Hassan, M. & Martius, C. 2021. *Water and sectoral policies in agriculture–forest frontiers: An expanded interdisciplinary research approach*. Ambio. Springer. 50(12):2311-2321. doi:10.1007/s13280-021-01555-5.

Sekhar, C.C. & Sekhar, C. 2017. Productivity improvement in agriculture sector using big data tools. *2017 International Conference On Big Data Analytics and computational Intelligence* (ICBDACI).2017:169-172. doi.org/10.1109/ICBDACI.2017.8070830.

Sen, D., Ozturk, M. & Vayvay, O. 2016. An overview of big data for growth in SMEs, *Procedia - Social and Behavioral Sciences*. 235:159-167. doi:10.1016/j.sbspro.2016.11.011.

Senarathna, I., Wilkin, C., Warren, M., Yeoh, W. & Salzman, S. 2018. Factors that influence adoption of cloud computing: An empirical study of Australian SMEs. *Australasian Journal of Information Systems* 22.. doi.org/10.3127/ajis.v22i0.1603.

Shah, S., Soriano, C.B. & Coutroubis, A.D. 2017. Is big data for everyone? The Challenges of Big Data Adoption in SMEs.2017 IEEE International Conference on Industrial Engineering and Engineering Management. 2017:803-807. doi.org/10.1109/IEEM.2017.8290002.

Shalaginov, A., Johnsen, J.W. & Franke, K. 2017. Cyber crime investigations in the era of big data. *Computer Science. 2017 IEEE International Conference on Big Data.* 2017:3672-3676. doi.org/10.1109/BigData.2017.8258362.

Shamim, S., Zeng, J., Shariq, S.M. & Khan, Z. 2019. Role of big data management in enhancing big data decision-making capability and quality among Chinese firms: A dynamic capabilities view. *Information and Management.* 56(6):103135. doi:10.1016/j.im.2018.12.003.

Sharma, R., Mithas, S. & Kankanhalli, A. 2016. Transforming decision-making processes: A research agenda for understanding the impact of business analytics on organisations. *European Journal of Information Systems*. 23(4):433-441. doi:10.1057/ejis.2014.17.

Shinde, P.P. Oza, K.S. & Kamat, R.K. 2017. Big data predictive analysis: Using R analytical tool. *International conference on I-SMAC (IoT in Social, Mobile, Analytics and Cloud).* doi.org/10.1109/I-SMAC.2017.8058297.

Shrouf, F. & Miragliotta, G. 2015. Energy management based on Internet of Things: practices and framework for adoption in production management. *Journal of Cleaner Production*. 100(40):235-246. doi:10.1016/j.jclepro.2015.03.055.

Sicari, R. 2021. Training, training and more training. *International Journal of Cardiovascular Imaging*. 37(7):2169. doi:10.1007/s10554-021-02264-6.

Silva, M.M., Poleto, T., E Silva, L.C., De Gusmao, A.P.H. & Costa, A.P.C.S. 2016. A grey theory based approach to big data risk management using FMEA. *Mathematical Problems in Engineering.* 2016(9175418). doi.org/10.1155/2016/9175418.

Simões, A., Soares, A. & Barros, A.C. 2020. Factors influencing the intention of managers to adopt collaborative robots (cobots) in manufacturing organisations. *Journal of Engineering and Technology* Management doi:10.1016/j.jengtecman.2020.101574.

Solé-Beteta, X., Navarro, J, Vernet, D., Zaballos, A., Torres-Kompen, R., Fonseca, D. & Briones, A. 2021. Automatic tutoring system to support cross-disciplinary training in big data. *Journal of Supercomputing*. 77(2):1818-1852. doi:10.1007/s11227-020-03330-x.

Soroka, A., Liu, Y., Han, L. & Haleem, M.S. 2017. Big data driven customer insights for SMEs in redistributed manufacturing. *Procedia CIRP*. 63:692-697. doi.org/10.1016/j.procir.2017.03.319.

Srinuan, C. 2018. Willingness to pay for cloud computing service of SME in Thailand. 2017 *3rd International Conference on Information Management (ICIM)*. doi:10.1109/INFOMAN.2017.7950340.

Sriram, G.S. 2022. Security challenges of big data computing. *International Research Journal of Modernization in Engineering Technology and Science*, *4*(1), pp.1164-1171.

Steinhoff, L., Arli, D., Weaven, S. & Kozlenkova, I.V. 2019. Online relationship marketing. *Journal of the Academy of Marketing Science*. 47(3):369-393. doi:10.1007/s11747-018-0621-6.

Stergiou, C. & Psannis, K.E. 2017. Efficient and secure big data delivery in cloud computing. *Multimedia Tools and Applications*. Springer.. 76(21):22803-22822. doi:10.1007/s11042-017-4590-4.

Storey, V.C. & Song, I. 2017. Big data technologies and management : What conceptual modeling can do. *Data & Knowledge Engineering.* 108:50-67. doi:10.1016/j.datak.2017.01.001.

Strydom, J.W. 2017. Longevity of SMMEs in Soweto: Does marketing play a role? *African Journal of Science, Technology, Innovation and Development*. 9(6):685-695. doi.org/10.1080/20421338.2017.1355603.

Sun, W., Zhao, Y. & Sun, L. 2018. Big data analytics for venture capital application: Towards innovation performance improvement, *International Journal of Information Management*. (November), 50:0-1. doi:10.1016/j.ijinfomgt.2018.11.017.

Sutaj, A. 2019. Computing adoption and usage in China and Nigeria, while section V covers the definition and description of what SMEs is in China and Nigeria giving an estimates value of their contribution to national GDP in the two countries. *Information and Management*. pp. 299-307. doi:10.1109/SmartTechCon.2017.8358386.

Tabesh, P., Mousavidin, E. & Hasani, S. 2019. Implementing big data strategies: A managerial perspective, *Business Horizons*. 62(3):347-358. doi:10.1016/j.bushor.2019.02.001.

Tashakkori, A. & Creswell, J.W. 2007. Editorial: The new era of mixed methods. *Journal of Mixed Methods Research*. Sage Publications. 1:3-7. doi:10.1177/2345678906293042.

Taylor, S.J.E., Kiss, T., Anagnostou, A, Tertyanszky, G., Kacsuk, P., Costes, J. & Fantini, N. 2018. The cloud SME simulation platform and its applications : A generic multi-cloud platform for developing and executing commercial cloud-based simulations. *Future Generation Computer Systems*. 88:524-539. doi.org/10.1016/j.future.2018.06.006.

Terezinho, F. 2017. How cloud computing works for industrial processes. *Engineering and Technology.* pp. 1-3.

Thoeni, A.T., Marshall, G.W. & Campbell,S.M. 2016. A resource-advantage theory typology of strategic segmentation. *European Journal of Marketing*. *50(12):2192-2215.* doi:10.1108/EJM-08-2015-0585.

Thore-Olsson, A.C., Johannesson, U. & Schweizer, R. 2019. Decision-making and cost deviation in new product development projects. *International Journal of Managing Projects in Business*. 11(4):1066-1085. doi:10.1108/IJMPB-02-2018-0029.

Thrassou, A. 2020. The changing role of business in global society. *Business ethics quarterly.* doi:10.5840/beq200919328.

Tornatzky, L.G., Fleischer, M. & Chakrabarti, A.K. 1990. *Processes of technological innovation*. Lexington..

Trabucchi, D., Buganza, T., Dell'Era, C. & Pellizzoni, E. 2018. Exploring the inbound and outbound strategies enabled by user generated big data: Evidence from leading smartphone applications. *Creativity and Innovation Management*. 27(1):42-55. doi:10.1111/caim.12241.

Trabucchi, D. & Buganza, T. 2018. Data-driven innovation: switching the perspective on big data. *European Journal of Innovation Management*.doi:10.1108/EJIM-01-2018-0017.

Troisi, O., Arco, M.D., Loia, F. & Maione, G. 2018. Big data management : The case of Mulino Bianco's engagement platform for value co-creation. *International Journal of Engineering Business Management.* 10(4):1-8. doi:10.1177/1847979018767776.

Turan, A., Tunç, A.Ö. & Zehir, C. 2015. A theoretical model proposal: Personal innovativeness and user involvement as antecedents of unified theory of acceptance and use of technology. *Procedia - Social and Behavioral Sciences*. 210:43-51. doi:10.1016/j.sbspro.2015.11.327.

Vahid, J.S., Hannan, A.M., Donatella, B. & Dorra, Y. 2022. Towards agility in international high-tech SMEs: Exploring key drivers and main outcomes of dynamic capabilities. *Technological Forecasting & Social Change.* 174:121272. doi:10.1016/j.techfore.2021.121272

Vaismoradi, M., Turunen, H. & Bondas, T. 2013. Content analysis and thematic analysis : Implications for conducting a qualitative descriptive study. Research Technology Management. *Journal of Nursing & Health Sciences*.15(2):398-405. doi:10.1111/nhs.12048.

Vajjhala, N. & Ramollari, E. 2016. Big data using cloud computing - Opportunities for small and medium-sized enterprises. *European Journal of Economics and Business Studies*. 4(1):129-137. doi:10.26417/ejes.v4i1.p129-137.

Van Maanen, J., Sørensen, J.B. & Mitchell, T.R. 2007. The interplay between theory and method. *Academy of Management Review.* 32(4):1145-1154. doi.org/10.5465/amr.2007.26586080.

Venkatesh, V., Morris, M.G., Davis, G.B. & Davis, F.D. 2003. User acceptance of information technology: Toward a unified view, *Management Information Systems Research Centre, Quarterly.* 27(3):425-478. doi.org/10.2307/30036540.

Verma, S. & Bhattacharyya, S.S. 2017. Perceived strategic value-based adoption of big data analytics in emerging economy. *Journal of Enterprise Information Management*. 30(3):354-382. doi:10.1108/JEIM-10-2015-0099.

Vogl, S., Schmidt, E.M. & Zartler, U. 2019. Triangulating perspectives: Ontology and epistemology in the analysis of qualitative multiple perspective interviews. *International Journal of Social Research Methodology*. 22(6):611-624. doi.org/10.1080/13645579.2019.1630901.

Wahyuni, D. 2012. The research design maze: Understanding paradigms, cases, methods and methodologies. *Journal of Applied Management Accounting Research*. 10(1):69-80. https://ssrn.com/abstract=2103082.

Walaza, M., Loock, M. & Kritzinger, E. 2020. A framework to enhance ICT security through education, training & awareness (ETA). Programmes in South African small, medium and micro-sized enterprises (SMMEs). CSOC. pp.1-40. doi.org/10.1007/978-3-030-51974-2_5.

Walker, R.S. & Brown, I. 2019. Big data analytics adoption : A case study in a large South African telecommunications organisation. *South African Journal of Information Management.* 21(1):1-10. doi.org/10.4102/sajim.v21i1.1079.

Walsham, G. 2018. Research methodologies for information systems in the development context: A tutorial. *The Digital Challenge: Information Technology in the Development Context:*. New York: Routledge.

Wamba, S.F., Gunasekaran, A., Akter, S., Ren, S.J.F., Dubey, R. & Childe, S.J. 2017. Big data analytics and firm performance: Effects of dynamic capabilities. *Journal of Business Research*. 70(24):356-365. doi.org/10.1016/j.jbusres.2016.08.009.

Wang, N.T. 2020. Managers' noticing of new organisational IT and influences of IT attributes. *Information and Management*. 57(5):103232. doi.org/10.1016/j.im.2019.103232.

Wang, S. & Wang, H. 2020. Big data for small and medium-sized enterprises (SME): A knowledge management model. *Journal of Knowledge Management*. 24(4):881-897. doi.org/10.1108/jkm-02-2020-0081.

Welman, C., Kruger, F. & Mitchell, B. 2005. *Research methodology.* 3rd ed. Oxford University Press. Cape Town.

White, A., Moore, D.W., Fleer, M. & Anderson, A. 2017. A thematic and content analysis of instructional and rehearsal procedures of preschool social emotional learning programs. Research technology management. *Australasian Journal of Early Childhood*. 42(3):82-92. doi.org/10.23965/AJEC.42.3.10.

Windheim, M. 2020. *Decision-making in modular product family design*. 17:79-94. Springer, Berlin. doi:10.1007/978-3-662-60715-2_4.

Wordsworth, S., Doble, B., Payne, K., Buchanan, J., Marshall, D., McCabe, C., Phillips, K. & Regier, D.A. 2018. Using big data in the cost-effectiveness analysis of next-generation sequencing technologies: Challenges and potential solutions. *Value in Health.* 21(9):1048-1053. doi: 10.1016/j.jval.2018.06.016.

Wu, Y. & Guan, H. 2018. An implementation of unified framework for big data management system. *Computer Science*. doi.org/10.48550/arXiv.1810.09378. Xiao, W., Bao, W., Zhu, X., & Liu, L. (2017). Cost-Aware Big Data Processing Across Geo-Distributed Datacenters. *IEEE Transactions on Parallel and Distributed Systems, 28*, 3114-3127.

Xiong, G., Zhang, C., Tongkai, J., Gaun, B., Sun, A., Hu, Y. & Nyberg T.R. 2016. Cloud computing platform and big data service for incubator cluster. *2016 IEEE International Conference on Service Operations and Logistics, and Informatics* (SOLI). Page 60-65. doi.org/10.1109/SOLI.2016.7551662.

Yadav, S.K., Khare, A. & Kavita, C. 2020. ASK approach: A pre-migration approach for legacy application migration to cloud. *Data management, analytics and innovation*. Singapore, Springer, page 15-27.

Yamin, M. & Al Makrami, A.A. 2015. Cloud computing in SMEs : Case of Saudi Arabia. *International Journal of Information Technology*. 7(1):853-860.

Yang, J., Grunsky, E. & Cheng, Q. 2019. A novel hierarchical clustering analysis method based on Kullback – Leibler divergence and application on dalaimiao geochemical exploration data. *Computers and Geosciences*. 123(29):10-19. doi.org/10.1016/j.cageo.2018.11.003.

Yarbrough, A.K. & Smith, T.B. 2007. Technology acceptance among physicians: A new take on TAM., *Journal of Science & Technology Policy Management*. 64(6):650-672. doi.org/10.1177/1077558707305942.

Yilmaz, T.K., Durak, U., Oguztuzun, H. & Çagiltay. 2018. Usability evaluation of a web-based ontology browser: The case of TSONT, *Turkish Journal of Electrical Engineering and Computer Sciences*. 26(2):1115-1128. doi.org/10.3906/elk-1610-50.

Yin, R.K. 2003. Applied social research methods series. Case study research: Design and *Methods*. 3rd editon. 5(1). Sage.

Zehnder, P.& Riemer, D. 2017. Towards automatic infrastructure provisioning for highly dynamic streaming applications. *IEEE International Conference on Big Data.* (2017):4869-4871. doi:10.1109/Bigdata.2017.8258571.

Zhan, Y., Tyan, K.H., Ji, G., Ching, L. & Tseng, M.L. 2019. A big data framework for facilitating product innovation processes. *Business Process Management Journal*. 23(3):518-536. doi:10.1108/BPMJ-11-2015-0157.

Zhang, C., Patras, P. & Haddadi, H. 2019. Deep learning in mobile and wireless networking: A survey. *IEEE Communications surveys & tutorials*. 21(3):2224-2287.

Zhang, S., Zhou, C. & Liu, Y. 2020. Consumer purchasing intentions and marketing segmentation of remanufactured new-energy auto parts in China. *Mathematical Problems in Engineering*. (2020):5647383. doi.org/10.1155/2020/5647383.

Zhao-hong, Y., Hui-yu, W., Bin, Z., Zhi-he, H. & Wan-lin, W. 2018. A literature review on the key technologies of processing big data, *IEEE 3rd International conference on cloud computing and big data analysis (ICCCBDA).* (2018):202-208. doi:10.1109/ICCCBDA.2018.8386512.

Zhao, Z. & Yang, Y. 2017. Influence of big data on manufacturing industry and strategies of enterprises : A literature review. *MATEC Web Conference*. (2)100:19. doi.org/10.1051/matecconf/201710002019.

Zhou, K., Fu, C. & Yang, S. 2016. Big data driven smart energy management : From big data to big insights. *Renewable and Sustainable Energy Reviews.* 56:215-225. doi.org/10.1016/j.rser.2015.11.050.

Zhu, Q., Kong, X. & Hong, S., Li, J. & He, Z. 2015. Global ontology research progress: A bibliometric analysis, *Aslib Journal of Information Management*. 67(1):27-54. doi:10.1108/AJIM-05-2014-0061.

Zhu, Y., Tan, Y., Luo, X. & He, Z. 2018. Big data management for cloud-enabled geological information services. *Scientific Programming.* doi.org/10.1155/2018/1327214.

APPENDIX A: INTERVIEW GUIDE QUESTIONS

Section A: participant's details

Personal Details:

What is your Name?

Name of your Company?

Your role in your Company (Job Title)?

How long have been working in this organisation?

What does your company do?

What is your highest level of education?

Do you have any employees working in your organisation, if yes how many?

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

RSQ 1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

- IQ1.1.1: To what extent does your organisation have experience with big data analytics?
- IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?
- IQ1.1.3: How does your organisation categorise your existing customers?
- IQ1.1.4: How do you manage your competitiveness against your competitors?
- IQ1.1.5: How would big data add value in the organisation?

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

- IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?
- IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?
- IQ1.2.3 Do they have a desire to know about big data?
- RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?RSQ 2.1: How do SMMEs determine the needs of their customers?

- IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?
- IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?
- IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs?
- RSQ 2.2: How can SMMEs incorporate BD in their business strategies?
- IQ2.2.1: Does your organisation have a strategy on big data or data analytics?
- IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?
- IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?
- IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

APPENDIX B: CONSENT LETTER

<<On company letterhead>>

Swaartb compan research	Andrew Moore Coppo ooi, a student at t y as part of his Mt and the nature o	, in my consent , for the cape Peninsula Universi ech (BIS) research. The stud f the data to be collected.	capacity as	art at
This con expecte withdray Research In additi	sent in no way con d that the student w this permission n will be done on t on, the company's	mmits any individual staff m will get explicit consent fro at some future time. the time and dates as agree s name may or may not be u	nember to participate in m any participants. I reso d with the student. Jsed as indicated below.	the research, and it is erve the right to (Tick as appropriate.)
	Thesis	Conference paper	Journal article	Research poster
Yes	V			incode an en poster
No				
	Alach			

128

<<On company letterhead>>

....., in my capacity as

at State Sta

w

This consent in no way commits any individual staff member to participate in the research, and it is expected that the student will get explicit consent from any participants. I reserve the right to withdraw this permission at some future time.

Research will be done on the time and dates as agreed with the student.

In addition, the company's name may or may not be used as indicated below. (Tick as appropriate.)

	Thesis	Conference paper	Journal article	Research poster
Yes				
No	\checkmark	<u> </u>		

<<Signature

12/12/2019

<<insert date>>

<<<<Insert name>>>>

<<On company letterhead>>

I. DEON ENGELIE , in my capacity as HEAD OF BUSILAT MITCH OPTIMAL give consent in principle to allow Luyanda Lincoln

Swaartbooi, a student at the Cape Peninsula University of Technology, to collect data in this company as part of his Mtech (BIS) research. The student has explained to me the nature of his/her research and the nature of the data to be collected.

This consent in no way commits any individual staff member to participate in the research, and it is expected that the student will get explicit consent from any participants. I reserve the right to withdraw this permission at some future time.

Research will be done on the time and dates as agreed with the student.

In addition, the company's name may or may not be used as indicated below. (Tick as appropriate.)

		conterence paper	Journal article	Research poster
Yes	-			/
No				

<<signature >>

<<insert date>>

<<<<Insert name>>>>

<<On company letterhead>>

1 Top lo allies in my capacity as they Individuality to allow Luyanda Lincoln

Swaartbooi, a student at the Cape Peninsula University of Technology, to collect data in this company as part of his Mtech (BIS) research. The student has explained to me the nature of his/her research and the nature of the data to be collected.

This consent in no way commits any individual staff member to participate in the research, and it is expected that the student will get explicit consent from any participants. I reserve the right to withdraw this permission at some future time.

Research will be done on the time and dates as agreed with the student.

In addition, the company's name may or may not be used as indicated below. (Tick as appropriate.)

0	Thesis	Conference paper	Journal article	Research poster
Yes				
No				

anature >>

<insert date

<<<<Insert name>>>>

<< On company letterhead>>

HTALANA BRIGKER, in my capacity as CEO at Swaartbooi, a student at the Cape Peninsula University of Technology, to collect data in this company as part of his Mtech (BIS) research. The student has explained to me the nature of his/her research and the nature of the data to be collected.

This consent in no way commits any individual staff member to participate in the research, and it is expected that the student will get explicit consent from any participants. I reserve the right to withdraw this permission at some future time.

Research will be done on the time and dates as agreed with the student.

In addition, the company's name may or may not be used as indicated below. (Tick as appropriate.)

	Thesis	Conference paper	Journal article	Research poster
Yes	N	\sim	X	K
No		/	1-	~

<<Signature >>

H.F.O. DRKIKM

<<<<Insert name>>>>

12/12

<<insert date>>
<< On company letterhead>>

I DYLAN FUNT, in my capacity as DICETOR at LIFECHED give consent in principle to allow Luyanda Lincoln Swaartbooi, a student at the Cape Peninsula University of Technology, to collect data in this company as part of his Mtech (BIS) research. The student has explained to me the nature of his/her research and the nature of the data to be collected.

This consent in no way commits any individual staff member to participate in the research, and it is expected that the student will get explicit consent from any participants. I reserve the right to withdraw this permission at some future time.

Research will be done on the time and dates as agreed with the student.

In addition, the company's name may or may not be used as indicated below. (Tick as appropriate.)

	Thesis	Conference paper	Journal article	Research poster
Yes	\sim			
No			\sim	

175

13 - 12 - 2019 <<insert date>>

<<signature >> DYLAN FLIN1

<<<<Insert name>>>>

<<On company letterhead>>

MARTIN Klopper , in my capacity as Owner at MKFD......give consent in principle to allow Luyanda Lincoln

Swaartbooi, a student at the Cape Peninsula University of Technology, to collect data in this company as part of his Mtech (BIS) research. The student has explained to me the nature of his/her research and the nature of the data to be collected.

This consent in no way commits any individual staff member to participate in the research, and it is expected that the student will get explicit consent from any participants. I reserve the right to withdraw this permission at some future time.

Research will be done on the time and dates as agreed with the student.

In addition, the company's name may or may not be used as indicated below. (Tick as appropriate.)

	Thesis	Conference paper	Journal article	Research poster
Yes				
No				

1/10/

<<insert date>>

<<Signature >>

KI-4p=1 TARTIN

<<<<Insert name>>>>

Appendix D1-D14: INTERVIEW TRANSCRIPTION P1-15

APPENDIX D1: INTERVIEW TRANSCRIPTION P1

Section A: participant's details Personally Details: What is your Name? Name of your Company? Your role in your Company (Job Title)? Chief Executive Officer (CEO). How long have been working in this organisation? 20 years, before that was with Old Mutual, before that was in Farming and before that was a Senior Agriculture Researcher. What does your company do? Primary we do few things: Compliances, Training And Regulator Exams. What is your highest level of education? Master's degree in Agriculture. Do you have any employees working in your organisation, if yes how many?

Yes, 150 full time staffs, and also use plus minus 120 contractors - Lot of these full time employees are compliance officers and on the contracting side lot of the contractors are being used as invigilators for the exams.

Can you tells us about your background of your organisation?

When we started our business 20 years ago, people didn't understand anything about information refinery because our business was talking about machine learning and artificial neural networks. At that time our business saw the value of big data and combination of individual data sources that would necessary benefit individuals with new information that they wouldn't been necessary got from their own datasets. We started the process called client value management (CVM), the purpose of CVM was to pull information from FSP and then to find information or gather information and to profile those clients, but typically one FSP wouldn't have enough data from a decent client profile exercise but if you combine them with many more your pull of information becomes a lot bigger and your learnings within basically becomes substantial better and more reliable because you gathering information across much large universe and so your repetitions / ends become much more. That was

an originally idea for us to open our organisation, over above the fact that we wanted to do compliance. We wanted to provide people which deals with us as a compliance business with information about their own clients that they couldn't gather from their information they typically had. That was 20 years ago.

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

RSQ 1.1: What are the perceptions of SMMEs towards the value of BD for their businesses? IQ1.1.1: To what extent does your organisation have experience with big data analytics? We sit on a lot of different data sources, we have not really done anything with the data rather than maybe doing a demographic slicing and dicing getting to understand the trends who is moving in to the industry who is leaving the industry because we have access to that kind of information via the exams and registrations but we haven't really done anything in my view valuable with the datasets we have.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

There are no obstacles rather than resource and time. It is only now we are working with a team of data scientist where we are basic pulling all the different datasets together creating the links between individuals in different datasets so that we can start seeing a more comprehensive profile of individuals, people that we have been dealing and working with, so that is in progress as we speak. But the really problem over time we get busy and start doing urgent things and not looking at the important things.

IQ1.1.3: How does your organisation categorise your existing customers?

For us as the business our customers has got a different views because we view our customers differently to that of a FSP, at the same time we need the same kind of information because I do believe that a FSP should understand what the customer value is, they should be able to understand enough about their customers to classify them and they should also start understanding enough about their customers to sort of classify them in terms of their personality standards because it's one of the things I do believe people do not focus enough about it, they look at numbers and they don't look at individuals they try to quantify things but they don't quantify and evaluate the fact that we are in a people's business not so much in numbers business. All my clients that I have deal with the compliance officer, so we transfer knowledge and the same thing happens with FSP, FSP value is the ability to transfer knowledge to you so that you give me information that I can use to help you to provide solutions.

IQ1.1.4: How do you manage your competitiveness against your competitors?

The more your know about your clients in this case as I said our clients are slightly different to what a FSP client would be because my clients are FSP so it's quite different. But the more I

understand my FSP clients the better I can service my clients and service all their different needs. I do believe that of all the competitors on this space we have the most information about our FSPs, we also have most information about the level of education, the number of years' experience and we understand more about themselves as many other people know about themselves. If I can explain we know in advance at this stage how long a person got before they meet a certain criteria to be complained, very few of our competitors will actually have that data, they would have that data but they wouldn't know about them and they wouldn't have that data in their databases which you can basic use them for other purposes. We busy developing a software application, where a solution calculates all the requirements that you would need in terms of experience and in terms of short term training, qualifications and continuous professional development. We also know from that system what kind of professional development you would need, also know from that system what qualifications.

IQ1.1.5: How would big data add value in the organisation?

Immensely because we know for instance by using big data, how many repetitions or datasets you would need to improve your reliability scores in your predictive modelling or information management. But for us what we would like to start getting into, is to start understanding when a client is getting to leave the industry, we would like to know when a client is not going to complete the qualification, we would like to know why a client is battling with a qualification, so all of that information helps us assisting and managing the outcomes of the people we view them as our clients. So depending on what angle people are involved in my business I need to understand that person so well, that person is almost well managed and coached before that person realize that these are my needs or my problems.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

Yes huge because some of the SMMEs don't have data, some of them do have data but they don't realise it, it is data, some of them actually have such a diverse set of data that it becomes almost meaningless unless you go the big data process or program. But very few FSPs that are work with or deal with really understands the bigger picture of what data can do for them. IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Very few of them are really focusing on big data, they focusing more on the client's needs or client's solutions without understanding the value of what they have in terms of big data. The other problem they sitting with is a lot of these guys don't have systems, they are having information, a lot of pieces of information stored in spreadsheets but it would be your data in that spreadsheet, it wouldn't be a spreadsheet combined with all the data and all the clients they have for that FSP. Some of the guys have a lot of data but because they have little

understanding of data management they don't necessary see the value of what they have within their datasets. I think there is a lot of potential but not necessary in understanding on what they can do with it.

IQ1.2.3 Do they have a desire to know about big data?

They all have a desire to know about big data, they all think they want more information, they want more knowledge about their client based. Very few people that I speak to or that I deal with would say I don't want more information about my clients, they always want more information about their clients but they do necessary understands what to do with it.

Furthermore question asked by the interviewer was, It seems that there is a really and serious need on how do I manage data in my business, and how do I know if it will give me value without being charge a fortune, without it costing me time, and without me putting a lot of effort?

That's very true, one of the things that we picked up as an organisation is that FSP will do basic data manipulations, but very basic they will say "they would select maybe group of same age, they would do selection on gender, they would do selection on maybe on occupation", but when it starts getting more complex when you used multiple factors to start selecting clients they just don't have the skills, lot of them have excellent excel skills in terms of doing spreadsheets, manipulating numbers and creating formulas. But when you getting to the statistics components of the excel abilities they do not have those skills. It's very difficult for them to find new clients from their existing client base because for them it will is very hard to say we will find new clients that don't know our product who are within this age group, with this gender, with these qualifications and who have these occupations and they don't use that information.

Furthermore question was how do you see that can change or what can we do to make that easier for them?

Other than developing short course interventions that can be provided electronically, eLearning or inviting people to have face to face discussions showing them using a small sub set of data how much information they can actual get that will give them insights and allow them to focus more on marketing, focus more and better on products solutions. The more you start inviting them the more they start understanding the value of big data, from them talking about data means complexity it doesn't necessary mean insight.

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage? RSQ 2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Absolutely because if you look the way we are going with the fourth industrial revolution, the person who will not be using data, is going to be probably obsolete in the next five years from financial service industry. I am saying that because it will take a while before we had enough

data and enough individuals and people who will sit down and start managing data to the benefits of the SMMEs but in time the value of an advisor is going to be redundancy as more and more data are available and more self-service applications and solutions are developed. IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? There is a set of solutions and another issue that these advisors belief they have is that think they understand the product and product outcomes better than anyone else, but products that being developed in South Africa becomes extremely complex with multiple numbers of variations, if a machine will be in a much better position to ultimately match a product combination to the client's needs about the individual's ability to select or put that solution together, advisors do not understand that they still belief because they know that particular product well, I can confirm that the basic 95 percent of the products they understand very well but there is still a 5 percent of products that are extremely hidden and complex and especially if you start looking at things like disability cover, income replacement they are so many implementations that you can actually look at where an individual just cannot make those combinations work. I mean anybody who's seen the big break through with the go game where people belief that it's impossible to beat an individual after doing a bit of machine learning, the system learned enough about the problem and the solution, so if you have half million problems and solutions the ultimate combination of solutions probably be a lot better predicted by the machine than an individual. So to satisfy the need of the client you will need to know so much about the client, so much the client's circumstances around or surrounding the client. But also understand so much about the products that you can actually provide unique solution to that client, which is very difficult for an individual. So you be probably satisfying 95 percent of the client's needs but not the full 100 percent.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

Yes we have a strategy and we are busy working on that strategy, is it fully operational at this particular stage no. But there is a different strategy because we understand the diversity of the information we had on individual in different datasets and we also understands the value of combining those datasets and starts finding information about people that we normal won't make leads on them.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

Probably every three months we have strategic session, we won't go in depicts about things but most about red flags, problem areas are looked at and once a year we had a depicts strategic session which is about two days we would look at different aspects about things that will influence our business. IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

It's an ongoing process, we continuous look at what are our competitors are doing. I can give you an example some of the compliance work is very repetitive, we looking in ways of how can we take a repetitive component and start using online solutions to take repetitive component out of the time the individual is spending with my client, so that the individual can almost start doing repetitive work rather than my compliance officer because my compliance officer is quite an expensive resource. And the compliance officer really starts adding value where the valuable contribution to my client would be sharing knowledge, skills upliftment and development rather than sitting and doing something a less skilled person can do, it's like taking a file and checking that all the information is in a file and not doing any interpretative work with that information itself, that's a very repetitive process. So in our view no other compliance officer solution is looking at that as part of their long term strategy, but I believe we have to start using time a lot more effective so that the value we provide our clients is a lot more than merely just spending lot of hours with the client where 30 percent of that time is being used for meaningless tasks rather than 100 percent of the time actually value adding with our client base. So things that a machine can't do that's where we should be adding value where we ask you a number of questions and that information is logged, we can write algorithms with datasets to say this is a red flag this is a green flag this is an umber and manage that almost from that point without intervention with any of our expensive resources. IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

No I don't think you have missed anything or there are things to add but I think the environment you are conducting these interviews within, unfortunately the skills these people have is to deal with people that's their skills, it is not the knowledge about financial products, it is not about the ability to identify needs and provide even solutions. The skills that these people all have collective, it is an ability to interact, to keep people, to build relationship and to build loyalty. Unfortunately the people that you have interviewed going on through this process are not well informed, they will tell you about data, and they will tell you they have data but they are not really focusing on managing data and optimizing data, just some the comments you made earlier they will tell you I am unique and my offers are unique it proves to you that they are not focusing a lot more than looking at what competitive information they gather, every clients they gather they have client information sheet and that is such repetitive. And they still believe they are unique. So what the problem is there is so much work needed to uplift them from their level of understanding big data so that they can really value, ask the correct questions and get the right information their clients to enable them automatically to provide the utmost value solutions for their clients. For me knowing that I have got the right solution is

137

more important than to have individual gathered information. The most important thing within the financial institutions is trustworthiness. People need to trust them because they are dealing with the most important information about people, they are given this piece of information hoping they have enough integrity, hope they have enough understanding to build or provide solutions to clients which will ultimately benefit the clients and their families. Another problem they don't understand is that clients are putting their families' wellbeing in their hands. They think about a client as person who provides income for them, not so much a client who they serve, who rely on them and trust on them to ultimately sure when a client cannot longer work or earn income, whatever they earn is in such a good environment that their future life or future life of their families is so well looked after at the way they don't run out of money or hardship. I don't think these people understand how important their job is, they get involved in financial services because they are told is easy money.

APPENDIX D2: INTERVIEW TRANSCRIPTION P2

Section A: participant's details Personally Details: What is your Name? Name of your Company? Your role in your Company (Job Title)? Head of risk compliance How long have been working in this organisation? 4 years, previous I was with Alexander Forbes group for 15 years as investment solutions , I started five years as a legally advisor, another five years as a head of compliance and another five years as a head of risk and compliance.

What does your company do?

We aim to help people all over the world reconnect with and maximise their UK pensions and explain things in straight forward language.

We constantly advise clients not to ignore the growth potential of their existing funds. A big drag on pension performance can be high charges. That's why we combine world class pension advice with low fees so you keep more of your money.

What is your highest level of education?

LLB degree plus a post diploma in compliance management.

Do you have any employees working in your organisation, if yes how many?

Yes we have about 30 employees working for our organisation.

Can you tells us about your background of your organisation?

Up to the 26 November 2019, we were Devere Acuma which is Devere internationally financial service provider. Based in Dubai, has got 53 offices around the world. Recently the

company has been taken over by Brite advisors who is mainly located in Hong Kong and Sydney and it's completely 100% shareholder.

We all want a brighter future, especially when we stop working. We believe we can grow your wealth so you can have a retirement to look forward to.

Brite Advisors offer you an all-in-one pension service – from advice to transfers to pension administration and asset management. This means we don't use middle men or third parties – so we keep control of costs and you keep more of your retirement money.

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

Very little, it's something that was always highlighted at but not by the name of data and analytics because the management structure in the Devere domain was very much more sales driven, less driven on software, data, analytics or academia which was more my passion so I brought it in because I chair most of the times in the exco meetings. We talked about it but we never put it under the umbrella of big data. It was always about what our competitors were doing, we didn't have formal tools and skills to manage data.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

This is my personal view on how I have experience at the management level, because our head office is outside the country the cost to provide money to the South African branch was always and is always a big problem to the guys, I think skills and understanding of big data and there is no specific focus personae for the skill, someone who could promote it. I don't think we have the necessary systems or tools and implementation and change management for me I saw that throughout South Africa even at Alexander Forbes when something new was brought to the organisation there was always a huge push back and at Alexander Forbes they used to brought in change management people to assist us with rolling out things and I assume that's the thing FSPs will need to implement when rolling out big data.

I think the biggest two things I could foresee will we be struggling with is what are we going to do with this big data once we have applied or use it, then it's a complete new thing, new process, new functionality and training that we will need to roll it out and there other thing I foresee we will struggle about it is the continuity of it.

IQ1.1.3: How does your organisation categorise your existing customers?

Very simple model, you look at your types of products you sell or provide (e.g. A, B and C) and your existing clients. Under A, B and C you got 1, 2 and 3, this is how big their portfolios are and then how long you have stored client B in your available product(s) before it matures.

And all of those thing impact your bottom line eventually and in retention of the client. That's how simple it is.

Further question was asked to the participant "Is it based on the affordability or just how long they have been paying their premiums?

It is based on how long they have been paying their premiums that also have an impact when you retain them because we look at the coming fees consistency and the reinvestment.

IQ1.1.4: How do you manage your competitiveness against your competitors?

Big issue for me we don't manage it because Devere was so big and they were so big in UAE, UK and US and they didn't worry they would say this our model and this our products then go. We are number one, never thought they are no longer number one because they have closed over 30 offices in the past four years. But overall they were doing well but not doing well in South Africa, for them I think they never understood South African market very well and also not huge on regulations and it's a company that they are very innovative but innovation comes with less regulations.

IQ1.1.5: How would big data add value in the organisation?

I think it's an internal and external leg for me, internal you will have more staff satisfaction, keep your staff happy and retain your staff. You can upgrade your systems and processes to me its systems and processes, you can do better for your employees to operate better and clients will benefit better services, keeping of clients and be better competitive to your market out there.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

We haven't really explore but I think there is a lot of data floating around but it gets used on an unstructured basis.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

No we will have to employee or get the external people to assist us with it. It will be the skills and tools that will have to get it in, get trained up and then roll it out. There is definitely will be people in the system that we can utilise or be interested in it to take it forward.

IQ1.2.3 Do they have a desire to know about big data?

I think the current management they are, I must say the bright model is also more on products.

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes definitely it will make our organisation competitive, we will be able to identify short or long term clients or customers' needs and real needs, and also maybe how to assist them to identify

140

their needs. We will be develop products for our customers and release them better with a more positive intake of them.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? I think once we are able to identify our clients in the right product that suits them and also I think your competitiveness, you will be able to measure it because in some instances you will be able to say I can't provide this products because not all the companies are the same, but at least you will know what you dealing with, currently you working in the dark and you think you are the best and maybe you are not the best.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

Unfairly because we are using CRM, we got a system that aligns with it called Core, we store all our clients data / information on it first and mostly all the documentations which is required in terms of various pieces of registration information like FICA, and then we keep our policy documents, we keep the variations and personal data.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

We got exco's at least two times a year, in these exco's I prepare a pack and also managers prepare some packs and if you look through our reports you will see a SWOT analyses but we don't call it that and I don't think they always realize they doing it but we do it in an unstructured way and still we have the same problems there that you would have had it, but I think what you do with the SWOT analyses outcomes if you don't put actions against it there is no gain for you. But I might say from 2017 and 2018 I think we got some traction on it, so I think it can be always better if we got the right tools and skills.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

I think not in this office per say, it happens often but when we get in the past when the CEO travelled down here, he is looking more with an eagle eye on the business and he comes with the say, did you see how our competitors doing that, they are offering this and did you know they brought down their fees but once again it doesn't really impact because the guys are very stuck in their model as it works because it makes money for them, it is so much better on compliance and as I said earlier the idea is that we are the better don't fix something that is not broken.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

I think for an academic point of view you have covered everything but if you were a consultant you would probably provide us guidance on how we should go about or what first steps to take if you are interested to bring big data in your organisation.

141

APPENDIX D3: INTERVIEW TRANSCRIPTION P3

Section A: Participant's details

Personally Details:

What is your Name?

Name of your Company?

LifeCheq

Your role in your Company (Job Title)?

CFO and look after the advisors

How long have been working in this organisation?

Since the start of the organisation, we are 4 years old, before that I was at Old Mutual and at Old Mutual I was quantitative analyst. Basically doing financial Maths that was more in big data than this role I am doing at LifeCheq.

What does your company do?

We are financial consultants, what we do is we help our clients organise their finances so that they can achieve their goals. We are independent and we work with 29 different insurance, investments and other companies.

What is your highest level of education?

Masters in Finance.

Do you have any employees working in your organisation, if yes how many?

We have around 55 employees.

Can you tells us about your background of your organisation?

Our founding team of highly qualified individuals spent years working for one of the largest financial services companies in South Africa, and are qualified across Finance, Actuarial science and Business science at UCT.

Collectively, this team has four decades of experience solving some of the financial industry's most challenging problems, and has now taken this experience and applied it to the personal finance sector.

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

RSQ 1.1: What are the perceptions of SMMEs towards the value of BD for their businesses? *IQ 1.1.1: To what extent does your organisation have experience with big data analytics?*

In Lifecheq simple gathering information is very difficult, so I think here we are not anywhere near to use big data, we collect lot of information but I think when you said big data is structured or unstructured we don't know how to use big data when it is not well reference somewhere and we are still building that. We are getting there but right now here we are not really using it that much, for us things around like clients emotional state for example is actually far bigger driver to the success of our business rather than how much information we had about the

client. So its little bit suckle in that way but we will get there I think that's like a revolution of the process but right now not that much. At my previous company Old Mutual we did evaluation of liabilities with hundreds of thousands of entries with each policy holder so there, there was a lot of analytics on information and a lot of them was about assumption based instead of being like factual based, its similar analytics skills and analytics programs that we use, but there we were determining the behaviour and patterns of which from that particular data the behaviour and patterns was based on assumptions, and from there the company had lot of millions policy holders with lot of information to build from there.

IQ 1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

I think here (Lifecheq) the main thing is collecting the right data and collecting it reliable. Our information for example is a lot about someone's personal life so our consultants feed that information in our system for example this is how much I earn, this is how much money I have left, this is how many people I am supporting in my family and we look at various things about the person and a lot of that is qualitative information getting that into the right structure and become usable is quite difficult for us at the moment and also getting that information from people in a consistency way is hard. For me getting the process to get that right is tricky.

IQ 1.1.3: How does your organisation categorise your existing customers?

There are different ways of categorising our customers, for example by location Cape Town's clients and Johannesburg clients are not the same, so Johannesburg clients are generally more kin to try new things, they are also in our world as we are giving financial plans like we looking at their finances they often like people who have side hustles, unlike Cape Town's clients generally don't have and they are little more engaging, and they ask lot of questions than Cape Town clients do. And our profitable because they want more investments plans for our business, although in terms of the fees we charge almost the same.

IQ 1.1.4: How do you manage your competitiveness against your competitors?

So I don't think there is another company in South Africa right now currently doing what we doing. We do have competitors because every company has competitors but not in the same way we do and for that one I will prefer not talking too much about it because that's quite a key for us right now like how we are positioned. You can go to our website we charge a fee for our service and they are people current doing that thing but they are rely offering the service to a very wealthy clients for them you need to have at least ten million rand to approach them otherwise they will not be willing to assist you. Us we are trying to bring that quality to the middle class.

IQ 1.1.5: How would big data add value in the organisation?

Yes definitely, because we deal with number of aspects on client's financial life one of them is an insurance that requires a lot of information about person's medical history and there is definitely a lot of applications on big data out there about client's medical history like what kind of medication information that will actually lead to higher claims and I think there are some of new insurances that will be starting in the next coming year or years to do this kind of thing. They will try for example take the medical aid data and use it to improve their pricing.

Further question was asked: Do you think it's going to be big companies doing that or startups?

Answer: It's going to be start-ups, I know one that's going to do that, and I know one of the big insurance company that was trying to do that and they are struggling and I think Discovery are also trying to do that, although they saying customer's information are confidential. In theory they are not allowed to release client's medical history information to their own insurance company because they are setup as different companies but how much they are allowed and how much they are not allowed I am not entirely sure.

In our organisation I think big data and the use of data will probably coming from the user interfaces, from how people interact with our systems because more and more people would be interacting with our systems and website and it will be like a good way of designing a good interface because it will be good use of using big data. Financial recommendations I am not entirely sure how much it will be use there.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ 1.2.1: Do you think SMEs are having challenges in adopting BD?

Yes the recording and how clean the data is, is the hardest part. I think if it's a system, if it's like a website that can provide a functionality to our clients to review their medical aid for example that is easy because the system can record it but we are recording all lot of information directly from our clients. They are sometimes type their information on our system or our consultants called them and type their information on their behalf.

Furthermore question was asked to explain the recording more?

Response was: so when a client comes to us, the first meeting is just to gather their information, we don't give advice in the first meeting we just gather their information. We normally ask question like your family, your finances, what you want to do in life, things like do you want to travel and would you like to start a family. So turning those things which are like golden dreams into data that can be recorded and aggregate is difficult in a way you can still have a quality conversation with a client. And we can never take away that quality conversation with our clients because clients would think they are interacting with a robot and we can't afford to have that. Setting up that whole process is very difficult. Its kinder a process design on how to gather the information, the actual data at the back is sort of like this person wants to travel to Paris more than 3 times a year, that is very easy to structure, gathering I think that is a the hardest task to do because we dealing with people and people's goals and dreams.

IQ1.2.2: Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Yes, I think we have do people here and my experience and other people's experience, we all do have an experience to analyse data because we have done it before.

Furthermore question was asked: do you have statistics degree or its self-trained experience? Response was: we have background degrees.

IQ1.2.3 Do they have a desire to know about big data?

Not really because most of us actually knows big data and have dealt with.

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Not in the short term, I think for example if we were operating like a company called Hippo, I see us we will be doing more of big data because hippo is a website that compares a lot of insurance quotes from different organisations, there they have a lot direct comparisons and their website is a direct service essential and I can foresee in their world doing a lot of recording on how many visitors they have and where are these people clicking and how many people click through right to the end, where are they getting lot drop outs.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? I think ultimately the use of information and statistics can be used a lot on the pricing of the product, which will results in cheaper products for the clients. Insurance companies will set rules on calculations premiums based on a gender, age and location and then offer a specific price for the customer, then a person can come along and notify them with his / her chronic disease and maybe the insurance will then decline / refuse to insure that person because of his / her medical risk.

Further question was: Do you see that one could use big data to manage the risks profile of clients and well as products?

Answer: I think in the pricing of the products you can use big data because there are lot of things you can pull from the customer's profile that will help you in identifying or answering some of the questions that you would have potential ask the client.

And in terms of the risk, yes you can use big data in identifying all the potential medical risk your clients might have before even offering the insurance because you can pull all the medical history of the customer and use that in identifying risks.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

No there is no strategy at the moment, just to elaborate we are only 4 years old operating, we are still very young, we are still looking at our five year strategy but in term of big data strategy

we first need to launch one or two services before implementing it and those services will probably launched in the next coming year or next coming two years.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

APPENDIX D4: INTERVIEW TRANSCRIPTION P4

Section A: Participant's details

Personally Details:

What is your Name?

Name of your Company?

Moore Wealth Management

Your role in your Company (Job Title)?

Managing Director

How long have been working in this organisation?

20 years with company and before that he was with liberty life for 12 years.

What does your company do?

A composite financial advisors, they provide advice and services across a broad range of financial products such as services to individual panning; pension fund; shares; medical aid; provident fund, group health care and covers a full spectrum

What is your highest level of education?

Certified Financial Planner (CFP)

Do you have any employees working in your organisation, if yes how many?

9 employees

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

No experience, they are little bit comfortable, it is a people centred organisation with 120 employees, 110 are on the accounting side rendering service in auditing, tax, book keeping. Their nine employees engage with the accountants based at Moore, Cape Town who provides them with leads and referrals. They don't need to go and find new clients and the leads that they receive are in a simplified manner i.e. please call so and so... she has Million to invest. They do not analyse or search the database for information on who to approach. They have existing products and based on the client needs (what their circumstances are and would they want to achieve) then they match their services and products to suit the desire of the client. IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data

analytics?

None, from theirs side is the collection and analysis of the information. How do we collect and analyse the information. If they would like to use the big data, they have to source their own BD, what would they look for in order to gain more clients and sell more products and that is the bit they don't know.

IQ1.1.3: How does your organisation categorise your existing customers?

They categories their client as good or bad clients, you are regarded as good of you pay and bad if u don't pay. The Moore Wealth Investment company is remunerated monthly the product provider companies in a form product advise fees provided the client has listed them as their service manager. They also decide whether to sign up with a client based on the ease of doing business with a client, this was substantiated by an example of A client with R10 000 000.00 and is difficult to work with and a client with a million rand which is easy to do business with. How much ease to deal with client i.e. customer satisfactory and ease of doing business.

IQ1.1.4: How do you manage your competitiveness against your competitors?

They improve their competiveness by attending number of road shows and numerous roadshows, presentation and engaging with their peers in other organisation to find out what they are charging for an advisory fee. They normally charge halve of a percentage of their annual fees. Lot of the companies goes above 1% of their advice charge, and Moore Wealth Management do understand where they stand in that competitive profile and more recently and those percentages are coming down due to poor return on investment. The performance of investment are analysed through number of programmes and they contact the clients to give them a review once, twice or three times a year. If it is one a year they go into a profile and identify what types of portfolios that the client is exposed to and provide an opinion on the their portfolio performance which is based on relatives to the average fund on that sector, and advise if they should stay sell or move. The review is done every single year. For February 2020, they already have 15 clients that are due for review. These reviews varies depending on the client's need, some are informal and some more formal conducted in boardrooms in a structured format with presentations and minutes. However, all depend on trust. IQ1.1.5: How would big data add value in the organisation?

It will add value if they can identify which client have got certain assent and income that will make them an attractive client that they would like to engage. In that there is also a risk profile which is already built in.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

They don't understand BD and how to use it or implement it to their advantage and to them BD is foreign. An additional question was asked if the interviewee has a desire to know more about BD? Responding to that they've indicated that they are at their 50s and comfortable, and have great business going and don't see the need to expand however, they have younger generation that needs to pursue and make use of BD to grow the business. The younger generation is much isolated and unlike the older generation who grew their business through social contacts and activities such as playing golf, they need to look into other ways to find clients.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

No they don't

IQ1.2.3 Do they have a desire to know about big data?

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes, if there is a way that the information at their disposal can be interpreted to their benefit so that it narrows down the field of the potential clients that they need to engage instead of knocking in every door, it then categories the top 10 clients that they can engage, based on their high income, wealth, geographical residence and etc. rather than wasting time on 100 clients that they are not quite sure where they fit in.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

It will save a lot amount of time because they can target the best prospect for success and the product to the need. Therefore, they would not knock into a 65years old to offer a life policy.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? From time point of view that they do not waste their time to market the unnecessary services to them i.e. more Cape Town has 400 client and the other branch has 3000 it would be advantageous to find out more about the 3000 instead of bombarding them on things that are not of value or relevance. What are about the products i.e. Assets management side noting the things that are happening with the unit-trust. Is there no opportunity to make a tailor made Moore product for them? They have something similar to that, their own unit trust which works

in this way, a client have a million rand and they see them as a good investment for Allan Grey and they go ahead and invest.

But when you need, to do reviews and change underlining funds of 400 to 500 clients is time consuming and difficult to make a single transaction of each client. What they then do is to create a structural unit trust, but when you want to effect changes on the trust it affects all 500 clients. The challenges with this approach is the competitiveness and a bit costly there as extra fee from their side and the from assets management consultancy fees from where they get the information from. Therefore they have to pay 0,4% more to get someone in that structure, however, they are busy exploring ways to reduce the funds.

Follow up Question: Those asset manager do they detect to what is in the unit trust?

They do not dictate what they want but rather look for an asset manage who has what they want i.e. design a shared portfolio that has for instance 50 shares to suit their needs, but at this juncture they make a judgement on who they need to invest with based on the performance, are growth, value and how well do they correlate, in essence they mixed them up for example Investec correlates well with Alan Grey.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

No they don't have

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

They do it formally in January, thereafter quarterly and informally once a week, to reflect on new ideas and exposure that the can adopt for implementation.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

They don't have much involvement with other colleagues in the same industry so, periodically look at the outputs from other financial advisors and compares it with theirs and in 90% of the cases they offer better services and confident on that.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

Do they get the copy of the output of the research? They will get an executive summary. Follow up on the BD presentation that was offered in London

APPENDIX D5: INTERVIEW TRANSCRIPTION P5

Section A: Participant's details Personally Details:

What is your Name?

Name of your Company? NFB Private Wealth Management Your role in your Company (Job Title)? Chief Executive Officer How long have been working in this organisation? 15 years with current employment 9 years with Old Mutual What does your company do? The company provides highly specialised, fiercely independent financial advice, products and services. What is your highest level of education? National Certificate in Wealth Management and International Capital Markets Qualification. Do you have any employees working in your organisation, if yes how many? 30 employees

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

They are a small company with no experience on BD analytics, but they do get their analysis from big companies like Discovery or old mutual which have one of the biggest database in the country. He gets most of data through connection, attending meeting to meet the right people and introduces himself, should he need any information he will contact them later.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

He has not come across any obstacles and it is difficult to answer, however obstacle is basically how easy to get the data, how reliable is the data for example Statistics South Africa is highly politicised and they don't provide the real stats.

IQ1.1.3: How does your organisation categorise your existing customers?

On occupation, gender, age, affordability, health. You said 500 tax cant go to a risk person with a product with a product like life insurance services, but he can service his existing life insurance. If he has a different view in a client, you can't make changes because they can't buy again.

IQ1.1.4: How do you manage your competitiveness against your competitors?

He is trying to be a one stop business, and see to it that his clients' gets advice from experts. Because they cannot be experts from everything, short-term, long-term, life investments, I they are a lot of companies now about 900-1000 that you can invest only in unit trust. There for you cannot do business with all those company, but you need to choose one that you can invest with. Currently they do most of their investment with Alan Greys because it is a company in the JSE and don't have to pay dividends to their clients. They have specialist that are always talking to the client, and always attend trainings. They are very good fund managers, always available even telephonically.

IQ1.1.5: How would big data add value in the organisation?

The more data you have the more business you can do, and focus on various things if for instance BD can tell him that company A, is looking for benefits for their people, you can approach them and sell services to them, inform them about your abilities, your experience, what you can do for them. On the computers or systems to make it easy, meet new client, retrieve data we need, currently of they go to someone they don't know they are just guessing and you don't know who you are walking to.

How is business, tough or easy?

Business is quite tough. He comes Worcester and has an ability to do work for the municipality but he does not have a BEE certificate, therefore he cannot provide services to them. The municipality gives business to others who resides a distance away from its vicinity while he is there and to provide personal services which is very important element of the business. It important to have personal relationship with your clients.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

No, they don't

Is this is something they are thinking of implementing?

They store most of their data in files and their backup system.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Yes they do, they belong to Moonstone in Stellenbosch a compliance officers, where they can source data in a certain way, their tools, electronics, internet are always 100% and if they can get big data they will be able to handle it.

But do they have necessary skills to handle the data?

Yes...

Is there any challenges that you think can affect SFPs to adopt BD, potentially?

Yes, they are being bullied by cooperates, which prefers people to work with them directly, they want people to do business through call centres, they do sponsorship, working with the politicians. They think private independents are taking away their business. For instance if you are looking for a quote of a short-term insurance and I give you a quote from 3-4 companies they fight like a dog over the bone. Mr Klopper tends not to like, but he gets a good rate for his client. He bullies them back, by saying that your client has been good paying you premiums for so long, small risk, and he can get a better rate for the client somewhere, if you still want

to keep the client reduce the rates. They don't like that at all. But hey need us for data i.e. information about marital status, number of kids, what kind of business he is in. it is difficult for them to get this information, hence the dependent on Mr Klopper.

IQ1.2.3 Do they have a desire to know about big data?

Yes

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes. There more you are informed the better the advice you give.

Coming from a semi-rural area do you think there is more opportunities to analyse this massive amount of data that is available for company to assist in improving your portfolio and offering? Yes...

If there is an opportunity of having access to BD will utilise it?

Yes.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

They use in trauma instances e.g. in this case they can go to a client and say they are still insurable and they have data that shows that people of your age are still suitable for a retake, you are smoker and have data from a reliable good source. I think we can increase your trauma insurance that can do you well than a life insurance in this instance i.e. breast cancer in women, 70% cause of death in women and medical aids has removed post trauma treatment. In some cases it can result in the removal of the breast and most medical aids don't pay for a transplant and reconstructive procedures.

Do you think with BD can you be able to improve profit, grow your business in terms of number of people?

Yes, it can improve the quality of services you provide and increase number of customers, however, you need to service your existing clients well and acquire one or two client a month that is a good business.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? Yes definitely depending on what BD gives you, you can be able to categories clients based on their statuses and prioritise accordingly.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

No strategy at the moment, but uses some contacts to solicit data.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

We know how many client they are losing and also find out why they are losing them if a client walks in they always ask for feedback on the clients experience on their services. They always discuss the feedbacks from clients in meetings, and what the causes of losing clients, e.g. we lost client A and he said we take long to provide feedback. We got a phone in client that was referred in by Mr and Mrs Y and they do offer Mr or Mrs Y with compensate.

Opportunities they do sponsorship, they are going to sponsor women golf in Worcester, and they are going to get data through Lukey. During the golf event they are going to run a competition with a prize, and participants will be required to write in a card their name, email address and contact details. They will also ask if they can contact them to inform them of their services and products. That how they collate this data which they sit will all the data.

Do you know the potential base of clients in your area and their profiles?

Yes more or less, that's if BD plays a role.

The threats in their business is more of empathy, sympathy type of thing. Example: a client comes from Pretoria they have a deaf or blind child they give up their jobs in Pretoria and insurance is a good start of a business and they tend to sympathise with in that type of situation.

Do you have lot of those kind of clients?

Yes young people always do come for advices

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

Daily, they have a company Martin Optimum specialised Investment Company, takes our clients' money and put a bit with Investec, Alan Gray and other companies and we are analysing that all the time. They have got specialist analysing the market making projections and forecasting on what will happen in the market, and they advise client that they must not get out of the market. Markets are very cheap even when there is no growth.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

It is difficult, the big problem they have in Financial Service Board (FSB) we don't know where they stand with them, they policy them like hawks, they attack individuals.

APPENDIX D6 AND D7: INTERVIEW TRANSCRIPTION P6 AND P7

Section A: Participant's details

Personally Details:

What is your Name?

Name of your Company?

Wealth Growth Financial Services

Simplicity Asset Management Your role in your Company (Job Title)? Fund Manager Financial advisor category 2 How long have been working in this organisation? 2004 14 years old 2009 10 years old He has 32 year experience What does your company do?

Financial services company provides financial service advice registered with the FSA. They provide 13 of the 16 services that can be performed by a category two licence, however, they do not provide services on funeral covers, short term insurance medical aids although they are registered for it. Decretive financial services has to have a mandate of every client as they are authorised to sign on behalf of their clients if they change something on the client's portfolio the client don't have to sign but they can sign on the clients behalf. The category 1 have to go the client with three option that to choose on, owning to lack of knowledge from the client, often they seek advice from the financial advisor on the best option.

Simplicity Assets Management, operates on category 2, they don't have any client but manage fund. So Mr de Viliers' role is to manage the fund that wealth growth provide to their client. Simplicity operates similarly to Alan Grey, Investec and Coronation. It is the investment vehicle which has three fund: 1) income fund; 2) manage protector regulation 28 fund; and 3) international fund.

What is your highest level of education?

BCOM degree and CFP

Do you have any employees working in your organisation, if yes how many?

Three employees WG

One employee

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

Zero for company 1

Zero for company 2

However they make use of Personal connection and also they have a company that they pay between R10 000 to R20-000 a month to source information for them. Upon receipt of those

information they set a personal interview with the prospective client to sell their services to them. All their client data they source it directly from their clients.

Simplicity assets management: as aforementioned, they do not have a client but they make use of international data i.e. financial market data (Specialised strings of data) on a specific specialised level that runs through bloomberg, MSCI international sites. The company that they have contracted source data and analyse it for them and put the funds together noting that the market keeps on changing. They compile the fund composition form the data received from different sources. This data is analysed daily but actioned in a quarterly basis. If you manage the fund actively does not mean that you do not have to do anything to the fund. If something goes out of margins they need to effect changes. The difference between simplicity and Alan Grey is that Alan Grey is a big asset fund management company and their assets is more than all investment companies in South Africa combined and they cannot invest in small companies and they are investing into specific shares at the stock exchange. Simplicity then invest into Alan Grey assets management fund giving them 10% of their fund and also look into six other fund for a greater in return in investment. Therefore to achieve this they look into an equation that provides them a better income.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

They don't use big data, however to answer this he indicated that he just attended a course on behavioural change in financial management, gone are the days wherein you sell products to a client but you provide services based on client needs. Make use of artificial intelligence and advocate for the human intervention to build relation. A company may make use of BD to source clients and have over 2000 of them but profit less but have numbers to run.

Simplicity, is compiling funder fund and their mandate is on the fund and not client orientated. They have three funds amounting to R670 000 000 divided into three funds; one being income funds, of which their desire id to have this fund to be the best performing fund in the sector irrespective of who is the client, and used for those client that have a purpose for an income fund. The flexible fund which is their flagship fund they want it to be the best international fund accessible in the SA market which is more than the 18%. They benchmark the fund against the MSCI world index and strive to be the best. If the rand devalue the funds goes up because is a rand denominated value but all the funds are in dollar terms. This fund is good for a long term and not for short term because it has an average of between five to seven years investment term, however, this fund does not need to suit and individual but strive to be the best performing fund within its category.

The only obstacle is to segment the data to get the correct information out. The Fair trees asset management company is running big data and they need over 6 hours to overnight to download such data from Bloomberg. Is not a cost issues but for quality and they are paying

a company to download data and analyse. However, the cost of dada is getting cheaper by day.

IQ1.1.3: How does your organisation categorise your existing customers?

They clients are categorised as A; B; and C based on the income they generate for them.

A client that is making them more than R30 000 a year

B client R15000 a year

C client making less

Simplicity don't have a client

IQ1.1.4: How do you manage your competitiveness against your competitors?

They believe that they do not have any competitors, there are 370 companies that operates similarly to Simplicity based on the client preferences to retain the clients.

Clarity question: Are your client relationship or performance driven, their relationship is built upon performance or perception of performance. They have lost some client based on poor market performance in the past and clients felt like it is the responsibility for the company to invest client funds appropriately and rather invest with Alan Grey directly rather going through them. Their income has lowered from as it was last ten years ago and their expenses are four times higher.

IQ1.1.5: How would big data add value in the organisation?

On Wealth growth uses specific data received form old mutual, Sanlam and etc. and pull the information into a spreadsheet and they do not use generic big data and on the other hand simplicity uses big data received from Bloomberg and other sources and allocate it for use in a particular fund of relevance to the information.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

They do not have any challenges,

As aforementioned, the Wealth growth they will not need BD for clients as they need to understand the client needs and provide the service. However, what they foresee is not a challenge but an obstacle that is to limit expenses because the moment you make use of BD, you need to appoint someone to analyse the data for you, and what you require is not data but a client information.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Wealth growth do not have analyst

Simplicity make use of the analytical tools but they have commissioned that service to a company to provide analytical services

IQ1.2.3 Do they have a desire to know about big data?

Not really for Wealth growth

For simplicity yes, but as soon as there is a new data the analyst will know about it, they have a Branch in Mauritius and also have people international that supply them with information daily. As soon as there is new information they will know about it and feed to simplicity.

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

No they don't see any benefit in using big data to create competitive in Wealth Growth. With further clarity provided on this point the interviewee felt that they need BD to retain the client and also to provide tailor made products.

On Simplicity, yes they see benefit in using big data to create competitive.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

It is compulsory to use BD to retain the client, there is no rationale for the business effectiveness if you cannot use BD for both companies.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? Not for profiling. But they need to be on the highest knowledge based on the latest information on the international economy to advise the client. The client perceive the financial manager to be knowledgeable as well as the source of knowledge.

For simplicity, they need to have the latest knowledge on the international markets trends and that information is retrieved on the internet through Bloomberg. Discovery analyse the national needs and identify the gaps and capitalise on it by developing a product that they make money out of. But it is a little different in the financial industry based on the regulations. Whatever product they provide to the client they are responsible for the product and if it fails they should account for that unlike big companies.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

Not really

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

Not through BD but personal introspection, however they are continuously on the learning path through courses and attending webinars with international role players.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

It is a continuous process which is performed daily by visiting financial markets and products but done regularly for both companies. IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

Yes, we cannot leave without BD any more but the cost is not to get it but by getting relevant data to add more value. Big companies like Discovery they have 10-20 people analysing data to develop a good product to sell to their client but a SMME can't do it. For example Sanlam legal team is 10x bigger because of compliance litigation on them and small companies cannot do that. As an SMME you cannot invest in a company which have under 3-5 years of existence and do not have over 100 000 000 rand in asset management if not they have to do a due diligence to them in a regular basis. This apply in both companies.

APPENDIX D8: INTERVIEW TRANSCRIPTION P8

Section A: Participant's details

Personally Details:

What is your Name?

Name of your Company?

First Avenue Investment Management

Your role in your Company (Job Title)?

Chief Operating Officer

How long have been working in this organisation?

Seven to Eight years

What does your company do?

Nedbank Capital

What is your highest level of education?

Investment management, pension fund for government, individual and companies as well as unit trust funds

Do you have any employees working in your organisation, if yes how many?

Seven employees

Can you tells us about your background of your organisation?

M.Sc. MBA, other people have different specialities such as engineering. His MBA focused on financial risk management and derivatives.

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

They have generally used BD on company information to analyse their stocks, not so much on customer needs, but on investment side and they have used data extensively. However, it is a challenge because the sources of data is not always available. IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

There is a big drive currently for ESG investment and all that information is not available because data regarding how companies are impacting the environment and not everybody is disclosing how companies are using "x" as amount of money, who record all the governance issues around it. If there is that kind of data available it is probable expensive and there might be a company that is making lot of business out of it. There is a limited data available e.g. you have a retailor that focuses on fashion, fashion has a lot of economic problems or environmental problems rather that is how much water is Mr Price is wasting on manufacturing of their garments rather. It is access and the quality of the data that is the main obstacle.

IQ1.1.3: How does your organisation categorise your existing customers?

You need to go back and determine what type of business it is or SFPs.

Some FSPs will focus on the retail market and will segment their clients based on high net income individuals and low income. Them specifically they specialise on institutional market, that is pension fund directly or as a consultant who advise the pension fund or alternatively institutional customer will be a multi manager and their clients are identified in a market as registered entities. But it is in a retail entities who interface with their customers and can be able to categories them accordingly.

IQ1.1.4: How do you manage your competitiveness against your competitors?

On the institutional space they do not focus mainly on the competitiveness against their competitors but theirs is to consider which asset classes is most effective to their business, their main focus is in equity focus on investor and analysing them in terms of style and that where they need to find their niche, whereas other managers may focus on equity property investor. In essence they are quality style investors, others focuses on value growth and momentum is more indexing and they compete among index investors.

Going back to BD, there is a limited data on various styles because everybody has their own interpretation and it is largely driven by the data provider body like MSCI who has defined indices based on various equity styles.

Do they develop their own suite of products?

They do have their own products, a quality investment fund, some competitors may define it differently but it all depend on what type of data it is in your disposal and also using your own discretion on what type of information you need, that also inform what is meant by "quality" which may vary with companies. For their fund they have do.ne research and valuation on it to package it to suite their style. It is however, difficult to do a comparison with other competitors because everybody wants to be unique in their styles.

Do they frequently develop this fund?

They don't really develop the fund frequently but always update their datasets that they have done the analysis on to make sure that the investment cases are still applicable and also

159

update with more recent data as well as analysing where they are in the cycle, if the analyse the fund they need to go back to where the expected outcome need to be. The don't necessary create too much products as they focuses in equity and the product stays and more refining the product outcomes. There BD is most financial data.

IQ1.1.5: How would big data add value in the organisation?

The more access you have in good quality data the better outcome you can create in terms of making sure that your analysis is based on clean data i.e. when you have a sort of investment process like the shortage of data e.g. the ESG data, some data providers do not have data that can add value and some data is difficult to obtain, how do you management of data e.g. you have a fraudulent financial statement that you can have a few built in character but if governance is the problem from bad management it is difficult to track those data points and having it reversed. People can say if they have better governance they could have been able to pick up Steinhoff early, because it may have a little bit of dubious financial statement but the problem being the management have governance problem, how you would have indicated data points allocated to that and that it becomes much more important in the investment world as they see the cooperate scandals evolving.

Do you think that the overall BD in the largest possible sense in the financial industry what could have been done better, and one would have picked up that Steinhoff are busy with dubious dealings or Ellerines?

If you have an adequate measuring tool one would say it is possible. They have already began to develop tools from their side for certain various indicators, there are textbooks on fraudulent accounting and indicators that you can build from you data to give you a warning signs as the account becomes less reliable, but there are some indicators that you can develop that relates to people as such how do you define that. They had a concern about African Bank when it occur and they didn't invest of it and also they didn't invest in Steihoff but EoH is one other type, they don't know what type of data is there in the market or is data related to how much is the contract the company have with government. When you look back is more related to certain business dealings, maybe there is more transparency in companies or government dealing in the economy will give you some information on that.

It seems that you can use BD for Risk Analysis?

Yes to avoid risk in the investment.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

Access to quality data it sometimes becomes an obstacle, and sometimes good quality data comes at a ridiculous price. Sometimes you are just need data related to indices and non-indices data is freely available. Such data comes in dollars most of it and you need dedicated

resources to analyse data like having a dedicated data manager. At this juncture they are a bit small but overtime they need to have this services.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

In this industry people use excel and some of the systems that they have internally because they have Bloomberg other use python data set. Access to tools is not too much of an obstacle for them unless if there are tools out there that they are not aware of. The challenge is the people to analyse and use that data and get meaningful analysis from them. There is always a shortage of skills when it comes to that.

Do you guys have the skills?

They do have the some guys but they just need to train them, they have taken in some graduates that they have to train them in terms of analysing. Most people have skills in data management/ quantitative management/ mathematical skills and not on the subject matter in terms of finance or stocks. It is a matter of bringing those skills together.

The data that they have is it structured or non-structured? Does it comes in different sources and different format?

It depend on the vender, but it is not structured and becomes a challenge when you use more than one source, and you need to spend time cleaning the data.

Do they have the right tools to structure the unstructured data?

They don't have and it is done manually. As a small company they don't have resources unlike the large cooperate they do not have IT department that is dedicated to work on data IQ1.2.3 Do they have a desire to know about big data?

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

There are benefits, and knowledge is power to have the data but it all depend on what you do with data matters at the end of the day. Data becomes more and more available but the challenge is the skills to analyse the data and applying it to your business or investment processes, finding a niche in your customer and developing the product that the customer wants.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

It will give you at least more topics to think about in terms of identifying blind spot because of the data is too much to process by individuals, it will give you areas to consider that you might have over looked before or develop products that you never considered or missed the opportunity that market certain types of product IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? If you understand the customers requirement or what the customer wants either be it a risk analysis on the customer needs or pick up trends on what the customers are doing even in term of different age groups for example what type of investment do customers really needs, provide customised solutions because not all 25 years old have the same objectives or act in the same manner and BD will identify the real nuisance of every customer segment.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

Not a clear defined strategy at this point of time but they would like to move towards a centralised data hub, to access more sensible data in the New Year. They do not use BD to a massive extent beside the financial risk management.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

Ongoing because they are a small team, at least once or twice per year, because the environment volatile changes regularly, regulations comes and go and some are outside their area and they need to look at them at least.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

They don't focus too much on the competition side and focus on their own offering maybe that is a draw back. On a monthly basis they do get surveys from their industry from there that where they can gage on where the competition is faring in terms of businesses, product offering and they try to look at that on a monthly basis in terms of how the competition is performing. But it can be improve in terms of Small Business

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

Yes you need to define some SFPs somewhere and what their focus areas is on about because some SFPs are client base, direct lines, some SFPs provide services to larger organisations and never deals with the client more like a business to business solution. If you want to look at data, define which type of client there is, if is in the selling side of the product, from their side is more on the underlining data for the actual investment that they look for, and the product side itself. And depending on how much data you need you can go to the underlining investment, fund level, customer level depending who you would want to services on this BD. Just get a sense of those SFPs that are out there in the marked and you can decide whose part of BD you would want to solve.

APPENDIX D9: INTERVIEW TRANSCRIPTION P9

Section A: participant's details

Personally Details:

What is your Name?

Name of your Company?

Miton Optimal

Your role in your Company (Job Title)?

Head of Business/ Chief Operating Officer

How long have been working in this organisation?

8 years

Helderburg Fruit juice business for 12 years

Sanlam (years not specified)

What does your company do?

They manage money, portfolio management on behalf of a lot of independent advisers. Regulations are busy changing where an advisor can do a needs analysis for a client and, decide on the product but as far as regulation is concern they cannot choose fund for the client. They are not qualified to do so. They are considered as Cat2 FSPs service provider meaning that they sign a discretionary mandate with a client to manage that money according to asset classes, risk profile and award of information.

What is your highest level of education?

B.Com Accounting

Do you have any employees working in your organisation, if yes how many?

Worldwide is 62 employees

South Africa is 37 employees

They have a offices in UK, Grenze and SA from the tax point of view

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics? Very little, they just recently appointed a business analyst nine month ago. Analysist has much to do with the systems that they have. Systems cost money, and the system that they have might not be exactly what they need but currently making use of the shelf products that they buy at a minimal cost. Those product provides 85% of what they require. The company is not prepared to spend 100 million to tick a box for the remaining 15%.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

Cost is their biggest obstacle. They have between 8000 and 10000 individual clients that they have a relationship with, however, if there was an easy and cheaper way to extract information and how to apply therefore they can consider. They are currently contracted with almost a 100 of FSPs that outsource their client investment management to Miton Optimal. That is what they did with Martin Klopper, who brought in a specialist in the financial advisor field to learn about financial investment, funeral cover, gap cover, medical cover and etc. It becomes more complicated such that the regulator keeps on changing the rules. As aforementioned one need to have a category 2 which is sub categorised in two where 2a requires you to run a fund that have a small of amount of 3 000 000 (running from their account not the companies money) or thirteen weeks of working capital in their banking account at any stage of the financial year. This is to avoid any adverse risk such as Ponzi scheme where clients has lost lots of money. In a nutshell they manage funds on behalf of the financial advisors. Scenario: if you have a million rand to invest and you approach Miton Optimal directly, a record of advice will be written, to understand the source of fund, fica requirements, proof of residence, identity verification and banking details as well as determination of the risk profile.

IQ1.1.3: How does your organisation categorise your existing customers?

Based on timeframe of relationship, their business started in 2001, old client based on the relation, client age and the size in investment (R100 000 is perceived as a small investment where as a R10 000 000 is a big investment). What they also do is if you have a R10 000 000, and you have a son that is saving R500 towards his retirement annuity, a wife with 30 000 tax free savings account they value the relationship (they manage relationship).

IQ1.1.4: How do you manage your competitiveness against your competitors?

Everything in the world is competitive, but all lies on good services, in the most realistic way. Their business runs on 25% base that company makes annually, that is for every Rand invested they got 0.25 return in investment and it takes time to build this kind of business and relationships as well.. Performance is key, if you are sour in performance you run a risk to lose a client the minute the market don't perform. Their priority is to be in the best performing, by taking a volatility risk and making it into a package that they can sell to advisors.

IQ1.1.5: How would big data add value in the organisation?

The business that they deal with does not use BD its focus is mainly on investment, they may require an upfront fee with the client but in most cases they do not require it. The only way that the business can be successful is to have a long term relationship with client for a period of 10years to make more profitable income. Online investment may be an approach but in South Africa is not yet forthcoming. Their approach was to then manage big pocket of fund online. They don't necessary develop a products, and most of the money they manage is discretionary fund, which different clients purchase in a form of unit trust, equity, commercial properties, bonds, money market what they blend into a package for a client. That is not really a product but investment money lying around different companies e.g. Alan grey, Investec.

The only product that they have is retirement annuity and tax free savings other that they do see indulgent policies or trade indulgent products as products but those as well they do not create them but adopt them from big investment companies, the tailor-made it according to the clients' need, some of them are great some poor based on the systems that they use. When they have a client the advisor will do record analysis but they conduct an objective analysis which is a needs analysis to determine the client's appetite and benchmark for the portfolio investment. They measure their performance using rolling terms which is CPI +3 over a three period.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

They don't have enough data. They have started the business in 2001 and they might be sitting with lots of information or trends that they do not realise that they have or in a position to extract them. At the moment they are using three various systems, for different purposes 1) for reporting, fees, and CRM perspective and the systems don't necessary talk to each other. IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

They think so, however, they don't think their skills are being deployed appropriately to generate more investment and more income at the end is to make money.

IQ1.2.3 Do they have a desire to know about big data?

Yes, they don't lose a lot of client but it will be interesting to know there are opportunities it offers, is there any trend in clients that they have lost in the last 24months, is it just about performance or service related issue. Also to determine the risk profile i.e. is it is a small investment and above a certain number they run a bespoke portfolio that is in a normal circumstances if they have two individuals with same portfolios (e.g. platinum clients) that have big investment, if you lose that client, the business get hit harder and it is therefore important to determine what they did wrong and what can be avoided to lose the other client.

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes they do.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

It goes back to knowing what to ask the client, in any area that they could be possible failing at the moment, perhaps their processes are wrong, and being in position of such information can put them in a position to be more competitive. Big Data can therefore be used to inform new business processes. The data that they currently have can be analysed from BD perspective to identify trends, and also provide opportunities to be ahead of the competition.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? A happy customer is for life. In the book I was your customer there is a phrase which says that the most dangerous customer is the one that never complains. Perhaps getting more information from their client will be of good advantage, what are they doing well, where to improve, what things are unnecessary. They have a regulatory responsibility that once a quarter, they must furnish their clients with a financial report.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics? No, to be consider in the New Year

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

In an annual basis, convened every year towards the middle of January, they also have a strategic session. The company do conduct, strategic asset management allocation that has to do with the investment side of things, they look at each of the asset classes and do research, forecasting and making prediction of how much they can get from the specific markets i.e. US equities the prediction are they will get 4%, bonds is 3% and based on that they put the whole things together. On the business side they have to look on compliance, accounting department, human resource, sales and etc., and determine what they are going to achieve in the next 12 months. With regard to compliance they have an in house compliance and have outsourced person that comes every second week.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

Every second week they have sales meeting, probably every second month they sit down analyse what their competitors are doing and what they are not doing. If there is something new they do consider if it is a necessity or not, does it add an additional responsibility. How do you sell?

They don't really sell but they have business portfolio manager working closely with the Independent Advisor that have the relationship with the client and also understand what is happening in the ground. Their biggest responsibility as Miton Optimal is to convince the SFPs to invest with them. In future there is an ACT that has been brought up in the UK and Australia that is called the Retail Distribution Review (RDR), where SFPs Category 1 will be forced to outsource their investment to Category 2 companies. Therefore, all those companies that are still running funds by themselves they will have to partner with to category 2 companies.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?
No, but require a transcript for preview purposes.

APPENDIX D10: INTERVIEW TRANSCRIPTION P10

Section A: participant's details

Personally Details:

What is your Name?

Name of your Company?

Martin Klopper Financial Services

Your role in your Company (Job Title)?

Owner and key individual

How long have been working in this organisation?

20 years with current employment

15 years with Old Mutual

What does your company do?

The company provides financial services, investments, medical funds, risk and life insurance, short-term insurance, strategic planning also working with investment specialist based in the US.

What is your highest level of education?

CFP-5; T2

Do you have any employees working in your organisation, if yes how many?

Six employees

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

They are a small company with no experience on BD analytics, but they do get their analysis from big companies like Discovery or old mutual which have one of the biggest database in the country. He gets most of data through connection, attending meeting to meet the right people and introduces himself, should he need any information he will contact them later.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

He has not come across any obstacles and it is difficult to answer, however obstacle is basically how easy to get the data, how reliable is the data for example Statistics South Africa is highly politicised and they don't provide the real stats.

IQ1.1.3: How does your organisation categorise your existing customers?

On occupation, gender, age, affordability, health. You said 500 tax cant go to a risk person with a product with a product like life insurance services, but he can service his existing life insurance. If he has a different view in a client, you can't make changes because they can't buy again.

IQ1.1.4: How do you manage your competitiveness against your competitors?

He is trying to be a one stop business, and see to it that his clients' gets advice from experts. Because they cannot be experts from everything, short-term, long-term, life investments, I they are a lot of companies now about 900-1000 that you can invest only in unit trust. There for you cannot do business with all those company, but you need to choose one that you can invest with. Currently they do most of their investment with Alan Greys because it is a company in the JSE and don't have to pay dividends to their clients. They have specialist that are always talking to the client, and always attend trainings. They are very good fund managers, always available even telephonically.

IQ1.1.5: How would big data add value in the organisation?

The more data you have the more business you can do, and focus on various things if for instance BD can tell him that company A, is looking for benefits for their people, you can approach them and sell services to them, inform them about your abilities, your experience, what you can do for them. On the computers or systems to make it easy, meet new client, retrieve data we need, currently of they go to someone they don't know they are just guessing and you don't know who you are walking to.

How is business, tough or easy?

Business is quite tough. He comes Worcester and has an ability to do work for the municipality but he does not have a BEE certificate, therefore he cannot provide services to them. The municipality gives business to others who resides a distance away from its vicinity while he is there and to provide personal services which is very important element of the business. It important to have personal relationship with your clients.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

No, they don't

Is this is something they are thinking of implementing?

They store most of their data in files and their backup system.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Yes they do, they belong to Moonstone in Stellenbosch a compliance officers, where they can source data in a certain way, their tools, electronics, internet are always 100% and if they can get big data they will be able to handle it.

But do they have necessary skills to handle the data?

Yes...

Is there any challenges that you think can affect SFPs to adopt BD, potentially?

Yes, they are being bullied by cooperates, which prefers people to work with them directly, they want people to do business through call centres, they do sponsorship, working with the politicians. They think private independents are taking away their business. For instance if you are looking for a quote of a short-term insurance and I give you a quote from 3-4 companies they fight like a dog over the bone. Mr Klopper tends not to like, but he gets a good rate for his client. He bullies them back, by saying that your client has been good paying you premiums for so long, small risk, and he can get a better rate for the client somewhere, if you still want to keep the client reduce the rates. They don't like that at all. But hey need us for data i.e. information about marital status, number of kids, what kind of business he is in. it is difficult for them to get this information, hence the dependent on Mr Klopper.

IQ1.2.3 Do they have a desire to know about big data?

Yes

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes. There more you are informed the better the advice you give.

Coming from a semi-rural area do you think there is more opportunities to analyse this massive amount of data that is available for company to assist in improving your portfolio and offering? Yes...

If there is an opportunity of having access to BD will utilise it? Yes.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

They use in trauma instances e.g. in this case they can go to a client and say they are still insurable and they have data that shows that people of your age are still suitable for a retake, you are smoker and have data from a reliable good source. I think we can increase your trauma insurance that can do you well than a life insurance in this instance i.e. breast cancer in women, 70% cause of death in women and medical aids has removed post trauma treatment. In some cases it can result in the removal of the breast and most medical aids don't pay for a transplant and reconstructive procedures.

Do you think with BD can you be able to improve profit, grow your business in terms of number of people?

Yes, it can improve the quality of services you provide and increase number of customers, however, you need to service your existing clients well and acquire one or two client a month that is a good business.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? Yes definitely depending on what BD gives you, you can be able to categories clients based on their statuses and prioritise accordingly.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

No strategy at the moment, but uses some contacts to solicit data.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

We know how many client they are losing and also find out why they are losing them if a client walks in they always ask for feedback on the clients experience on their services. They always discuss the feedbacks from clients in meetings, and what the causes of losing clients, e.g. we lost client A and he said we take long to provide feedback. We got a phone in client that was referred in by Mr and Mrs Y and they do offer Mr or Mrs Y with compensate.

Opportunities they do sponsorship, they are going to sponsor women golf in Worcester, and they are going to get data through Lukey. During the golf event they are going to run a competition with a prize, and participants will be required to write in a card their name, email address and contact details. They will also ask if they can contact them to inform them of their services and products. That how they collate this data which they sit will all the data.

Do you know the potential base of clients in your area and their profiles?

Yes more or less, that's if BD plays a role.

The threats in their business is more of empathy, sympathy type of thing. Example: a client comes from Pretoria they have a deaf or blind child they give up their jobs in Pretoria and insurance is a good start of a business and they tend to sympathise with in that type of situation.

Do you have lot of those kind of clients?

Yes young people always do come for advices

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

Daily, they have a company Martin Optimum specialised Investment Company, takes our clients' money and put a bit with Investec, Alan Gray and other companies and we are analysing that all the time. They have got specialist analysing the market making projections and forecasting on what will happen in the market, and they advise client that they must not get out of the market. Markets are very cheap even when there is no growth.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

It is difficult, the big problem they have in Financial Service Board (FSB) we don't know where they stand with them, they policy them like hawks, they attack individuals.

APPENDIX D11: INTERVIEW TRANSCRIPTION P11

Section A: participant's details Personally Details: What is your Name? Name of your Company? FC Wealth and Investments Your role in your Company (Job Title)? How long have been working in this organisation? Since the company opened in 2009, that's 11 years from now. What does your company do: Agriculture What is your highest level of education: Chartered Accountant (SA) Post Graduate diploma in Financial Planning Masters in Financial Management Certified Financial Planner (CFP) Do you have any employees working in your organisation, if yes how many: 9 permanent employees Can you tells us about your background of your organisation? The FC Group is a well-established Financial Firm of Chartered Accountants and Certified Financial Advisors. We offer comprehensive Wealth Management, Insurance, Tax and Accounting Solutions. RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage? SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses? IQ1.1.1: To what extent does your organisation have experience with big data analytics? We have enough data that we can use to improve our client products and also we can use it to analyse our client needs but we have not done anything with the data. As much we have the data but the skill within the company to use the data is not available and also it hasn't been

our focus.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

We do not have the skill for data analytics within the organisation at the moment and it will take too much time to set it up.

IQ1.1.3: How does your organisation categorise your existing customers?

We start by understanding enough about their needs and therefore that helps us to classify them in terms of their requirements standards because it's one of the things I do believe people do not focus on. For example, for us, to be able to produce sufficiently and meet our customer needs, I do need to know more about the informal market and formal market clients so that I can be able to meet their demands. We look at the number of clients and the data we get for those clients at a specific period vs a specific period in the previous year and that influences our planting.

IQ1.1.4: How do you manage your competitiveness against your competitors?

The more we know about our clients the better we can service them and service all their dissimilar needs. We all have some information about our clients and we service them according to what we know about them. So, our best strategy for now is to collaborate with other organisations so we can better service our clients.

IQ1.1.5: How would big data add value in the organisation?

As mentioned previously, it would help me to better understand my client needs and the peak periods so I can sufficiently cater for their services.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

Yes because most SMMEs do not understand big data at all let alone that some of them do not have data nor even know that they do have data. Some of the data is even too complex for them to even understand which is why I say this requires a special skill and time to understand and implement it so the organisation can reap the good benefits it is meant to bring.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Personally, I do not think they do have the skill and even understanding the concept of big data. They mostly have the data but not knowing what to do with it and what it can also do for them.

IQ1.2.3 Do they have a desire to know about big data?

Maybe there is interest in knowing more about big data but they do not know where to start as it probably seems like it is something complex and probably need more time. No one is ever willing to spend time and money on something they do not know how it will help them. More especially when they even know what problem is this going to solve.

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

We do not have a strategy yet but we are thinking about it because technology advances every day and as an organisation we would not want to be left behind in the technology evolution.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

Yes, we do and it is a continuous assessment that we do weekly as things change frequently. Any problem areas or issues, we talk about and address immediately. There is no better time to address a threat/risk than to address it instantly. And at most times that is a less costly approach as we are dealing with the situation before it expands or becomes costly.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

We constantly look at what can add value in contributing to our client. Looking at things that can be done e.g. Knowledge sharing, skills and development, reducing redundant tasks, etc. IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

No, I still need to think what to add here

APPENDIX D12: INTERVIEW TRANSCRIPTION P12

Section A: Participant's details
Personally Details:
What is your Name?
Name of your Company?
Optiversal health services services
Your role in your Company (Job Title)?
IT Manger
How long have been working in this organisation?
8 years
Lomil polymers for three years as IT manager
What does your company do?
They are a medical aid brokerage, they are the fulfilment channel for the medical aids leads that are coming into the industry. They have three separate companies:

Med-quote aid lead generator, with all online webpages and platforms that you can fill in what you are interested in and it collate data which feed into optivest

Optivest fulfilment channel and authorised financial service provider and they also provide advices to clients

Insuremed is a complementary medical aid gap cover which you can get it if you are a member of a medical aid. This is a complementary services of group of companies that works together. They provide services to all open medical aid in the industry. They do not provide this services to close medical aids like GEMS as they are not allowed to have a broker linked to it. For individuals they always do a full need analysis and from there determine what is the best option.

What category are they on:

Category 3 commission based company from the medical aid rates.

The medical aid pays the commission to Optivest on the first payment they receive from the client. In terms of the medical aid Act, for them to receive the payment, there are certain requirements that they need to fulfilment like how they services the clients. Each company has its own webpage i.e. <u>www.optivest.co.za</u>; <u>www.meqoute.co.za</u>; and <u>www.insuremed.co.za</u>.

What is your highest level of education?

B.Sc. (Hons) Computer Science and a degree in business science

Do you have any employees working in your organisation, if yes how many?

Optivest 105 employees and quite few sales consultants starting next year

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

They are aware of BD and the only challenge is time constraints. I am sure you will get this with many companies that you have interviewed. He has an entire list of projects for 2020 that are in the pipeline and some of them are carried over from the 2019 because they didn't finish them, therefore at this current moment, his focus in BD and analysing it is not a big priority. Will you consider outsourcing it?

They outsource very little and prefer keeping things in house. He has fulltime developers in his department; all server infrastructure; and the owner of the company likes to keep all the intellectual properties (IP). The owner of the company is Marcel du Toit. He is very sensitive to outsourcing, their databases are in house.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

Two things: Time constraints as aforementioned

Unstructured data, talking about emails they have thousands of emails that goes out of their server per day, under number of unstructured data sitting there and it is difficult to search. They do have some project for next year and without going into too much details, but they are planning on making their emails far more structured so that they can have them system based, such that they can be able to search through them and indexed, and with the whole point of BD they can be able to get information that can assist in making business that they cannot actually do in this as it currently stands with their emails. They currently have a data lake (vs data warehouse). They don't have a data warehouse and no analytics that is specifically being done. As it stands now their data is in a data lake coming as raw information and they query it as and when needed, if is requested from higher up. The developer has an experience with sequel database writing queries (the IT manager can still do it). They have an expertise and the challenge is just the time constraints

IQ1.1.3: How does your organisation categorise your existing customers?

The way they see their customers and how they categories them is based on the services and what are the complimentary services that they offers them. What they are offering vs what they can get from them. The complementary services are for specific clients but it boils down to need specific per client. For example, Medshield as a scheme have a lot of co-payment cost involved and their Gap cover product covers it. On the other hand you will find that another medical scheme does not offer co-payment as much as the Medishield therefore, you are prime to market the gap cover product is with the clients that are in the Medshield than the other ones. So in a nutshell is based on the customers' needs.

IQ1.1.4: How do you manage your competitiveness against your competitors?

By comparing our services and products to similar competitive products and services to see where a gap can be obtained, certainly they do this with their gap product cover as well annually and compare their products with other gap products out there. They have their own gap cover and therefore they are in charge of its benefit, check how much it cost, did it just get underwritten by another company. So the idea is to compare its benefit, what happen in the year, what types of claims did comes through, what were the needs of the client vs what other competitors are doing and adjust their benefits accordingly. This is a yearly exercise to try and make their product competitive.

IQ1.1.5: How would big data add value in the organisation?

The specific statistics that you can get especially from the data warehouse it will show us how many clients do they have, how many are in gap cover and how many will need gap-cover. It is mainly to obtain this kind of information that you cannot just get by looking at raw data, and it has to be analysed and to get trends and that is where they can draw that benefit from

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

As an organisation yes they are having challenges in adopting big data because they do not understand big data and are only aware about it.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Currently as it is stands they do have analytical tools. As aforementioned, analytical systems is not something that they have looked at, but if they do obviously they will have to make it a project that they can focus in more time, and they will definitely do through research on it to understand all the different tools that are out there in the market and specific choose the one that is tailored or carted to the kind of product and services that they offered by the company. Follow up question: Does you company has ability to handle unstructured data?

As aforementioned no, but they have a project to look into structuring their emails.

It seems though there is so much potential for big data but it gets to be re- reprioritised by other projects?

Yes

What is the benefits that you can get?

This is the battle that he has to fight for it to get higher priority, is for him to discuss the with the CEO BD benefits that the company can get out of it before tackling BD.

Does your organisation has necessary skills?

Yes, in his department he has people that have necessary experience to tackle that project. But just don't have the opportunity to do so, but he looks forward to it.

IQ1.2.3 Do they have a desire to know about big data?

Yes

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes they do,

Where do you see the benefits?

It is very much of what has been said, the analytics and the type of trends that you can pick out of it. That influences the decision making, of what offering you can give to your client, so that will give them competitive advantage.

Question: I have picked up two things that you have said: In the product development potential sale there is some benefit there and market hard-core market and market segment that you service in?

It is their servicing division which is a separate company called Medexpert which services all the clients that are already in their books and sales market intended for new clients.

Do they develop their own products?

Only from the gap cover side and from a service side they have a product that clients can opt to pay for and have separate contract with them, it has all lots of benefit that it comes with it. They could be some good opportunities coming from that poolside of the lake i.e. opportunities for new products development?

Not just for trends and the competitiveness of the product but from the employment side, shift kind, if we can start seeing quires coming to the client, what is the busiest time of the year. See time and day and turnaround time that client expect from them that will help them to deploy necessary resources from human point of view in a time of the day and the contractual point.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

It will put them in a position to identify gaps on products and services they provide to their client and allows them to get the benefits of the remaining loyalties. The loyalty of the client obviously depend on the gain to life. To run a risk to losing a client you need to offer them something that add to their life.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? It helps identify gaps in their service to their clients i.e. Customer segmentation...

How do you determine your customer needs so that you can improve their needs? This comes back to how do you categories your customers?

Obviously to them it depend on what scheme you are, what benefit does that scheme offers vs the complementary product, whether it add value, compared to what their gap cover is offering. It also has to do with clients that wants to pay for very less for their medical aid, and their gap cover is highly priced and goes for almost R220 per month. So people who are in a comprehensive skills are paying more for those services that they are not using and not taking in the kids or family into account. Those type of clients you can give them hospital plans and add on gap cover on it and they complement each other perfectly. They also have another product that they launched two months ago which is a health insurance product, it gives you all the day to day benefits such as dental and covers doctors visitation. The product is called Ubuntu-Med. If you add those three together gives you a good option. The product goes for R180 per month. If you add those three covers together with a hospital plan of R2000 and compare it to a comprehensive plan that a client pays between R6000-R7000 a month, you would have saved a client with almost R3000 a month. Clients don't know because medical aid are such a complex cell that they don't understand. They have a quoting system which they built years ago that you compares plans or schemes. Basically this system puts four schemes together and compares them and how do all the different schemes measures up to each other. It makes it easier and simpler for clients to understand, and that's where the lead comes from. You don't want on a website put lots of information because people are reluctant to put their personal information on a website and again you can't put long question. So on

the website is only information like name, telephone number and email and the consultant will call them and take them through all the process. So their services are telephonic based, recorded and stored in a telephone server. The telephonic data information is unstructured however, they have looked at software that they can tell you the words by listen to it but most of those software are oversea manufactured and they don't do well with the South Africa accent. They are also required by law that all conversation be recorded in case of investigation or the scheme can request the recording and a cell can take about 45 minutes. Finding the recording is easy particularly if you know the number, date and the person but getting through the recoding and trying to find what the people said is quite a challenge. Every consultant is unique, they do have a script but they don't have to follow the order. But from the compliance point of view they will have to reflect on everything required.

Information richness in those recording is huge?

Yes that where data security is very important especially with data encryption, offside backups because of private information of clients that is recorded.

Do you do that on a club?

No they have server hosted in their company and external backup drivers done daily.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

They do not have it now, it is something that they need to structure into a project and need to discuss with the CEO and try to sell it. This will definitely come from him and also develop a business case.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

They assess it annually, they go through everything that they do and assess how to better it, by going through the products, benefits and see where they can add something that will add to the client.

Are you part of those meeting?

Yes, his part of the Senior Management, and thus an opportunity for him to bring new ideas.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

They do it annually and based on the fact that by law the products that they offer needs to be reviewed annually, it also coincides with the benefit reviews that we do with the medical schemes and the new products are always effected from 1 January, the also assess all the benefits from the product annually and makes changes as required.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

No, I am happy with all the questions.

APPENDIX D13: INTERVIEW TRANSCRIPTION P13

Section A: Participant's details Personally Details: What is your Name? Name of your Company? We Trade Global Your role in your Company (Job Title)? CEO How long have been working in this organisation? 2012 Almost 8years Before that he was with the Green Road Bank CEO of Blank empowerment company called Siphumelele Investment What does your company do? They are product provide, they produce Unit Trust that are global property Unit Trust, they

want to go to institution finance advisor and individuals to go into their global property products What is meant by global product is that they trade in stock exchange around the globe. They decide on the allocation in each of those fund but in South Africa terminology they've got Unit Trust both here, in Malta, Australia and other places in due course and all of them are portfolio of property vehicles and most of them are Real Estate Investment Trust (REIT). They focus on Property and they got REIT on the JSE growth point, Redefined, Vukile they decide on the universal of REIT, what goes into their portfolio is about 2000 of REIT Globally and they construct a portfolio that performance the average of about 40 and they increase, decrease and remove and add onto those portfolio for certain Unit Trust that they manage. This is done on a daily basis.

How does your portfolio perform against the others?

They run for the Raging bull award which is a top performance award, they are in the line to possible win again this year. There are 16 competitors here in South Africa that produce global property funds and they are always in the top six.

And against other types of funds?

Well, it is a niche and they believe that global property is a component of your portfolio so they outperform South African Unit Trust that are focusing on JSE by far. This year they have 25% in Rands, in dollar 23% and JSE is -3% but it does have its ups and downs. Last year they only comes out with 4%, it is not a straight line.

What do they mean by 4%?

4% in total returns. On an average for the past two years they had 15% of the returns.

What is your highest level of education?

Charted accountant

Do you have any employees working in your organisation, if yes how many?

Six employees, the staff in the company are highly qualified, two employees with MBA, CA and CFA and Business administration

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

They are now starting to do it and they are working with an organisation in Stellenbosch for six months to use BD to analyse, extract, to enhance their performance and to reduce the emotive elements in their stock selection, allocation, which no matter how much you may want to avoid this but you cannot.

However, the crucial thing is they cannot totally depend on it because there is human intervention and BD cannot see certain things no matter how much of machine learning and BD that you have. For the next three months (end of March) what they have requested from their associates in Stellenbosch they must supply them with BD information so that they can clematises or adjust themselves and see if they will be able to use the data to enhance their performance because everything in investment is based on performance.

Will it be okay to give you a call in May/June to just get feedback on how it is going as he will be writing up his thesis?

They have their Portfolio and BD can give them a different portfolio which can throw them into a spin, they have a process to arrive at that and they are doing well and they suspect that BD will deliver something that is relatively close and they go and do it their way, actually they are now within days of getting it done.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

It comes with a cost and knowledge on how to use it. It has taken them a long time to understand property, REIT and process, and to go and develop skills now on how to access BD, it is a completely new different set of skills. They are working with people who has experience, right now if you ask me to go find data, I would know but this guys do and they got it and they have given them certain information. You need to negotiate contracts to get information and one wouldn't know who to call and request for BD. This organisation that they are working with specialises with financial BD, not marketing BD and that is so crucial. To arrive to them, they had to speak to different organisations and they listen in and could see that they were going to give the marketing BD whereas to them performance consist of probably 49% of the weight to success and it takes time and the other 51% you have to stand

in front of people to get it, you have to tell them what is all about, and they talk to other people. Right now they are in a process of developing a consumer product, identify that as well. The best way to answer is that there are some organisation that says that they have funds called X, Y and Z (artificial intelligent) and general equity funds, this funds are not growing and they have spoken to a number of people and they don't trust them, that is the marketing things and they go back to 2008 before the crush of things and they have blackbox and they were secretive on their dealings and they crashed and the public wanted to know what is going on and they said they are using AI. So there are a lot of scepticism about AI, not all of them are delivering magnificent results and this organisation in Stellenbosch they concentrate on information to enhance performance and they said it has to have human intervention. Human intervention is not just talking it, should also include explaining to the public what you are doing, if we say we have a process and we are using this to select components is a complete different story that they will just sit and push the button and generate results, nobody will go for it.

It seems that the context and the experience in the specific gain like property is very important in the analytics as well, one couldn't just go and do?

We use it to identify prospects and thereafter they go and do the work because is 2000 it goes and cuts down, that is the way they are going to use it. They are still going to do their own evaluation, assess management, also it will give them a guide of the allocations. Do you know that stock selection is secondary to performance and is very secondary because of the allocation, you may find a fantastic stock and you allocate too much to it and everything goes wrong, often when is a good prospects it comes with a risk and is a good prospects and put little into your portfolio. Allocation is a primary contributor to performance, the secondary one is your selection, whereas the public thinks that selection is the actual allocation and they will help us with the allocation (this is what we think you are doing), they also decide on the selection, they may know that there are some issues there that Al doesn't know and the machine learning doesn't know, and that how they are using it.

IQ1.1.3: How does your organisation categorise your existing customers?

They have got three categories:

Firstly direct as they call them, people comes to them. Do people call you? Generally they are peoples family and friends who then speck to other friends and family and see what is it and that's what they call direct customers. Lots of them referrals;

Financial advisor attached to them are discretionary fund managers who often provide advice to financial advisors on what they should do; and

Institution, they absolutely do not have institution because BEE blocks them no matter what they do. Even if it is small? Even if they performing is a frustrating thing. If they are out performing someone by 10%, the institutions are forced by legislation to the demand of South

Africa and the investor and they absolutely have no chance. The PIC has been producing 2% of compound growth for the last 10 -15 years and even if inflation is negative? His also talking from being a CEO of a black empowerment company and there are whole lot of obstacles and institution like Sanlam are being forced to direct their investment to this underperforming ones because of race. Even if they can give away 90% of their shareholding and went and went as they are they wouldn't do it. It takes years to develop somebody. What they believe on is that sanity will prevail and continue to outperform. They are not an institution but is one of their prospects.

Do you look at high net worth?

Their minimum is 25 000 and they take the whole range

IQ1.1.4: How do you manage your competitiveness against your competitors?

We outperform them, that's how you manage performance and of course the cost structure plays a role, you can be beating someone by 5% but if they got higher than theirs the DFM will say is a cheaper cost and you can outperform them when your fund is low.

IQ1.1.5: How would big data add value in the organisation?

Performance, in due course like now they are doing work with ABSA bank now that are more consumer. Unit trust are sold on trust, where people knows us and tell each other (the directs), then the financial advisors and that is really the market whereas they will be doing products like the ETF exchange traded fund which is more consumer oriented. And in due course they will using marketing data to actual get to those people. It takes long time and money to set up but they will have an ETF up and going by the end of February and they can run for a while. Probably by 2021 they will use BD in marketing materials.

Do you use Centrex approach with your trust?

No, Centrex is an index exchange traded funds so you have two categories which are active managers, where you actively go and research and put forward a portfolio and you have an exchange traded fund that have got an index and they replicate and index. They got a bench mark that shows how they are performing relatively to their index but the down side of the ETF is that nobody has done any research of the underlining holding, nobody can tell you that they thinks there are enough spares or run out of steam or is time to go down. The two ETFs makes about 50% of the market that is active managers are 50%, in South Africa is only 5% point in time. What you can do is to blend the two, as aforementioned they are launching ETF in February, and what they did they took a component of their global ETF and source some behavioural things and created an index REIT way index that out performs the global property index. So don't start a product already there are two global property in SA and there is no need to start a new one because is going to be two so they have and index that outperforms the index, it will be with ABSA in February and we saw that behaviour and it will be a new fund REIT Way Global Property ETF which is partnership with their company and ABSA Bank.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

Yes, some of them don't not understand big data.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

Yes and No, they are taking the results of the analysis. You need to have the skills. They are running a system in which seven days later it comes out...

Can you handle unstructured data?

They are receiving structured data and they do not have necessary skills to handle data.

Do you want to or happy with outsourcing the skills?

His fanatical about outsourcing and stick to what he is good at for instances the administration of the unit trust fund they don't do it, they have a custodian that does the calculation, pricing, receives the cash, publishes the daily price but it is called the REIT Way Global property Fund. They do the buying and selling, certain organisation they do their own administration like Old Mutual.

IQ1.2.3 Do they have a desire to know about big data?

Definitely performance

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Definitely performance

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

Performance

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? Customers comes to them because they want performance. As aforementioned he sells to friend and families therefore he has to deliver performance. As long as you perform the customers are happy. There is a guy called Dave Ford, he delivers great performance and he has lost his way now he is very treacherous now and noxious and his underperforming and he should lose 20% of his assets under management. He is treacherous and inaccessible and lost 40% of his assets under management. You need to be a possible person.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

Yes, they will be outsourcing the BD to create structured data from the unstructured data. So they will be defining what they mean of structured data for their needs to people who provides. You still need to define their structured data?

Yes we will tell them what they want and how they want it and they will deliver it accordingly. They know exactly what they want.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

All day every day but when they really put their heads together is at least once a month. 2020 will be Bloodbath in this industry and they have to prepare for it.

By saying it will be bloodbath do you mean that the industry is going to underperform or because of the difficulties for FSPs to perform?

ETFs goes for 5%-10% and in America is already 15%, so if they go up to 15% it is going to be a major disruption, internally people are going to be kicked out if you are not performing. With regard to the pensions, the challenge that is depriving South Africans is the regulation 28 which says that 70% can be invested in SA and 30% abroad. Their fund fits within the 30%, they've got their own retirement annuity that they put 15% of it to REIT Way Global and it has been the best performer. The 70% in the JSE and this is the government ideology at the end of the day and you have to invest in SA, the JSE floods when commodity and resources floods. It sucks oxygen when commodities and resources don't perform. Commodity and resources goes up and down and even when they construct their property funds they increase their exposures to Japan, and you can increase exposure to the US and so on. Commodity and resources are synchronised to different things you can go and buy Tessellar at the bottom, facebook and you don't want to be exposed to Naspers is too risk and you need to cut out emotion. They are actively making changes and that 70% of their portfolio was -5% and the 30% of which 15% is with REIT way is 25%. The net results is they are positive 5% or 6/7%. You go to explain this Gordan and Abrahim Patel to trade and industry and of course they won't agree to offshore investment, but it is artificial and at the JSE you are buying Glencoe, Kumba and they are highly priced. As pensioner you need to reduce your exposure and requirement of Regulation 28 and put your money in the money market. Money market has done very well and is 3% but there is no globally property.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

Once a month and they put their heads together is at least once a month. There are newspapers and media publishes their performance in a daily basis, they have a thing called funds on Friday and they give performance of that week, and then you have for a month, three months, three years, five years and seven years performance. So they are seeing what their competitors are doing, and analyse them from the performance point of view weekly.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

The whole conversation about AI machine learning and BD are synonymous, the human element and how is going to take over our lives is over stated by small degree. It will take generation and people will go to tally machine generated investment decisions. Many years ago you had human stock brokers taking your orders and online was going to come along to replace those people, what happen is that they found their niche and that is there are brokers who still feels the market and they you have the online trading.

ETF vs Active Management, the power of active management is decreasing because ETFs are taking over and it won't be forever because you still have to allocate.

Robot advisors have been around for long, they are providing advises and they are getting more sophisticated, a small portion of investment that comes from robot advisors and it is going to grow.

BD is overstated, and it is going to make certain thing to be redundant.

People don't trust machine

SANLAM has got Eight Fund called AI general Equity funds, they are not growing.

APPENDIX D14: INTERVIEW TRANSCRIPTION P14

Section A: Participant's details

Personally Details:

What is your Name?

Name of your Company?

Reason Consulting

Your role in your Company (Job Title)?

Key individual for short-term insurance and representative on all the other line of business thus include providing advises on life insurance cover and medical schemes.

How long have been working in this organisation?

3 years

What does your company do?

8 years as a Professional accounting in an accounting and auditing firm in Pretoria

What is your highest level of education?

Bachelor degree in Internal Auditing and member of the South African Institution of Professional Accountant and South African Institution of Internal Auditors

Do you have any employees working in your organisation, if yes how many?

Yes, one employee who handles administration and some of the cases.

Can you tells us about your background of your organisation?

They are a financial service provider, and they are not linked to any insurer or investment house, they are independent. They have contract with different insurers, different investment houses and different medical schemes. Therefore, they are licenced to sell or to consult regarding those products. In a nutshell they are financial service prodder, providing advices in different products.

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses? IQ1.1.1: To what extent does your organisation have experience with big data analytics? Not really, the data that they have is for their clients which is confidential. The data is electronically and they do not work on paperwork. They do not analyse the data but just store it. According to the FSA which is a professional body that requires them to at least have records for five (5) years for all advice given and correspondence to clients before disposal. IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data

analytics?

No, he doesn't think so.

Why are they not analysing data, is it shortage of skills or do not have time?

He doesn't think they have time for that or in a line of their profession nevertheless there is a benefit to analyse the data. Other insurers has approached them to provide with advises on the analysis to push/increase business however this is not done as yet. He doesn't think there is a need currently for that beyond what they are current business process which is data storage and so forth.

IQ1.1.3: How does your organisation categorise your existing customers?

Based on the four spectrum:

Young to old

Professional to non-professional

Basically there are two representatives in the company, the much older one is Ernest his about 64 years and look after the much older client and himself Johan who is 35years old n look after the younger clients' portfolio.

IQ1.1.4: How do you manage your competitiveness against your competitors?

He believes the way they do business, the operation of the business is conducted electronic and overall they manage their competitiveness based on the service they provide. Their correspondence are mostly electronically and on record, sometimes you have brokers that go visit the clients and their engagements are not well documented. So far 2/3 of their client haven't even met them.

IQ1.1.5: How would big data add value in the organisation?

He believe so, there is a space for BD analytics to push/take their business to the next level and he doesn't think they have the capacity at this moment to drive that.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

Yes, they SMMEs do have challenges in adopting BD. Big companies like Discovery the data that they have is incredible and they can use that data to improve their whole business model. He doesn't think that SMMEs has capacity since they are smaller than the cooperate front. When you need to employ people the business doesn't rectify that change.

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

No,

IQ1.2.3 Do they have a desire to know about big data?

Yes

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage?

SQ2.1: How do SMMEs determine the needs of their customers?

IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes. Obviously to come up with the whole spectrum to serve the client and BD analytics can give you that opportunity.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? Improve service delivery, supplying the client with holistic service, covering all the needs of the client basically.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

No, currently with the service provider that they use, they sometimes have information of their clients. Basically all their clients that have Discovery, Discovery does have their information of that client, they give it to them when they submit all the information with application and Fica requirements documents. Discovery has approach them before with information to improve business. As aforementioned they don't have capacity for BD at the moment.

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

They receive that information in a weekly basis from their service providers informing them on what is going on in the world of investment and risk covers. They basically as part of their ongoing training they have to look at that information and sometimes do an assessment to make sure that they are constantly learning.

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

He doesn't think there is a big competition attitude among brokers or financial advisors. Therefore, there is not really a need to assess competiveness because there is not a lot of

financial consultancy out there and lot of people out there. There is enough business for one out there without having a need to compete, you much just make sure that your business model, advises and knowledge is up to scratch.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

No.

APPENDIX D15: INTERVIEW TRANSCRIPTION P15

Section A: Participant's details

Personally Details:

What is your Name?

Name of your Company?

Broker-Fin Insurance solutions

Your role in your Company (Job Title)?

Principal owner of the business

How long have been working in this organisation?

37 years and they have 2600 clients all over SA and they manage 3billion in Rand

What does your company do?

They are an investment brokerage company and they specialise in investment.

What is your highest level of education?

Three year diploma in management and qualified company secretary, he also have all the FSP certificates. Level 1, they access level -2 through the other company they use.

Do you have any employees working in your organisation, if yes how many?

14 staff members

Can you tells us about your background of your organisation?

RQ1: What are the challenges SMMEs face when using BD analytics in building and differentiation customer segmentation to create a competitive advantage?

SQ1.1: What are the perceptions of SMMEs towards the value of BD for their businesses?

IQ1.1.1: To what extent does your organisation have experience with big data analytics?

They have a sizeable experience on BD. In the past they had experience and spent sometimes analysing client information. To give an indication on what they do, they are an investment brokerage company and they specialise in investment and do not spend time on short-term insurance, medical aid and others, instead they have a specialised areas of where they do their operations. So what they have done per client: they have a whole variety of investment and obviously that comes in a form of a statistic data and from time to time analyse it and

make it their business to ensure that all their clients various investments on their brokerage code. They enjoy analysing the data and segmenting it to various forms like investment, retirement annuity and etc.

Yes, they have devoted a lot of time on analysing data and it has been working for them tremendously.

IQ1.1.2: What are the main obstacles that would prevent your organisation in using big data analytics?

In their specific circumstances, what is a bit of a problem for them is to get information pertaining people that are retiring for an example at the University of Stellenbosch is not easy to access information of people that are retiring and it is an obstacle. If they would be availability of such specific data regarding people retiring they could obviously be pressurised to get to the people and convince them that they can be their broker concerned. To them is an obstacle because that data it is secured and they cannot get to it.

IQ1.1.3: How does your organisation categorise your existing customers?

They sub-divide their clients as a); b); c) and d) in terms of the amount money that they have investment with them.

IQ1.1.4: How do you manage your competitiveness against your competitors?

This is something that they don't really have a problem with in the market place based on the quality of advice that they give and that serve as a sort of a divider between them and the next person. For them is not really a problem in a marketplace.

IQ1.1.5: How would big data add value in the organisation?

It will add value if the data is specifically to help the source the people that they would work with and if is not that specifically it won't assist at all.

RSQ 1.2: What are the factors affecting SMMEs when adopting BD analytics for creating a competitive advantage?

IQ1.2.1 Do you think SMMEs are having challenges in adopting BD?

Yes, they spend lot of money updating their computer and systems specifically their databases (databases are so much expensive and it host a lot of information and it needs to be dealt with by specialised staff and it makes whole of life easily).

IQ1.2.2 Do you think SMMEs have the right analytical tools or skills to handle unstructured data?

They do not use unstructured data. If it is possible that it could add on what they are doing obviously they would like to investigate but at the moment it is not an issue at this stage.

IQ1.2.3 Do they have a desire to know about big data?

Yes, they have and as aforementioned, they have devoted a lot of time on specialised and training them first and those are in house people.

RQ2: How can SMMEs utilise BD analytics in order to create a competitive advantage? SQ2.1: How do SMMEs determine the needs of their customers? IQ2.1.1 Do you see any benefit in using big data to create competitive advantage in your organisation?

Yes, if you can use BD positively it will enhance growth of the company, the more clients they have the more money they have under management and the bigger the possibility of negotiating rates in the market. Therefore BD can create a competitive advantage from the administration and management point of view and within that you can save your clients lots of money.

IQ2.1.2 To what extent would big data analytics add to the competitive advantage of your company?

Only to the extent that they have lots of information or attain more information of people that are retiring it will obviously enhance what they do.

IQ2.1.3 In your own view, what can big data analytics do to improve your customer's needs? In their own fields there is little that can be done because they spend a lot of time per client identifying specific needs are in a variety of fields, so if it is not specifically same investments to your client it won't help much.

Obviously he doesn't know.

SQ2.2: How can SMMEs incorporate BD in their business strategies?

IQ2.2.1: Does your organisation have a strategy on big data or data analytics?

They do, basically they analyse all the information that they have and devote lot of time on the information to ensure that whatever the clients' investments they are, would be under their management

IQ2.2.2: How often does your organisation assess its strengths, weaknesses, opportunities and threats in order to understand the current business climate?

They do it annually, they have an annual planning session which last for three days, the meetings has a structured agenda and also consider the company's strategy amongst other things. The meeting also plan for the following year

IQ2.2.3: How often does your organisation analyse the competition in order to understand competitive advantages and disadvantages as well as identify areas for investment or needs for improvement?

They do it in a continuous bases. They are part of a big organisation called Finsonet, which is a group of 64 independent brokers. They have quarterly meetings which constantly analysing what the competitors are doing, with regard to strategies, investment, costing and etc. Basically it is an ongoing thing and gets a lot of attention.

IQ2.2.4: Is there anything you would like us to add or anything you feel we have missed asking?

It is difficult in the moment to provide an indication on that, however, he thinks that an attitude towards this is being confirmed in a questions above. This is so important particularly for new business and be incorporated into new business strategies for their benefits and it would also

be nice if they can have a system that could have a system in which they can use BD specifically in their fields of expertise.